Linköping Studies in Science and Technology

Thesis No. 1329

## Deductive Planning and Composite Actions in Temporal Action Logic

by

## Martin Magnusson



Linköping University

Submitted to Linköping Institute of Technology at Linköping University in partial fulfilment of the requirements for degree of Licentiate of Engineering

> Department of Computer and Information Science Linköpings universitet SE-58183 Linköping, Sweden

> > Linköping 2007

## Deductive Planning and Composite Actions in Temporal Action Logic

by

Martin Magnusson

September 2007 ISBN 978-91-85895-93-9 Linköping Studies in Science and Technology Thesis No. 1329 ISSN 0280-7971 LiU-Tek-Lic-2007:38

#### ABSTRACT

Temporal Action Logic is a well established logical formalism for reasoning about action and change that has long been used as a formal specification language. Its first-order characterization and explicit time representation makes it a suitable target for automated theorem proving and the application of temporal constraint solvers. We introduce a translation from a subset of Temporal Action Logic to constraint logic programs that takes advantage of these characteristics to make the logic applicable, not just as a formal specification language, but in solving practical reasoning problems. Extensions are introduced that enable the generation of action sequences, thus paving the road for interesting applications in deductive planning. The use of qualitative temporal constraints makes it possible to follow a least commitment strategy and construct partially ordered plans. Furthermore, the logical language and logic program translation is extended with the notion of composite actions that can be used to formulate and execute scripted plans with conditional actions, non-deterministic choices, and loops. The resulting planner and reasoner is integrated with a graphical user interface in our autonomous helicopter research system and applied to logistics problems. Solution plans are synthesized together with monitoring constraints that trigger the generation of recovery actions in cases of execution failures.

This work is supported in part by a grant from the Swedish national aeronautics research program NFFP04 S4203 and a Swedish research council grant 50364201.

Department of Computer and Information Science Linköpings universitet SE-581 83 Linköping, Sweden

## Acknowledgments

I would like to thank my advisor Patrick Doherty for his contagious enthusiasm for research, for entrusting me with great freedom in pursuing whatever topics that have interested me, and for paying me to enjoy my hobby. Two other great researchers that I have had the pleasure of collaborating with are Jonas Kvarnström and Andrzej Szalas. I am looking forward to more opportunities for joint work in the future.

My colleges at the department of Computer Science at Linköping University has made it a wonderful environment to work in. I sincerely thank Per Nyblom, David Landén, Gustav Nordh, Per-Olof Petersson, Fredrik Heintz, and Tommy Persson for being such good friends. Thank you Jenny Ljung, Anna Maria Uhlin, Madeleine Häger Dahlqvist, and Piotr Rudol for invaluable assistance.

The choice to pursue academic studies was made with the encouragement and support from Eva-Lena Lengquist with whom I have had the great fortune to share many happy moments. Steve Pavlina has unknowingly helped me tremendously towards my goal through his innumerable inspirational writings on how to improve any and every area of your life. My final note of gratitude goes to my parents and sister for your unconditional love.

# Contents

1 Intr		oduction	1	
	1.1	The Case for Deductive Planning	2	
	1.2	Contributions	2	
	1.3	Publications	4	
	1.4	Thesis Outline	4	
<b>2</b>	Ten	aporal Action Logic	5	
	2.1	Basic Concepts	5	
	2.2	A TAL Narrative	6	
	2.3	The High-level Language	7	
		2.3.1 The Translation Function	8	
	2.4	Foundational Axioms	0	
	2.5	Circumscription Policy	1	
	2.6	Reasoning in TAL 1	2	
		2.6.1 Natural Deduction	2	
		2.6.2 Automated Reasoning	4	
3	Deductive Planning 15			
	3.1	Reified Action Occurrences	5	
	3.2	Interval Occlusion	7	
		3.2.1 Temporal Constraint Formalisms	8	
	3.3	A Constructive Plan Existence Proof	9	
<b>4</b>	Cor	npiling TAL into Logic Programs 2	7	
	4.1	$\mathcal{L}(ND)$ Narrative	8	
	4.2	Translation from $\mathcal{L}(ND)$ to $\mathcal{L}(FL)$	9	
	4.3	Introducing Interval Occlusion	0	
	4.4	Transformation to Horn Form	2	
	4.5	Circumscription Policy	4	

	4.6	Clauses and Rules			
	4.7	Summary 4			
<b>5</b>	Composite Actions 4				
	5.1	Syntax and Semantics			
	5.2	Fixpoint Expansion			
		5.2.1 Expansion Example $\dots \dots \dots$			
	5.3	Compiling Composite Actions			
		5.3.1 The Compilation Function			
		5.3.2 An Example Composite Action			
	5.4	Reified Composite Actions			
6	A U	AV Logistics Application 53			
	6.1	Graphical User Interface			
	6.2	Execution and Monitoring			
7	Rela	ated Work 59			
8	Disc	cussion 65			
	8.1	Future Research			
	8.2	Conclusions			
A	The	Logistics Scenario Prolog Code 69			
в	Pro	ofs 75			
	B.1	Interval Persistence Formula			
	B.2	Interval End-point Equivalence			
	B.3	Point-interval Rewrite			
	B.4	Shared Timepoint Overlap Equivalence			

## Chapter 1

# Introduction

Artificial intelligence is the challenge of creating a thinking machine. Although definitions of intelligence and thinking are plentiful, some capabilities seem absolutely necessary for any system proposed as either. Of central importance is the capability for rational reasoning, especially reasoning about what can be done to achieve ones goals. Work within the methodology of formal logic provides a comprehensive toolset for correct reasoning and is thus a natural choice for progress toward this end. However, the standard philosophical logic turns out to be inadequate, and artificial intelligence researchers are therefore creating new powerful logics that are applicable to human-like commonsense reasoning as well as more traditional logical reasoning.

Temporal Action Logic (TAL) [8] is one such logic developed at Linköping University. TAL has, since its inception more than ten years ago, proven to be a highly versatile and expressive formalism. Nevertheless, many important areas of research can be identified that provide excellent opportunities for improving the logic. In this thesis we present extensions to TAL that make it applicable to deductive planning, i.e. reasoning about what actions achieve a given goal, and we touch upon the topic of composite actions, i.e. complex actions built from simpler actions, conditionals and loops.

This and most other artificial intelligence research pursued at Linköping University is part of the Autonomous Unmanned Aerial Vehicle (UAV) Technologies Laboratory and its long-term basic research initiative [6] that focuses on the development of autonomous robots that display high-level cognitive behaviour with the primary experimental platforms being autonomous UAVs. Having such working systems in mind helps put emphasis on techniques that are not purely of theoretical interest but also useful in practice. We want to make logic work.

## 1.1 The Case for Deductive Planning

By *deductive planning* we shall understand the direct synthesis of plans through a deductive proof process in a logic of action and change. In contrast most of the work in automated planning makes use of algorithms, formalisms, and programs specialized for the planning task. This kind of automated planning has seen great progress and has developed into a well-defined research topic with specialized conferences and a respectable body of work. In light of this fact, one might question the need to cast the planning process as deduction.

To answer this question we would like to appeal to the bigger picture. The artificial intelligence challenge requires an attempt at building a generally intelligent system that is applicable to all problems of any complexity. To be practically possible this would seem to require formalisms and technologies that exhibit two important properties. First, note that it is clearly impossible to foresee all possible problems requiring intelligence, in advance of the artificial agent being confronted by them, and to develop special purpose solutions to each and every one of them. The alternative is a striving for a property of *uniformity* that would allow most problems to be expressed in a similar way, and different or unforeseen problems to be attacked using the same techniques. This necessitates the second property of *expressivity*, to have the ability to encode all the different complexities of problems, and to be able to reason with them in the same framework. Deductive planning contributes uniformity since the planning process is cast as deduction in the same way as other types of reasoning such as prediction. It also supports expressivity since the logics used are often among the most expressive formalisms known that can be used to encode complex problems in a well understood way.

Before we get too carried away, however, let us admit that the present state of deductive planning still has a long way to go before fulfilling these ideals. The point we would like to make is that work in this direction is important and that deductive planning enjoys properties that could help progress towards the long term (as well as the short term) goal.

## **1.2** Contributions

One of the distinguishing characteristics of Temporal Action Logic is its use of explicit time structures whose meaning can be explained through semantic attachment or through axiomatizations in classical logic. This feature can be taken advantage of by combining automated theorem proving with specialized temporal constraint propagation algorithms for dealing with temporal structures in TAL narratives. First-order automated theorem provers can be inefficient and difficult to use in practical applications, but logic programming, exemplified by Prolog, is a well-known alternative that has been successfully applied to many logical reasoning problems. Although logic programming technologies are based on theorem proving, they sacrifice some of the expressiveness of first-order logic to gain simplicity of use and, often, efficiency.

We present a translation from a subset of the full TAL formalism to logic programs that use Prolog's finite domain constraint solver and the Constraint Handling Rules [10] framework. The resulting programs take advantage of these constraint solving tools to make TAL directly applicable to practical planning problems in our autonomous Unmanned Aerial Vehicle (UAV) system. TAL has been used in previous work as a formal specification language for a very powerful forward-chaining planner called TALplanner [7]. A formally specified TAL goal narrative is input into the procedural planner, which in turn outputs a plan narrative whose formal semantics is also based on TAL. In this case, TAL is used as a formal specification tool and the plan generation mechanism operates outside the logic. In contrast, this thesis introduces additional extensions to TAL that enable deductive planning at the object level rather than metatheoretically. In order to do this both action occurrences and sets of action occurrences (a form of plan or narrative) will be introduced as terms and sets in the object language, respectively. In this manner, one may generate partially ordered plans deductively through the execution of the constraint logic program that was translated from the TAL narrative.

Since the reasoning is deductive, the process works both ways. One may provide a complete plan as input and deduce its effects, specify only declarative goals that need to be satisfied, or explore the middle ground between these extremes. Sometimes, though, one may want to provide a complete plan for the robot to execute that can not be expressed as a sequence of simple actions. Many tasks involve choices that are conditional on some property of the world, or the repeated application of some sequence of actions until a condition has been satisfied. We extend the TAL language with macros that enable the encoding of such composite actions and extend the constraint logic programming translation so that they can be executed in the same logic programming framework.

This framework, named PARADOCS for Planning And Reasoning As DeductiOn with ConstraintS, has been integrated with our UAV system and applied to a logistics scenario. A UAV operator can set up a mission graphically through a user interface, which also visualizes the resulting plan and its execution. Value persistency constraints in the plan are continually monitored, and trigger failure recovery when violated. Recovery is possible by providing the plan fragment that has yet to be executed as input to the deductive planner and reasoner. It will complete the partial plan with additional recovery actions as needed to make it valid again.

## 1.3 Publications

Parts of this thesis have previously been presented in the following publications:

- [24] Martin Magnusson and Patrick Doherty. Deductive planning with temporal constraints using TAL. In Proceedings of the International Symposium on Practical Cognitive Agents and Robots (PCAR'06), 2006.
- [25] Martin Magnusson and Patrick Doherty. Deductive planning with temporal constraints. In Logical Formalizations of Commonsense Reasoning: Papers from 2007 AAAI Spring Symposium, Technical Report SS-07-05, 2007.

## 1.4 Thesis Outline

The structure of this thesis is as follows. Chapter 2 provides an introduction to Temporal Action Logic sufficient to support understanding of the subsequent chapters. Chapter 3 introduces extensions that enable deductive planning with TAL and provides an example constructive proof of a simple planning goal from which a solution plan can be extracted. Chapter 4 details the compilation process that automates the planning process using constraint logic programming and temporal constraint formalisms. Chapter 5 specifies the semantics of composite actions and complements the compilation process so that they can be evaluated. Chapter 6 presents an integration of the developed methods with the UAV platform and demonstrates the application of the system to a logistics problem. Chapter 7 relates the work to some of its sources of inspiration. Chapter 8 sums up the results and discusses future directions of research. Finally, an appendix contains the Prolog clauses that were used in the logistics application.

## Chapter 2

# **Temporal Action Logic**

Temporal Action Logic has its roots in Sandewall's book *Features and Fluents* [32] published in 1994 but has since then been developed in a direction of its own by Doherty, Karlsson, Gustafsson, Kvarnström, and others (see e.g. [5, 17, 12, 19]). TAL differs from the well-known Situation Calculus and other logical formalisms for reasoning about action and change with its use of explicit time, occlusion, and a high-level language that provides an abstraction layer free from technical constructions introduced to deal with the frame and other problems.

This introduction to TAL is meant to be self contained, but by no means a complete feature list. A more detailed presentation of TAL is available elsewhere [8].

### 2.1 Basic Concepts

Most logics have an ontological base in properties and relations between objects. In TAL, properties and relations are modelled using *fluents* that assume *values*. Fluents are typed, so a given fluent only accepts arguments and only takes on values from the pre-specified *domains*. E.g., the location of a robot could be modelled using a boolean fluent *loc(robot,location)* that is associated with the value *true* or *false* for robot and location arguments from the domains *robot* and *location*. Alternatively one could use a *location*-valued fluent *loc(robot)* that has as value the object representing the location of the robot.

Fluents could be written as predicates in first-order logic. However, predicates can not have any values other than true or false and never change their values. In contrast, the value of the *loc* fluent can be a location that may change over time as the robot moves around. One could get around this problem by introducing a timepoint and a value argument to the predicate. However, by representing fluents as terms instead of predicates one also gains the possibility of quantifying over them. A special predicate *Holds(timepoint,fluent,value)* is introduced to associate a value with a fluent term at a specific *timepoint*. Timepoints are explicit objects in the form of numbers or terms that refer to a time-line consisting of positive integers starting with zero.

Suppose we are interested in robotic UAVs and would like to model their ability to move to new locations. The effects of an action fly(uav, location) would be formally specified and that specification could later be used to reason about the consequences of occurrences of this action at specific timepoints. An Occurs(timepoint, action) predicate is used for these purposes and represents an action occurring over the time interval between two timepoints.

Finally, the concept of *occlusion* is introduced as a very flexible solution to the frame and related problems. The basic idea is to make fluent values persist over time by minimizing their opportunities for change. A predicate *Occlude(timepoint,fluent)* represents the possibility of a fluent to change its value. Negated occlusion is then the property of a fluent retaining its value from the previous timepoint, axiomatized as:

$$\forall t \left[\neg Occlude(t+1, f) \rightarrow \forall v \left[Holds(t, f, v) \leftrightarrow Holds(t+1, f, v)\right]\right] \tag{1}$$

Formulas that model a reasoning domain must be written so that fluents are explicitly occluded in situations where they are known to change values, such as in the effects of an action. Instead of adding additional negated occlusion formulas one would like to minimize the number of timepoints where fluents are occluded to implement the default assumption that things do not change without a reason. This minimization is effected through circumscription of the *Occlude* predicate with respect to parts of the theory as detailed in Section 2.5. The result is a solution to the frame problem with fine grained control over which fluents are persistent at which timepoints.

### 2.2 A TAL Narrative

The TAL language thus includes three predicates, *Holds*, *Occurs*, and *Occlude* in an order-sorted (i.e. supporting a type hierarchy) first-order logic. An encoding of a limited domain of knowledge is called a TAL *narrative*. Consider a very simple reasoning domain involving the flight of a UAV to a new location, expressed by the following narrative:

$$\begin{aligned} Holds(0, \operatorname{loc}(\operatorname{uav1}), \operatorname{loc1}) & (2) \\ Occurs(3, 8, \operatorname{fly}(\operatorname{uav1}, \operatorname{loc2})) & (3) \\ \forall t_1, t_2, u, l \ [t_1 < t_2 \land Occurs(t_1, t_2, \operatorname{fly}(u, l)) \to & (4) \end{aligned}$$

$$Holds(t_2, loc(u), l) \land \forall t \ [t_1 < t \le t_2 \rightarrow Occlude(t, loc(u))]]$$

$$\forall t, u \left[\neg Occlude(t+1, \operatorname{loc}(u)) \rightarrow \\ \forall v \left[Holds(t, \operatorname{loc}(u), v) \leftrightarrow Holds(t+1, \operatorname{loc}(u), v)\right] \right]$$
(5)

Formulas (2) and (3) state that uav1 is located at loc1 and then flies to loc2 between timepoints 3 and 8. The action specification formula (4) occludes the location fluent from persistence during the flight and makes sure it assumes the correct value at the final timepoint of the occurrence interval. Finally, formula (5) instantiates the general persistence formula (1) for the *loc* fluent.

## 2.3 The High-level Language

The reification of fluents as terms enable the application of regular first-order logic to the problem of reasoning about actions and change over time. But formulas in the resulting theory must necessarily contain predicates and supporting axioms of a technical nature that are not present in any natural language description of the problem domain that one is trying to formalize. TAL provides a high-level narrative description language that helps circumvent this problem through a clear and concise syntax that lifts the abstraction level up above technical details. To differentiate this language from the base logic they are denoted  $\mathcal{L}(ND)$  (for Language of Narrative Descriptions) and  $\mathcal{L}(FL)$  (for Language of First-order Logic) respectively. The high-level language  $\mathcal{L}(ND)$ does not have a proof theory but is seen as a macro language that can be compiled into  $\mathcal{L}(FL)$  through a translation function *Trans.* Reasoning can then proceed using first-order proof techniques as described in Section 2.6.

Consider the same UAV flight domain introduced above but expressed as a narrative in the high-level language  $\mathcal{L}(ND)$ :

$$[0] \operatorname{loc}(\operatorname{uav1}) \stackrel{\circ}{=} \operatorname{loc1} \tag{6}$$

$$[3,8] fly(uav1, loc2) \tag{7}$$

$$[t_1, t_2] \text{ fly}(u, l) \rightsquigarrow R((t_1, t_2) \text{ loc}(u) = l)$$

$$\tag{8}$$

$$\forall t, u \left[ Per(t, \text{loc}(u)) \right] \tag{9}$$

Notice how fluents are associated with values using  $\hat{=}$ , as in the observation formula (6), and how temporal contexts are indicated simply by preceding formulas with timepoint or interval indications, as in the action occurrence (7). Formula (8) denotes through  $\rightsquigarrow$  and the reassignment macro R that the

fly action will cause the location of the UAV to be released from persistence during flight and settle on the destination when the flight is completed. The release from persistence is effected through occlusion of the *loc* fluent, denoted by  $X((t_1, t_2 | loc(u) = l))$ , but the domain modeller need not be concerned about occlusion since this mechanism is built into the reassignment macro. Finally, formula (9) will make the location fluent *loc* persistent by default using the macro *Per*.

In addition to the type of formulas illustrated above, a narrative may also include domain and dependency constraints that express conditions that always hold or dependencies that change the values of fluents when triggered by other fluent value changes rather than by an explicit action occurrence. These are powerful features, although we will not make further use of them in this thesis.

#### 2.3.1 The Translation Function

In its full generality, the syntax of  $\mathcal{L}(ND)$  and its translation into  $\mathcal{L}(FL)$  is given by the translation function *Trans*. The three  $\mathcal{L}(FL)$  predicates have a corresponding  $\mathcal{L}(ND)$  syntax given by:

$$\begin{aligned} Trans([t] \ f \doteq v) &\stackrel{\text{def}}{=} Holds(t, f, v) \\ Trans([t_1, t_2] \ a) &\stackrel{\text{def}}{=} Occurs(t_1, t_2, a) \\ Trans(X([t] \ f \doteq v)) &\stackrel{\text{def}}{=} Occlude(t, f) \end{aligned}$$

Action specifications translate into implications with the occurrence of the action during a non-unit time interval as the antecedent and the (possibly conditional and non-deterministic) effects in the conclusion:

$$Trans([t_1, t_2] \ a(\overline{x}) \rightsquigarrow \Phi) \stackrel{\text{\tiny def}}{=} \forall t_1, t_2, \overline{x} \ [t_1 < t_2 \land Occurs(t_1, t_2, a(\overline{x})) \to Trans(\Phi)]$$

The value of a fluent that is occluded at a timepoint is unknown without the presence of additional information. Persistent fluents are associated with persistency formulas, such as (5), that bind the fluent's value to the value at the previous timepoint when not occluded. A durational fluent, on the other hand, resumes a default value when not occluded. An example could be a fluent *noise* that is true while some loud action is being executed but reverts to false otherwise. Fluents are specified as persistent or durational using the *Per* and *Dur* macros:

$$\begin{aligned} Trans(Per(t,f)) &\stackrel{\text{def}}{=} \neg Occlude(t+1,f) \rightarrow \\ \forall v \left[ Holds(t,f,v) \leftrightarrow Holds(t+1,f,v) \right] \\ Trans(Dur(t,f,v)) &\stackrel{\text{def}}{=} \neg Occlude(t,f) \rightarrow Holds(t,f,v) \end{aligned}$$

The reassignment operator R and interval reassignment operator I ensure both that the fluent assumes the new value and that it is released from any persistence assumptions using the occlusion operator X:

$$Trans(R((t_1, t_2] \Phi)) \stackrel{\text{def}}{=} Trans(X((t_1, t_2] \Phi)) \land Trans([t_2] \Phi)$$
  

$$Trans(R([t] \Phi)) \stackrel{\text{def}}{=} Trans(X([t] \Phi)) \land Trans([t] \Phi)$$
  

$$Trans(I((t_1, t_2] \Phi)) \stackrel{\text{def}}{=} Trans(X((t_1, t_2] \Phi)) \land Trans((t_1, t_2] \Phi)$$
  

$$Trans(I([t] \Phi)) \stackrel{\text{def}}{=} Trans(X([t] \Phi)) \land Trans([t] \Phi)$$

Temporal intervals can be open or closed in either end, may be infinite, and can occur inside the occlusion operator X. Their translations are illustrated by some representative examples:

$$\begin{aligned} Trans([t_1, \infty) \Phi) &\stackrel{\text{def}}{=} \forall t \ [t_1 \le t \to Trans([t] \Phi)] \\ Trans((t_1, \infty) \Phi) &\stackrel{\text{def}}{=} \forall t \ [t_1 < t \to Trans([t] \Phi)] \\ Trans([t_1, t_2] \Phi) &\stackrel{\text{def}}{=} \forall t \ [t_1 \le t \le t_2 \to Trans([t] \Phi)] \\ Trans(X((t_1, t_2] \Phi)) &\stackrel{\text{def}}{=} \forall t \ [t_1 < t \le t_2 \to Trans(X([t] \Phi))] \end{aligned}$$

The translation function ignores connectives and quantifiers, and temporal contexts are distributed over connectives:

$$\begin{aligned} Trans(\neg \Phi) &\stackrel{\text{def}}{=} \neg Trans(\Phi) \\ Trans(\Phi \ C \ \Psi) \stackrel{\text{def}}{=} Trans(\Phi) \ C \ Trans(\Psi) \text{ where } C \in \{\land, \lor, \rightarrow, \leftrightarrow\} \\ Trans(Qv[\Phi]) \stackrel{\text{def}}{=} Qv[Trans(\Phi)] \text{ where } Q \in \{\forall, \exists\} \\ Trans([t] \neg \Phi) \stackrel{\text{def}}{=} \neg Trans([t] \ \Phi) \\ Trans([t] \Phi \ C \ \Psi) \stackrel{\text{def}}{=} Trans([t] \Phi) \ C \ Trans([t] \Psi) \text{ where } C \in \{\land, \lor, \rightarrow, \leftrightarrow\} \\ Trans([t] Qv[\Phi]) \stackrel{\text{def}}{=} Qv[Trans([t] \Phi)] \text{ where } Q \in \{\forall, \exists\} \end{aligned}$$

However, the occlusion operator X needs special attention. For the circumscription of *Occlude* to be computable, as described in Section 2.6, all the fluents that might be influenced by an action must be occluded. This is accomplished by removing negations and replacing disjunctions and existential quantifiers by conjunctions and universal quantifiers:

$$\begin{aligned} Trans(X([t] \neg \Phi)) &\stackrel{\text{def}}{=} Trans(X([t] \Phi)) \\ Trans(X([t] \Phi C \Psi)) \stackrel{\text{def}}{=} Trans([t] \Phi) \land Trans([t] \Psi) \text{ where } C \in \{\land, \lor, \rightarrow, \leftrightarrow\} \\ Trans(X([t] Qv[\Phi])) \stackrel{\text{def}}{=} \forall v \left[ Trans(X([t] \Phi)) \right] \text{ where } Q \in \{\forall, \exists\} \end{aligned}$$

. .

Finally, some common expressions have shorthand notation. If f is a boolean fluent then f and  $\neg f$  are short for  $f \stackrel{c}{=} true$  and  $f \stackrel{c}{=} false$ , and when several timepoints are ordered in a series  $t_1 < t_2 < t_3$ , it is short for a conjunction  $t_1 < t_2 \land t_2 < t_3$ .

## 2.4 Foundational Axioms

Some intuitions about narratives need to be expressed as additional axioms before the domain of interest can be considered formalized. Specifically, these include unique names assumptions, unique value assumptions, and the meaning of functions and relations on timepoints.

When two different objects, such as loc1 and loc2, are used one most probably assumes that they are different, i.e. that the terms satisfy  $loc1 \neq loc2$ . These assumptions have to be made explicit in first-order logic using additional unique names axioms. Since the logic is order-sorted one may still assume that two objects of different types are not the same object, but for any pair of object terms o and o' of the same domain the following axiom is added:

 $o \neq o'$ 

The same assumption of uniqueness holds for fluents although two sets of formulas are needed. One with each pair of different fluents f and f', and one where, for each fluent, one or more of its arguments are different:

$$\forall \overline{x}, \overline{y} \left[ f(\overline{x}) \neq f'(\overline{y}) \right]$$
  
$$\forall x_1, \dots, x_n, y_1, \dots, y_n \left[ x_1 \neq y_1 \lor \dots \lor x_n \neq y_n \rightarrow f(x_1, \dots, x_n) \neq f(y_1, \dots, y_n) \right]$$

Similar unique names axioms are added for action terms:

$$\forall \overline{x}, \overline{y} [a(\overline{x}) \neq a'(\overline{y})]$$
  
 
$$\forall x_1, \dots, x_n, y_1, \dots, y_n [x_1 \neq y_1 \lor \dots \lor x_n \neq y_n \rightarrow a(x_1, \dots, x_n) \neq a(y_1, \dots, y_n)]$$

Unique value axioms, two for each fluent, state that the fluent holds one, and exactly one, value at each timepoint:

$$\forall t, \overline{x} \left[ \exists v \left[ Holds(t, f(\overline{x}), v) \right] \right] \\ \forall t, \overline{x}, v_1, v_2 \left[ v_1 \neq v_2 \rightarrow \neg (Holds(t, f(\overline{x}), v_1) \land Holds(t, f(\overline{x}), v_2)) \right]$$

Finally, some of the formulas we have used include the addition function + and the "less than" relation < and "less than or equal" relation  $\leq$  on integer timepoints. The intention is for them to behave like the mathematical concepts with corresponding notation. It is possible to include an axiomatized theory for this purpose, such as Presburger arithmetic [39], and continue using regular theorem proving with the additional axioms. However, from a practical viewpoint, a more realistic approach is the use of a specialized but incomplete reasoning system, e.g. some variant of a constraint solver as discussed in Section 3.2.1 and carried out in Chapter 4.

## 2.5 Circumscription Policy

Value changes of fluents that have been specified as persistent are minimized through circumscription, in effect implementing a default assumption of persistence unless an exception was explicitly stated using the *Occlude* predicate. A circumscription policy specifies what predicates to minimize in which formulas, thereby making sure the expected behaviour is obtained. To specify the policy we need to be able to refer to the different sets of formulas that a narrative consists of. For this purpose, let  $\Gamma_{\rm fnd}$  denote the foundational axioms,  $\Gamma_{\rm per}$ persistence formulas,  $\Gamma_{\rm obs}$  fluent value observations,  $\Gamma_{\rm domc}$  domain constraints,  $\Gamma_{\rm occ}$  action occurrences,  $\Gamma_{\rm depc}$  dependency constraints, and  $\Gamma_{\rm acs}$  denote action specifications. The TAL circumscription policy is then given by:

 $\Gamma_{\rm fnd} \wedge \Gamma_{\rm per} \wedge \Gamma_{\rm obs} \wedge \Gamma_{\rm domc} \wedge CIRC[\Gamma_{\rm occ}; Occurs] \wedge CIRC[\Gamma_{\rm depc} \wedge \Gamma_{\rm acs}; Occlude]$ 

The *Occurs* predicate is minimized to remove the possibility of unknown action occurrences. Since action effects entail occlusion they would otherwise prevent proofs of persistence. Note also that persistence formulas ( $\Gamma_{per}$ ) are excluded from the minimization of *Occlude* even though they contain the predicate. The contrapositive of the persistence formula (1) illustrates the reason:

$$\forall t \left[ \neg \forall v \left[ Holds(t, f, v) \leftrightarrow Holds(t+1, f, v) \right] \rightarrow Occlude(t+1, f) \right]$$
(10)

If (1), which is equivalent to (10), had been included, a fluent value change would have been reason in itself for the occlusion of a fluent, and the minimization of Occlude would not have been effective.

## 2.6 Reasoning in TAL

The goal of the development of Temporal Action Logic is not just to formalize reasoning domains, but also to make it possible for an intelligent agent to use the logic to perform practical reasoning tasks. The availability of reasoning techniques is therefore of great importance, and it is no coincidence that firstorder logic was chosen for the base language. Nevertheless, after narratives written in the high-level language  $\mathcal{L}(ND)$  have been compiled into the firstorder language  $\mathcal{L}(FL)$ , the circumscription policy is applied and the resulting theory is second-order, where no complete proof methods are possible. By enforcing certain syntactic restrictions on the  $\mathcal{L}(ND)$  formulas one can show that the circumscribed predicates *Occurs* and *Occlude* occur only positively in the relevant parts of the theory,  $\Gamma_{occ}$  and  $\Gamma_{depc} \wedge \Gamma_{acs}$  respectively [8]. Circumscription then reduces to predicate completion [22], which can be computed with relative ease. Any first-order proof method of choice can then be applied.

#### 2.6.1 Natural Deduction

A natural deduction proof of a simple consequence of the UAV narrative introduced in Section 2.2 will serve to illustrate reasoning in TAL. The notation used is that of Copi [31], where each row is made up of a row number (starting from 11 to avoid confusing the row numbers with numbered formulas in previous sections of this chapter), the deduced formula, and a justification of the deduction. We will not explicitly list the natural deduction rules used in the justification but instead assume proof steps that are sufficiently small so as to only require justification in the form of references to the proof rows that the deduction depends on, or a letter indicating a premise P, hypothesis H, tautology T, or a direct consequence of the TAL foundational axioms F. The scope and nesting of hypotheses are indicated by vertical lines in the left margin.

Assume that we would like to prove e.g. that uav1 will be at loc2 at timepoint 9. The proof begins by predicate completion of *Occurs* in the action occurrence formula subset  $\Gamma_{occ}$  of the narrative (simply formula (3) above) and of *Occlude* in the action specification and dependency constraint formula subset  $\Gamma_{depc} \wedge \Gamma_{acs}$  of the narrative (simply formula (4) above), according to the circumscription policy. The resulting premises are:

11 
$$Holds(0, loc(uav1), loc1)$$
 P

12 
$$\forall t_1, t_2, a \ [t_1 = 3 \land t_2 = 8 \land a = \text{fly}(\text{uav}1, \text{loc}2) \leftrightarrow Occurs(t_1, t_2, a)]$$
 P

13 
$$\forall t_1, t_2, u, l [t_1 < t_2 \land Occurs(t_1, t_2, fly(u, l)) \rightarrow Holds(t_2, loc(u), l)]$$
 F

 $14 \quad \forall t, u \left[ \exists t_1, t_2, l \left[ t_1 < t \leq t_2 \land Occurs(t_1, t_2, \mathrm{fly}(u, l)) \right] \leftrightarrow Occlude(t, \mathrm{loc}(u)) \right] P$ 

$$\begin{array}{ll} 15 \quad \forall t, u \left[ \neg Occlude(t+1, \operatorname{loc}(u)) \rightarrow \\ \forall v \left[ Holds(t, \operatorname{loc}(u), v) \leftrightarrow Holds(t+1, \operatorname{loc}(u), v) \right] \right] \end{array} \qquad \qquad P$$

Note that the action specification formula gave rise to both an action effect formula and the occlusion definition in premises (13) and (14), and that the foundational axioms are not present in this listing.

We proceed to show that the fly action takes the UAV to loc2 between timepoints 3 and 8:

$$16 \quad 3 < 8 \land Occurs(3, 8, fly(uav1, loc2)) \rightarrow Holds(8, loc(uav1), loc2)$$

17 
$$3 < 8$$
  
18  $3 = 3 \land 8 = 8 \land \text{fly}(\text{uav1}, \text{loc2}) = \text{fly}(\text{uav1}, \text{loc2}) \leftrightarrow Occurs(3, 8, \text{fly}(\text{uav1}, \text{loc2}))$ 

19 
$$3 = 3 \land 8 = 8 \land \text{fly}(\text{uav1}, \text{loc2}) = \text{fly}(\text{uav1}, \text{loc2})$$
 T

20 Occurs(3, 8, fly(uav1, loc2))18, 19

21 
$$Holds(8, loc(uav1), loc2)$$
 16, 17, 20

The minimization of Occlude has made it possible to prove non-occlusion of the *loc* fluent for timepoints outside the fly action occurrence interval. This fact, together with the persistence formula (15), is required to propagate the fluent's value loc2 from timepoint 8 to timepoint 9:

$$22 \neg Occlude(8+1, loc(uav1)) \rightarrow \\Holds(8, loc(uav1), loc2) \leftrightarrow Holds(8+1, loc(uav1), loc2)$$
15

23 
$$8+1=9$$

$$\begin{array}{ll} 25 & \neg \exists t_1, t_2, l \left[ t_1 < 9 \leq t_2 \land Occurs(t_1, t_2, \operatorname{fly}(\operatorname{uav} 1, l)) \right] \leftrightarrow \\ \neg Occlude(9, \operatorname{loc}(\operatorname{uav} 1)) & 14 \\ 26 & [\neg \exists t_1, t_2, l \left[ t_1 < 9 \leq t_2 \land Occurs(t_1, t_2, \operatorname{fly}(\operatorname{uav} 1, l)) \right] & H \end{array}$$

$$\begin{array}{c|c} 27 \\ 28 \\ 28 \\ 29 \\ 30 \\ \end{array} \left[ \begin{array}{c} -t_1 < 9 \le t_2 \land Occurs(t_1, t_2, \mathrm{fly}(\mathrm{uav1}, l)) & H \\ t_1 = 3 \land t_2 = 8 \land \mathrm{fly}(\mathrm{uav1}, l) = \mathrm{fly}(\mathrm{uav1}, \mathrm{loc2}) & 12, 27 \\ 3 < 9 \le 8 & 27, 28 \\ false & 29, F \end{array} \right]$$

31 
$$false$$
 26,27 - 30

$$32 \quad \neg \exists t_1, t_2, l \, [t_1 < 9 \le t_2 \land Occurs(t_1, t_2, fly(uav1, l))] \\ 33 \quad \neg Occlude(9, loc(uav1)) \\ 25, 32$$

$$33 \neg Occlude(9, loc(uav1))$$

F12

F

34	$Holds(8, loc(uav1), loc2) \leftrightarrow Holds(9, loc(uav1), loc2)$	24, 33
35	Holds(9, loc(uav1), loc2)	21, 34

#### 2.6.2 Automated Reasoning

Beside handcrafted proofs, a whole host of automated theorem proving and model theoretic tools are in principle applicable to the first-order narrative. Unfortunately it is far from self-evident that the direct application of theorem provers provides reasoning capabilities that are sufficiently efficient to be of practical use. Especially not when an axiomatization of TAL's integer time structure is used.

In practice we most often rely on a model theoretic tool for evaluating TAL formulas and visualizing their models developed by Kvarnström called VITAL [18]. The tool, which is available online, provides excellent opportunities for experimentation with formalizing reasoning problems as TAL narratives.

This thesis explores a third option, the use of logic programming and constraint solving technology. By introducing further restrictions on  $\mathcal{L}(ND)$  formulas it is possible to define a compilation into logic programs that can be efficiently executed by Prolog. Furthermore, the close integration between Prolog and constraint solvers, in the constraint logic programming paradigm, provides a good fit to the TAL logic with its explicit time structures. This topic is explored in Chapter 4 and Chapter 5.

## Chapter 3

# **Deductive Planning**

Deductive planning requires modifications to the Temporal Action Logic presented in Chapter 2. When action occurrences are specified using the first-order *predicate Occurs* there is no mechanism that allows the addition of new occurrences through a deductive proof process. Hypothesizing new instances of this predicate given a goal would require abductive techniques or the use of some special-purpose planning algorithm such as that employed by TALplanner [7]. This chapter introduces a *deductive* mechanism through the reification of action occurrences as terms in the language that can be reasoned with and quantified over. In addition, TAL is extended with the concept of interval occlusion that makes the use of temporal constraint formalisms easier. Constraint solvers eliminate the need for an axiomatization of the integer timeline for the types of temporal reasoning needed in our applications to deductive planning. A hand crafted proof illustrates how reified actions and interval occlusion make planning possible, although still not practical.

## 3.1 Reified Action Occurrences

A deductive planning goal  $\Phi$  is usually expressed as the existence of a plan that satisfies  $\Phi$ . In TAL we would add an explicit timepoint at which the goal is satisfied:

 $\exists t, p \left[ \Phi(t, p) \right]$ 

While the conclusion in proving the above is the *existance* of a plan, the actual plan can be retrieved from that proof. However, since plans are composed of action occurrences that are denoted using the TAL *predicate Occurs*, there is

no way to specify a (first-order) existential quantifier over plans in TAL. This problem can be resolved by reifying action occurrences as terms, which can be quantified over. Doing so necessitates the introduction of a *Member* predicate, to group action occurrence terms into sets, or plans, and the addition of a new action occurrence set argument to the other two TAL predicates *Holds* and *Occlude*, so that we can talk about e.g. a fluent's value in the context of a specific plan.

We start by introducing a function occ that replaces action occurrences  $Occurs(t_1, t_2, a)$  by terms of the form  $occ(t_1, t_2, a)$ . A set of action occurrences is a set of such terms. The temporal ordering of the actions themselves is given by their relations to the explicit time line, determined by the  $t_1$  and  $t_2$  timepoint arguments. In consequence, the order of the action occurrence terms in the set is not important, which means that its behaviour is just like a regular mathematical set.

Additionally, a vital property is incompleteness of the action occurrence set. If a reasoning problem involves a fully specified set of action occurrences, then the problem corresponds to prediction or postdiction of the effects of executing the actions or the conditions that hold when executing those actions. If a reasoning problem involves an under-specified (or empty) set of action occurrences, then the problem often involves the addition of new action occurrences to the occurrence set to satisfy goal constraints that were expressed as part of the problem. This constitutes a process of deductive planning.

We introduce a composition function term cons(a, p) to add an individual action occurrence term a to an action occurrence set (or plan) p, and the term *nil* to represent the empty set. A predicate *Member* is associated with a simple axiomatization of set membership:

 $\begin{aligned} &\forall x \left[ \neg Member(x, nil) \right] \\ &\forall x, y, s \left[ Member(x, cons(y, s)) \leftrightarrow x = y \lor Member(x, s) \right] \end{aligned}$ 

A new *Occurs* predicate with an additional action occurrence set argument is defined as:

$$\forall t_1, t_2, a, p \left[ Occurs(t_1, t_2, a, p) \leftrightarrow Member(occ(t_1, t_2, a), p) \right]$$

Lastly, an action occurrence argument is added to the other two  $\mathcal{L}(FL)$  predicates *Holds* and *Occlude*. The meaning of a narrative is preserved by introducing a new universally quantified action occurrence set variable p in each of the narrative formulas. E.g., the *fly* action specification formula from the example narrative in Chapter 2, repeated here:

$$\begin{array}{l} \forall t_1, t_2, u, l \; [t_1 < t_2 \land \mathit{Occurs}(t_1, t_2, \mathrm{fly}(u, l)) \rightarrow \\ \mathit{Holds}(t_2, \mathrm{loc}(u), l) \land \forall t \; [t_1 < t \leq t_2 \rightarrow \mathit{Occlude}(t, \mathrm{loc}(u))]] \end{array}$$

would receive an additional universally quantified action occurrence set argument p simply by adding it to each of the  $\mathcal{L}(FL)$  predicates:

$$\forall t_1, t_2, u, l, p \left[ t_1 < t_2 \land Occurs(t_1, t_2, fly(u, l), p) \rightarrow \\ Holds(t_2, loc(u), l, p) \land \forall t \left[ t_1 < t \le t_2 \rightarrow Occlude(t, loc(u), p) \right] \right]$$

Using the same variable p throughout the formula ensures that dependencies on action occurrences are preserved just as if the occurrences had been specified separately using the old *Occurs* predicate. Then, by universal quantification over p, the formula is made applicable for any set of action occurrences that one might want to reason about.

### 3.2 Interval Occlusion

The TAL Occlude predicate determines at which timepoints a fluent may change value. Negated occlusion is, accordingly, the property of a fluent not being allowed to change, i.e. the property of persistence. Proofs of persistence between two timepoints are very frequent in the context of planning and the most straightforward way of proving persistence involves deducing  $\neg Occlude(t_i, f)$  for the fluent of interest f and for all intermediate timepoints  $t_i$ . A fluent persistency formula is then applied to propagate the fluent's value from one timepoint to the other.

However, when the number of intermediate timepoints is not known the simple proof sketched above is not applicable. This is the case e.g. when only qualitative timepoints are used, or when a least commitment strategy concerning the timepoints and orderings of action occurrences is strived for. In such cases it is more useful to look at all the intermediate timepoints as an interval and persistence over all the timepoints as *interval persistence*. As a first step in this direction we introduce the *interval occlusion* predicate  $Occlude(t_1, t_2, f)$  (or  $Occlude(t_1, t_2, f, p)$  if action occurrences are reified). The intended meaning is that fluent f is interval occluded over  $(t_1, t_2]$  iff it is occluded at some timepoint in that interval. Conversely, if we manage to prove that fluent f is *not* interval occlusion in terms of the regular timepoint occlusion as:

$$\forall t_1, t_2, f \left[ Occlude(t_1, t_2, f) \leftrightarrow \exists t \left[ t_1 < t \le t_2 \land Occlude(t, f) \right] \right] \tag{1}$$

In Appendix B.1 we prove that (1) together with the persistence formula for timepoint occlusion entails an interval persistence formula:

$$\forall t_1, t_2, f \left[\neg Occlude(t_1, t_2, f) \rightarrow \\ \forall t \left[t_1 < t \le t_2 \rightarrow \forall v \left[Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)\right]\right] \right]$$

$$(2)$$

Note that even if a fluent is interval occluded over a given interval, it is not necessarily occluded in all sub-intervals. However, if the fluent is interval persistent over the interval it must also be persistent in all sub-intervals. As shown in Appendix B.2, formulas (1) and (2) entail the following formula:

$$\forall t_1, t_2, f \left[\neg Occlude(t_1, t_2, f) \rightarrow \forall v \left[Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v)\right]\right]$$
(3)

Using (3), the truth value of a fluent can be made to "jump" any number of timepoints in a single proof step. This technique is essential in the translation to constraint logic programs described in Chapter 4.

#### 3.2.1 Temporal Constraint Formalisms

Another essential step towards efficient proof procedures for TAL in the context of only qualitative and partial knowledge of temporal relations is the introduction of temporal constraint propagation techniques. While the Holds(t, f, v)predicate is defined on timepoints, both the  $Occurs(t_1, t_2, a)$  and the new  $Occlude(t_1, t_2, f)$  predicates are defined over intervals characterized by their end-points  $t_1$  and  $t_2$ . Furthermore, considering persistence over entire intervals was found to result in conveniently simple persistence proofs. These facts suggest that the well-known interval algebra formalism introduced by Allen [1], which adopts the interval as a primitive temporal object, might be useful. Allen shows that thirteen primitive relations exhaust all possible qualitative relations between two intervals, and the algebra is complete for reasoning with them, even in the context of incomplete information. Allen also notes that intervals can be represented by their end-points and primitive relations by ordering conditions on these end-points.

Thornton et. al. [38] go on to show how their *end-point ordering* model can be used to express the qualitative interval algebra in a quantitative constraint solver with finite domains. They recognize that all solutions to an interval algebra problem must be possible to express by different integer instantiations of the end-points that satisfy the ordering conditions. Furthermore, if there are m intervals in the problem, 2m is an upper bound on the number of different integers needed. Hence, an interval algebra network can be encoded as a finite domain constraint satisfaction problem where variables represent interval endpoints and belong to the range  $1 \dots 2m$ .

We have chosen to work with a finite domain constraint solver and the end-point ordering encoding of the interval algebra, and Chapter 4 provides details on how this is used to implement temporal relations in TAL narratives expressed as logic programs. Although, other temporal constraint formalisms than the interval algebra are certainly possible. We have in fact experimented with general temporal constraint networks [28], a unifying framework that provides both intervals and timepoints as primitives and both qualitative and quantitative relations between them. Such networks are also associated with complete reasoning procedures of the same (exponential) computational complexity. However, the simpler interval algebra end-point ordering model uses only the  $\langle , =,$  and  $\rangle$  relations between timepoints, requiring no extensions of TAL. General temporal constraint networks support more complex relations whose description in TAL require some form of semantic attachment. Moreover, the upper bound on the domains of timepoint variables, required when working with the finite domain constraint solver, enforces an upper bound on the number of actions in action occurrence sets. This follows from the fact that each action occurs over an interval that is added to the interval algebra graph. Such a bound prevents infinite sequences of actions that can in certain domains keep the planner busy, preventing it from finding a solution. Completeness can be achieved through an iterative deepening of this bound. In practice, the finite domain constraint solver implementation is also considerably faster.

## 3.3 A Constructive Plan Existence Proof

The above extensions realize planning as deduction. Consider, e.g., the following narrative in  $\mathcal{L}(FL)$  that extends the narrative in Section 2.2 towards a logistics scenario for UAVs. A location-valued fluent loc(object) and a boolean fluent *carrying(uav,crate)* are used to represent the location of an object (either a UAV or a crate) and the fact that a UAV is carrying (or not) a crate respectively. Two actions, fly(uav, location) and attach(uav, crate), can be used for flying a UAV to a location and to instruct a UAV to attach its winch to a crate. For the purpose of this example, the narrative formulas make use of interval occlusion. A principled way of introducing interval occlusion in a narrative expressed using regular occlusion is detailed in Section 4.3.

Listing the narrative formulas as premises in the natural deduction style notation introduced in Section 2.6.1 results in the following rows:

$$4 \quad \forall x \left[\neg Member(x, nil)\right] \qquad \qquad P$$

5 
$$\forall x, y, s [Member(x, cons(y, s)) \leftrightarrow x = y \lor Member(x, s)]$$
 P

$$6 \quad \forall t_1, t_2, a, p \left[ Occurs(t_1, t_2, a, p) \leftrightarrow Member(occ(t_1, t_2, a), p) \right] \qquad P$$

$$\begin{array}{l} 7 \quad \forall t_1, t_2, o, p \; [t_1 < t_2 \land \neg Occlude(t_1, t_2, \operatorname{loc}(o), p) \rightarrow \\ \forall v \; [Holds(t_1, \operatorname{loc}(o), v, p) \leftrightarrow Holds(t_2, \operatorname{loc}(o), v, p)] \end{array} \qquad \qquad P \end{array}$$

$$\begin{array}{lll} 8 & \forall t_1, t_2, u, c, p \left[ t_1 < t_2 \land \neg Occlude(t_1, t_2, \operatorname{carrying}(u, c), p) \rightarrow \\ & \forall v \left[ Holds(t_1, \operatorname{carrying}(u, c), v, p) \leftrightarrow Holds(t_2, \operatorname{carrying}(u, c), v, p) \right] \right] P \\ 9 & \forall t_1, t_2, u, l, p \left[ t_1 < t_2 \land Occurs(t_1, t_2, \operatorname{fly}(u, l), p) \rightarrow Holds(t_2, \operatorname{loc}(u), l, p) \right] P \\ 10 & \forall t_1, t_2, u, c, l, p \left[ t_1 < t_2 \land Occurs(t_1, t_2, \operatorname{attach}(u, c), p) \land \\ & Holds(t_1, \operatorname{loc}(c), l, p) \land Holds(t_1, \operatorname{loc}(u), l, p) \rightarrow \\ & Holds(t_2, \operatorname{carrying}(u, c), true, p) \land \neg Holds(t_2, \operatorname{loc}(c), l, p) \right] P \\ 11 & \forall t_1, t_2, t_3, t_4, u, c, l, p \left[ t_2 > t_3 \land t_1 < t_4 \land Occurs(t_3, t_4, \operatorname{attach}(u, c), p) \land \\ & Holds(t_3, \operatorname{loc}(c), l, p) \land Holds(t_3, \operatorname{loc}(u), l, p) \leftrightarrow \\ & Occlude(t_1, t_2, \operatorname{carrying}(u, c), p) \right] P \\ 12 & \forall t_1, t_2, t_3, t_4, u, c, l, p \left[ t_2 > t_3 \land t_1 < t_4 \land Occurs(t_3, t_4, \operatorname{attach}(u, c), p) \land \\ & Holds(t_3, \operatorname{loc}(c), l, p) \land Holds(t_3, \operatorname{loc}(u), l, p) \leftrightarrow \\ & Occlude(t_1, t_2, \operatorname{loc}(c), p) \right] P \\ 13 & \forall p \left[ Holds(0, \operatorname{loc}(\operatorname{uav1}), \operatorname{base}, p) \right] P \\ 14 & \forall p \left[ Holds(0, \operatorname{loc}(\operatorname{crate1}), \operatorname{loc1}, p) \right] P \\ 15 & \forall u, c, p \left[ Holds(0, \operatorname{carrying}(u, c), \operatorname{false}, p) \right] P \\ \end{array}$$

Given this narrative one would like to find constructive proofs of plan existence queries. As an example, it should be possible to prove that there exists some set of action occurrences p (a solution plan) that results in *uav1* carrying *crate1* at some future timepoint t. The main goal of the proof can then be expressed as the existence of such a timepoint and plan for which the goal holds:

 $\exists t, p [Holds(t, carrying(uav1, crate1), true, p)]$ 

The exact value of timepoint t can be unspecified as long as the actions in p will fit between timepoints 0 and t. A single intermediate timepoint  $t_k$  suffices for the purposes of this proof:

16 
$$0 < t_k < t$$
 P

One way of proving the main goal is by instantiating the *attach* action specification with *uav1*, *crate1*, and *t*:

$$\begin{array}{ll} 17 \quad \forall t_1, l, p \ [t_1 < t \land Occurs(t_1, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1}), p) \land \\ Holds(t_1, \operatorname{loc}(\operatorname{crate1}), l, p) \land Holds(t_1, \operatorname{loc}(\operatorname{uav1}), l, p) \rightarrow \\ Holds(t, \operatorname{carrying}(\operatorname{uav1}, \operatorname{crate1}), true, p)] \end{array}$$

If the implication antecedent of Row 17 can be proven, the goal will follow from Modus Ponens. Call the four antecedent conjuncts Subgoal 1 - 4. Subgoal 1 can be satisfied immediately:

18 
$$t_k < t$$
 16

Subgoal 2 requires the use of the definition of *Occurs*. This creates an action occurrence set that includes an occurrence of the *attach* action and a new unknown plan tail  $p_1$ :

- $\begin{array}{ll} 19 & occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})) = occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})) \lor \\ & Member(occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})),p_1) \to \\ & Member(occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})), \\ & cons(occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})),p_1)) & 5 \\ 20 & occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})) = occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})) & T \\ 21 & Member(occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})), \\ & cons(occ(t_k,t,\operatorname{attach}(\operatorname{uav1},\operatorname{crate1})), p_1)) & 19, 20 \\ \end{array}$
- 22  $Occurs(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1}), cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1))$  6,21
- 23  $\forall p_1 \left[ Occurs(t_k, t, \text{attach}(uav1, crate1),$  $cons(occ(t_k, t, \text{attach}(uav1, crate1)), p_1)) \right]$  22

One way of satisfying Subgoal 3,  $Holds(t_1, loc(crate1), l, p)$ , would be to use Row 14 that specifies the location of *crate1* in the initial state. But  $t_1$  would then be instantiated to 0, which would leave the UAV no time to fly there and satisfy Subgoal 4. Instead, we use the persistence of the *loc* fluent to show that *crate1* remains at *loc1* from 0 to an intermediate timepoint  $t_k$ . An automated theorem prover, without foresight, would try both the initial state fact and the persistence formula, either simultaneously, or through backtracking. The persistence formula in Row 7 is reformulated as an implication:

$$\begin{array}{ll} 24 & \forall v, p_1 \left[ 0 < t_k \land \\ \neg \mathit{Occlude}(0, t_k, \operatorname{loc}(\operatorname{crate1}), \\ \mathit{cons}(\mathit{occ}(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \land \\ Holds(0, \operatorname{loc}(\operatorname{crate1}), v, \mathit{cons}(\mathit{occ}(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \rightarrow \\ Holds(t_k, \operatorname{loc}(\operatorname{crate1}), v, \mathit{cons}(\mathit{occ}(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \right] \end{array}$$

If the three antecedent conjuncts of Row 24 are referred to as Subgoal 3.1 - 3.3, then Subgoal 3.1 is given by:

25 
$$0 < t_k$$
 16

In order to succeed with the proof of non-occlusion in Subgoal 3.2 it is necessary to close the action occurrence argument  $p_1$  with respect to any unknown *attach* actions. The TAL circumscription policy usually closes action occurrences through circumscription of the *Occurs* predicate. The same effect can be achieved with reified action occurrences by recording assumptions made about the plan argument p. Such assumptions can be proven after instantiating unbound plan variables to *nil* when no new occurrences need to be added to the final plan. The proof of Subgoal 3.2 records the assumption that  $p_1$  contains no occurrences of *attach* in Row 26. Non-occlusion of the *loc* fluent can be proved using its definition from Row 12. Assuming that the conditions for occlusion are satisfied, in Row 28, leads to a contradiction. It would either have meant that the persistence interval  $(0, t_k]$  overlaps the *attach* action interval  $(t_k, t]$ , disproved in Row 29 – Row 34, or that there are some other *attach* actions, which we assumed false as shown by Row 35 – Row 37:

40 
$$\neg Occlude(0, t_k, loc(crate1), cons(occ(t_k, t, attach(uav1, crate1)), p_1))$$
 27,39

$$\begin{array}{l} 41 \quad \neg \exists t_1, t_2, u, c \left[ Member(occ(t_1, t_2, \operatorname{attach}(u, c)), p_1) \right] \rightarrow \\ \neg Occlude(0, t_k, \operatorname{loc}(\operatorname{crate1}), \\ cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \end{array}$$

$$42 \quad \forall p_1 \left[ \neg \exists t_1, t_2, u, c \left[ Member(occ(t_1, t_2, \operatorname{attach}(u, c)), p_1) \right] \rightarrow \\ \neg Occlude(0, t_k, \operatorname{loc}(\operatorname{crate1}), \\ cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \right]$$

Knowledge about the initial state in Row 14 is then used to satisfy Subgoal 3.3 with v instantiated to loc1:

43 
$$\forall p_1 [Holds(0, loc(crate1), loc1, cons(occ(t_k, t, attach(uav1, crate1)), p_1))]$$
 14

Subgoal 3 can now be completed, although it is still qualified by the assumption on occurrences of *attach*:

$$\begin{array}{c} 44 & \neg \exists t_1, t_2, u, c \left[ Member(occ(t_1, t_2, \operatorname{attach}(u, c)), p_1) \right] & H \\ 45 & Holds(t_k, \operatorname{loc}(\operatorname{crate1}), true, \\ & cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) & 24, 25, 42, 44, 43 \end{array}$$

$$46 \quad \forall p_1 \left[ \neg \exists t_1, t_2, u, c \left[ Member(occ(t_1, t_2, \operatorname{attach}(u, c)), p_1) \right] \rightarrow \\ Holds(t_k, \operatorname{loc}(\operatorname{crate1}), \operatorname{loc1}, \\ cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1})), p_1)) \right] \qquad \qquad 44-45$$

The *fly* action specification in Row 9 can be used to prove Subgoal 4. The definition of *Occurs* is used to add a *fly* action to the plan, resulting in a new plan tail  $p_2$ . The new action causes the UAV to end up at the location of the crate, thereby satisfying the last condition for attaching it:

$$\begin{array}{rl} 47 \quad 0 < t_k \land Occurs(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1}), \\ & cons(occ(t_k, t, \mathrm{attach}(\mathrm{uav1}, \mathrm{crate1})), \\ & cons(occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2))) \rightarrow \\ & Holds(t_k, \mathrm{loc}(\mathrm{uav1}), \mathrm{loc1}, cons(occ(t_k, t, \mathrm{attach}(\mathrm{uav1}, \mathrm{crate1})), \\ & cons(occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2))) & 9 \\ \\ 48 \quad occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})) = occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})) \lor \\ & Member(occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2) \rightarrow \\ & Member(occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), cons(occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2)) & 5 \\ \\ \\ 49 \quad occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})) = occ(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})) & T \end{array}$$

50 
$$Member(occ(0, t_k, fly(uav1, loc1))) = bcc(0, t_k, fly(uav1, loc1))$$

$$cons(occ(0, t_k, fly(uav1, loc1)), p_2))$$

$$48, 49$$

4

51	$occ(0, t_k, fly(uav1, loc1)) = occ(t_k, t, attach(uav1, crate1)) \lor$	
	$Member(occ(0, t_k, fly(uav1, loc1))),$	
	$cons(occ(0, t_k, fly(uav1, loc1)), p_2)) \rightarrow$	
	$Member(occ(0, t_k, fly(uav1, loc1))),$	
	$cons(occ(t_k, t, \operatorname{attach}(\operatorname{uav1}, \operatorname{crate1}))),$	
	$cons(occ(0, t_k, fly(uav1, loc1)), p_2)))$	5
52	$Member(occ(0, t_k, fly(uav1, loc1))),$	
	$cons(occ(t_k, t, \text{attach}(\text{uav1}, \text{crate1}))),$	
	$cons(occ(0, t_k, fly(uav1, loc1)), p_2)))$	50, 51
53	$Occurs(0, t_k, fly(uav1, loc1),$	
	$cons(occ(t_k, t, \text{attach}(\text{uav}1, \text{crate1}))),$	
	$cons(occ(0, t_k, fly(uav1, loc1)), p_2)))$	6,52
54	$Holds(t_k, loc(uav1), loc1,$	
	$cons(occ(t_k, t, \text{attach}(\text{uav}1, \text{crate}1)),$	
	$cons(occ(0,t_k,{\rm fly}({\rm uav1},{\rm loc1})),p_2)))$	25, 47, 53
55	$\forall p_2 [Holds(t_k, loc(uav1), loc1,$	
	$cons(occ(t_k, t, \text{attach}(\text{uav1}, \text{crate1}))),$	
	$cons(occ(0, t_k, fly(uav1, loc1)), p_2)))]$	54

At this point, Subgoal 1-4 in the antecedent of Row 17 have been proved for the following instantiations of the universally quantified variables:

$$\begin{split} t_1 &= t_k \\ l &= \mathrm{loc1} \\ p_1 &= \mathrm{cons}(\mathrm{occ}(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2) \\ p &= \mathrm{cons}(\mathrm{occ}(t_k, t, \mathrm{attach}(\mathrm{uav1}, \mathrm{crate1})), \\ &\quad \mathrm{cons}(\mathrm{occ}(0, t_k, \mathrm{fly}(\mathrm{uav1}, \mathrm{loc1})), p_2)) \end{split}$$

The qualification on Subgoal 3 can be satisfied by choosing a plan tail  $p_2$  that does not contain any *attach* actions. The most obvious choice is  $p_2 = nil$ :

56 
$$\forall x, y, s [\neg(x = y) \land \neg Member(x, s) \rightarrow \neg Member(x, cons(y, s))]$$
 5

57 
$$\neg (occ(t_1, t_2, \operatorname{attach}(u, c)) = occ(0, t_k, \operatorname{fly}(\operatorname{uav1}, \operatorname{loc1})))$$
 F

58 
$$\neg Member(occ(t_1, t_2, \operatorname{attach}(u, c)), nil)$$

59 
$$\forall t_1, t_2, u, c \left[\neg Member(occ(t_1, t_2, \operatorname{attach}(u, c)), cons(occ(0, t_k, \operatorname{fly}(\operatorname{uav1}, \operatorname{loc1})), nil))\right]$$
 56, 57, 58

$$\begin{array}{c} 60 \quad \neg \exists t_1, t_2, u, c \left[ Member(occ(t_1, t_2, \operatorname{attach}(u, c)), \\ cons(occ(0, t_k, \operatorname{fly}(\operatorname{uav1}, \operatorname{loc1})), nil)) \right] \end{array}$$

This completes the proof of the main goal:

By extracting the plan from the action occurrence set one can utilize deductive proofs for planning. The above proof was one of many possible constructive proofs of the goal. Through the application of an automated theorem prover that considers all the possibilities in the different choice points in the proof process one obtains a complete planner. However, even the trivial planning problem solved above had a relatively long and complicated proof. Direct application of automated theorem provers to an  $\mathcal{L}(FL)$  narrative would not result in a practical planning system. Clearly, more work needs to be done before deductive planning becomes truly useful in practice. This is the topic of the next chapter.

## Chapter 4

# Compiling TAL into Logic Programs

We introduce a four step translation from  $\mathcal{L}(ND)$  narratives into constraint logic programs that makes use of the reified action occurrences, interval occlusion, and temporal constraint formalisms described in Chapter 3. The four steps are as follows:

(Section 4.2)	Compile the TAL narrative from the surface language
	$\mathcal{L}(ND)$ to the first-order logic language $\mathcal{L}(FL)$ .
(Section 4.3)	Replace timepoint occlusion in formulas by interval occlu-
	sion.
(Section 4.4)	Rewrite all formulas into Horn clause form.
(Section 4.6)	Encode the Horn clauses as Prolog clauses, constraint han-
. ,	dling rules, and finite domain constraints.

The resulting programs are efficiently evaluated using Prolog and can be used for both deductive planning and reasoning. The compilation provides a means for the application of deductive planning in TAL to solving practical reasoning problems such as those encountered by our UAV research platform.

Any translation to constraint logic programs is bound to be partial since the source language has the expressivity of first-order logic while the target language is limited to Horn clause form and the constraint formalisms used. Restrictions are therefore set up to isolate the class of narratives of interest. To prevent the implications of this from being lost in the technical details of the translation process we conclude this chapter with a summary and discussion in Section 4.7.

## 4.1 $\mathcal{L}(ND)$ Narrative

The first restriction is to narratives without domain or dependency constraints. Such narratives will only consist of action specifications that detail the conditions and effects of actions, persistence statements that specify which fluents retain their value over time, and observation statements that describe the initial state of the world. Action specifications are assumed to be of the form:

$$[t_1, t_2] \ a(\overline{w}) \rightsquigarrow \forall \overline{x} \left[ P(\overline{y}) \to E(\overline{z}) \right] \tag{1}$$

where  $\overline{y}, \overline{z} \subseteq \overline{w} \cup \overline{x}$ .  $P(\overline{y})$  is a conjunction of conditions on fluent values, each having the form:

 $[t_1, t_2] f(\overline{y}) \stackrel{\circ}{=} v$ 

where the end timepoint  $t_2$  is optional and the value v could be one of the variables in  $\overline{w} \cup \overline{x}$ , or a constant, including the boolean constants *true* and *false*.  $E(\overline{z})$  is a conjunction of reassignments of fluent values, using the reassignment macro R, each of the form:

 $R((t_1, t_2] \ f(\overline{z}) \stackrel{\circ}{=} v)$ 

where  $\hat{=}$  can be replaced by  $\hat{\neq}$  and, again, v belongs to  $\overline{w} \cup \overline{x}$  or is a constant. Fluents will usually be persistent, as specified by persistence statements of the form:

 $\forall t, \overline{x} \left[ Per(t, f(\overline{x})) \right]$ 

Finally, the values of fluents at the start of the planning process are (completely) specified using observations at timepoint zero of the form:

$$[0] f \stackrel{\circ}{=} v$$

where v is a constant. In order to make these general forms more concrete they will be accompanied by a running example consisting of the UAV logistics scenario fragment from Section 3.3. Again, the example consists of an action *attach*, used by a UAV to attach a crate that can then be moved to another location. The action specification involves a location-valued fluent loc(object), representing the location of an object, and a boolean-valued fluent carrying(uav, crate) that is true whenever the UAV given as the first argument is carrying the crate in the second argument. The *attach* action will make the *carrying* fluent true and the value of the *loc* fluent unspecified given that the UAV remains in the same location as the crate during the execution of the action. The  $\mathcal{L}(ND)$  specification is given by:
$$\begin{array}{l} [t_1, t_2] \mbox{ attach}(u, c) \rightsquigarrow \\ \forall l \ [[t_1] \ \log(c) \triangleq l \land [t_1, t_2] \ \log(u) \triangleq l \rightarrow \\ R((t_1, t_2] \ \operatorname{carrying}(u, c)) \land R((t_1, t_2] \ \neg \mathrm{loc}(c) \triangleq l)] \end{array}$$

Both *carrying* and *loc* are persistent fluents:

```
\forall t, o \left[ Per(t, loc(o)) \right] \\ \forall t, u, c \left[ Per(t, carrying(u, c)) \right] \end{cases}
```

Finally, an initial state is set up. We introduce a fluent *pos* that can be used to refer to locations using coordinates, e.g. as follows:

- $[0] \log(uav1) = pos(300, 90)$
- $[0] \ loc(crate1) \stackrel{\circ}{=} pos(300, 90)$
- $[0] \neg carrying(uav1, crate1)$

### 4.2 Translation from $\mathcal{L}(ND)$ to $\mathcal{L}(FL)$

The first step in translating the above  $\mathcal{L}(ND)$  narrative to an executable logic program is to use the transformation function *Trans* to compile the narrative into the first-order logic language  $\mathcal{L}(FL)$ . Using the definition of *Trans* given in Section 2.3.1, augmented with the reified action occurrences argument p, the action specifications (1) are translated into:

$$\forall t_1, t_2, \overline{w}, p \ [t_1 < t_2 \land Occurs(t_1, t_2, a(\overline{w}), p) \to \forall \overline{x} \ [P(\overline{y}) \to E(\overline{z})]] \tag{2}$$

Each condition in the conjunction  $P(\overline{y})$  has been translated into:

 $Holds(t_1, f(\overline{y}), v, p)$ 

or to  $\forall t [t_1 \leq t \leq t_2 \rightarrow Holds(t, f(\overline{y}), v, p)]$  if the condition was specified to hold throughout the entire action interval by supplying an end timepoint  $t_2$ . Each reassignment in the conjunction  $E(\overline{z})$  is translated into:

$$(\neg)Holds(t_2, f(\overline{z}), v, p) \land \forall t \ [t_1 < t \le t_2 \rightarrow Occlude(t, f(\overline{z}), p)]$$

Persistence statements are instantiations of the general persistence formula for a given fluent f, with the addition of the action occurrence argument:

$$\begin{array}{l} \forall t, \overline{x}, p \left[ \neg Occlude(t+1, f(\overline{x}), p) \rightarrow \\ \forall v \left[ Holds(t, f(\overline{x}), v, p) \leftrightarrow Holds(t+1, f(\overline{x}), v, p) \right] \right] \end{array}$$

Each initial fluent value observation becomes:

 $\forall p [Holds(0, f, v, p)]$ 

Finally, the *Occurs* predicate for reified action occurrences was defined in Section 3.1 as:

$$\forall t_1, t_2, a, p \left[ Occurs(t_1, t_2, a, p) \leftrightarrow Member(occ(t_1, t_2, a), p) \right]$$
(3)

Returning to the example from the UAV logistics narrative fragment, its action specification, persistence statements, and initial state observations are translated into the following set of formulas:

$$\begin{split} \forall t_1, t_2, u, c, l, p & [t_1 < t_2 \land Occurs(t_1, t_2, \operatorname{attach}(u, c), p) \rightarrow \\ & Holds(t_1, \operatorname{loc}(c), l, p) \land \forall t & [t_1 \leq t \leq t_2 \rightarrow Holds(t, \operatorname{loc}(u), l, p)] \rightarrow \\ & Holds(t_2, \operatorname{carrying}(u, c), \operatorname{true}, p) \land \\ & \forall t & [t_1 < t \leq t_2 \rightarrow Occlude(t, \operatorname{carrying}(u, c), p)] \land \\ & \neg Holds(t_2, \operatorname{loc}(c), l, p) \land \\ & \forall t & [t_1 < t \leq t_2 \rightarrow Occlude(t, \operatorname{loc}(c), p)]] \\ \forall t, x, p & [\neg Occlude(t+1, \operatorname{loc}(x), p) \rightarrow \\ & \forall v & [Holds(t, \operatorname{loc}(x), v, p) \leftrightarrow Holds(t+1, \operatorname{loc}(x), v, p)]] \\ \forall t, u, c, p & [\neg Occlude(t+1, \operatorname{carrying}(u, c), p) \rightarrow \\ & \forall v & [Holds(t, \operatorname{carrying}(u, c), v, p) \leftrightarrow Holds(t+1, \operatorname{carrying}(u, c), v, p)]] \\ \forall p & [Holds(0, \operatorname{loc}(\operatorname{cate1}), \operatorname{pos}(300, 90), p)] \\ \forall p & [Holds(0, \operatorname{carrying}(\operatorname{uav1}, \operatorname{crate1}), \operatorname{false}, p)] \end{split}$$

#### 4.3 Introducing Interval Occlusion

The second step consists of obtaining a definition of timepoint occlusion specific to the given narrative and then replacing it by an equivalent definition of interval occlusion. To accomplish this we split the action specification formulas, each of the form (2), into fluent value reassignments and occlusion formulas. The occlusion predicates are then collected into a single *Occlude* predicate using the equivalence:

$$\Phi(f_1) \wedge \dots \wedge \Phi(f_n) \equiv \forall x \left[ (x = f_1 \vee \dots \vee x = f_n) \to \Phi(x) \right]$$
(4)

The resulting form is:

$$\forall t_1, t_2, \overline{w}, \overline{x}, p \ [t_1 < t_2 \land Occurs(t_1, t_2, a(\overline{w}), p) \land P(\overline{y}) \to E'(\overline{z})] \tag{5}$$

$$\forall t, f, t_1, t_2, \overline{w}, \overline{x}, p \ [t_1 < t \le t_2 \land Occurs(t_1, t_2, a(\overline{w}), p) \land P(\overline{y}) \land (f = f_1(\overline{z}) \lor \dots \lor f = f_n(\overline{z})) \to (6)$$

$$Occlude(t, f, p)]$$

where E' are the reassignments without the occlusion sub-formulas. Applying this rewrite to the example *attach* action specification produces:

To be able to conveniently prove persistency over qualitative time intervals we would like to express these timepoint occlusion formulas using interval occlusion. The definition of interval occlusion in terms of timepoint occlusion, repeated here with the additional argument for reified action occurrences, is:

$$\forall t_1, t_2, f, p \left[ Occlude(t_1, t_2, f, p) \leftrightarrow \exists t \left[ t_1 < t \le t_2 \land Occlude(t, f, p) \right] \right]$$
(8)

Before this definition can be applied we need to put the occlusion formulas (6) into a form where they contain the sub-formula  $\exists t [t_1 < t \leq t_2 \land Occlude(t, f, p)]$ . According to (8) this is equivalent to, hence can be replaced by, interval occlusion. The following equivalence, proven in Appendix B.3, serves this purpose:

$$\forall t \left[ \Phi(t) \to \Psi(t) \right] \equiv$$

$$\forall t_1, t_2 \left[ \exists t \left[ t_1 < t \le t_2 \land \Phi(t) \right] \to \exists t \left[ t_1 < t \le t_2 \land \Psi(t) \right] \right]$$

$$(9)$$

Let the antecedent and consequent of the occlusion implication (6), with  $t_1$  and  $t_2$  renamed to  $t_3$  and  $t_4$ , be  $\Phi$  and  $\Psi$  as in:

$$\begin{split} \Phi(t) \stackrel{\text{\tiny def}}{=} t_3 < t \leq t_4 \land Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \land \\ (f = f_1(\overline{z}) \lor \cdots \lor f = f_n(\overline{z})) \\ \Psi(t) \stackrel{\text{\tiny def}}{=} Occlude(t, f, p) \end{split}$$

Then the occlusion formula (6) is rewritten according to theorem (9) as:

$$\begin{aligned} \forall t_1, t_2, t_3, t_4, f, \overline{w}, \overline{x}, p \left[ \exists t \left[ t_1 < t \leq t_2 \land t_3 < t \leq t_4 \land \\ Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \land \\ (f = f_1(\overline{z}) \lor \cdots \lor f = f_n(\overline{z})) \right] \rightarrow \\ \exists t \left[ t_1 < t \leq t_2 \land Occlude(t, f, p) \right] \end{aligned}$$

The right hand side of the implication is now in the form required for the application of definition (8). Replacing it and reducing the scope of the existential quantifier on the left hand side results in:

$$\forall t_1, t_2, t_3, t_4, f, \overline{w}, \overline{x}, p \left[ \exists t \left[ t_1 < t \leq t_2 \land t_3 < t \leq t_4 \right] \land \\ Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \land \\ (f = f_1(\overline{z}) \lor \cdots \lor f = f_n(\overline{z})) \to \\ Occlude(t_1, t_2, f, p) \right]$$

The sub-formula  $\exists t \ [t_1 < t \leq t_2 \land t_3 < t \leq t_4]$  expresses the existence of a shared point between the intervals  $(t_1, t_2]$  and  $(t_3, t_4]$ . One can construct the corresponding interval algebra relation  $\{o, s, d, f, =, fi, di, si, oi\}$  by collecting all primitive interval relations that express some form of overlapping. Using the end-point comparison tree described by Thornton et. al. [38] this seemingly complex relation can be simplified to the end-point comparison  $t_2 > t_3 \land t_1 < t_4$ . To point out the correspondence between the existentially quantified timepoint sub-formula and the end-point comparison definition of interval overlap, proved in Appendix B.4, we will make use of a new predicate *Overlap*, defined as follows:

$$\forall t_1, t_2, t_3, t_4 \ [t_2 > t_3 \land t_1 < t_4 \leftrightarrow \text{Overlap}(t_1, t_2, t_3, t_4)]$$

Introducing the *Overlap* predicate into the occlusion formula puts all occlusion formulas into the general form:

$$\forall t_1, t_2, t_3, t_4, f, \overline{w}, \overline{x}, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land \\ Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \land \\ (f = f_1(\overline{z}) \lor \cdots \lor f = f_n(\overline{z})) \to \\ Occlude(t_1, t_2, f, p) \right]$$
(10)

The specific form of the example *attach* occlusion formula is:

$$\forall t_1, t_2, t_3, t_4, f, u, c, l, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land \tag{11} \right. \\ \left. \begin{array}{l} Occurs(t_3, t_4, \text{attach}(u, c), p) \land \\ Holds(t_3, \text{loc}(c), l, p) \land \\ \forall t \left[ t_3 \leq t \leq t_4 \rightarrow Holds(t, \text{loc}(u), l, p) \right] \land \\ \left. (f = \text{carrying}(u, c) \lor f = \text{loc}(c)) \rightarrow \\ Occlude(t_1, t_2, f, p) \right] \end{aligned}$$

To complete the transformation to interval occlusion, and take advantage of the possibility to prove a fluent persistent over an entire temporal interval, the persistence formulas must be replaced by formulas of the same form as formula (3) in Section 3.2 but including the action occurrence argument:

$$\forall t_1, t_2, \overline{x}, p \ [t_1 < t_2 \land \neg Occlude(t_1, t_2, f(\overline{x}), p) \to \\ \forall v \ [Holds(t_1, f(\overline{x}), v, p) \leftrightarrow Holds(t_2, f(\overline{x}), v, p)] ]$$
(12)

#### 4.4 Transformation to Horn Form

The third translation step rewrites formulas into Horn clause form. The reassignment part of action specifications (5) is now of the general form:

$$t_1 < t_2 \land Occurs \land P_1 \land \dots \land P_m \to E'_1 \land \dots \land E'_n$$

where, as before, each  $P_i$  is a condition and each  $E'_j$  is a reassignment without the occlusion part. Each such formula is expanded into the implications:

$$t_1 < t_2 \land Occurs \land P_1 \land \dots \land P_m \to E'_1$$

$$\vdots$$

$$t_1 < t_2 \land Occurs \land P_1 \land \dots \land P_m \to E'_n$$
(13)

This is in Horn form except for those conditions  $P_i$  that were specified to hold over the whole action occurrence interval  $[t_1, t_2]$ , which are still of the complex form:

$$\forall t \ [t_1 \le t \le t_2 \to Holds(t, f(\overline{y}), v, p)]$$

For persistent fluents this is equivalent to  $f(\bar{y})$  holding the value v at the first timepoint  $t_1$  and, by virtue of not being occluded at any timepoint during the interval, the value v persisting until timepoint  $t_2$ :

$$Holds(t_1, f(\overline{y}), v, p) \land \neg \exists t [t_1 < t \le t_2 \land Occlude(t, f(\overline{y}), p)]$$

According to the definition of interval occlusion (8) this is simply:

$$Holds(t_1, f(\overline{y}), v, p) \land \neg Occlude(t_1, t_2, f(\overline{y}), p)$$
(14)

The occlusion formulas (10) are in Horn form except for the disjunction over fluents. Rewriting the formula by distributing conjunction over the fluent disjunction, and then distributing the resulting conjunction over the implication, results in one implication for each equality  $f = f_i(\bar{z})$ . Applying equivalence (4) in the left direction to each implication results in one occlusion definition for each fluent:

$$\begin{array}{l} \forall t_1, t_2, t_3, t_4, \overline{w}, \overline{x}, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \rightarrow \\ Occlude(t_1, t_2, f_1(\overline{z}), p) \right] \\ \vdots \\ \forall t_1, t_2, t_3, t_4, \overline{w}, \overline{x}, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, a(\overline{w}), p) \land P(\overline{y}) \rightarrow \\ Occlude(t_1, t_2, f_n(\overline{z}), p) \right] \end{array}$$

Each of these formulas contain the action conditions  $P(\bar{y})$ . This reflects the fact that attempting to execute an action when its conditions are not satisfied will have no effect. We will perform an additional simplification on the occlusion formulas by removing these conditions. In our application, the synthesis and reasoning about plans and actions to execute in the future, considering actions that are not executable is not really relevant. When only considering actions

that can actually be executed this simplification has no effect on the results. Each simplified occlusion formula is then of the form:

$$\forall t_1, t_2, t_3, t_4, \overline{w}, \overline{x}, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, a(\overline{w}), p) \rightarrow Occlude(t_1, t_2, f(\overline{z}), p) \right]$$
(15)

Returning to the example logistics narrative fragment, the occlusion formula (11) is instantiated over the fluents *carrying* and *loc*, and simplified by ignoring the conditions for executing the action:

$$\begin{aligned} \forall t_1, t_2, t_3, t_4, u, c, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, \text{attach}(u, c), p) \rightarrow \\ Occlude(t_1, t_2, \text{carrying}(u, c), p) \right] \\ \forall t_1, t_2, t_3, t_4, u, c, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, \text{attach}(u, c), p) \rightarrow \\ Occlude(t_1, t_2, \text{loc}(c), p) \right] \end{aligned}$$

To complete the transformation to Horn clause form we need to deal with the equivalence in persistence statements (12) and the *Occurs* definition (3). Both can be rewritten in the form of two implications, however, in each case only one direction is of interest. Persistence will only be proved forward in time, and occurrences are only proved from the occurrence set argument. These implications are then in Horn form:

$$\forall t_1, t_2, \overline{x}, v, p \ [t_1 < t_2 \land \neg Occlude(t_1, t_2, f(\overline{x}), p) \land$$

$$Holds(t_1, f(\overline{x}), v, p) \rightarrow$$

$$Holds(t_2, f(\overline{x}), v, p)]$$

$$\forall t_1, t_2, a, p \ [Member(occ(t_1, t_2, a), p) \rightarrow Occurs(t_1, t_2, a, p)]$$

$$(17)$$

#### 4.5 Circumscription Policy

Before moving on to the encoding of the formulas as logic programming clauses we need to clarify the relation between the current state of the formulas and the TAL circumscription policy, which was defined in Section 2.5. It is designed to capture the commonsense intuition that change does not spontaneously happen unless there is some reason for it. E.g. if the UAV does not fly anywhere then, by default, its location is unchanged. More formally, given a suitable TAL narrative describing the workings of the UAV, we would like to conclude that the value of the UAV's location fluent will persist over a time interval, unless we in fact know that the UAV was involved in some flying activity during the interval.

In practice, restrictions on the form of TAL narratives ensure that the circumscription policy is reducible to predicate completion, even in the general case without our additional restrictions added to ensure a proper translation into constraint logic programs. In the compilation above, the occlusion formulas are already in a form amenable to completion by replacing the implications by equivalences, which is the standard way of circumscribing *Occlude* and *Occurs* in the first-order version of TAL narratives. In the case of our logic program encoding, predicate completion can be viewed as the declarative semantics of the negation as failure mechanism of logic programs. This fact suggests that we should leave the occlusion formulas as they are and rely on the "built-in" completion of Prolog to implement the circumscription policy.

However, the situation is complicated when one considers planning. There is no action sequence given beforehand that determines occlusion once and for all. On the contrary, the occlusion concept is continually used to prove fluent value persistence while, at the same time, actions are added to the reified action occurrence set that might affect and modify the very same occlusion. Consequently, we need to continually ensure that considered actions do not violate any persistence proofs that have already been assumed. The Constraint Handling Rules (CHR) framework [10] together with the use of temporal constraint formalisms make a flexible solution to this problem possible. By expressing occlusion using constraint handling rules instead of Prolog clauses we ensure that the conditions are automatically reevaluated whenever needed. The next section presents these rules and explains how they implement a dynamic action/persistence conflict resolution.

Finally, action occurrences should also be circumscribed. Since occlusion will be represented by constraint handling rules, and occlusion depends on action occurrences, we will need to express action occurrences as constraints too. Every time an action is added to the action occurrence set, passed around in the p argument, we will also add it to the constraint store. While the action occurrence argument represents a possibly incompletely specified action sequence, the constraint store is implicitly closed. Therefore the collection of action occurrences added to the store serves as a representation of the circumscription of *Occurs*.

## 4.6 Clauses and Rules

The fourth and final translation step encodes the TAL formulas as Prolog clauses, constraint handling rules, and finite domain constraints. We start with the *Occurs* definition (17):

```
occurs(T1,T2,A,P) :-
    member(occ(T1,T2,A),P), circ_occurs(T1,T2,A).
```

```
member(A,[A|Pp]).
member(A,P) :-
       nonvar(P), P = [A1|P1], A \setminus == A1, member(A,P1).
```

This deviates from a literal translation into Prolog in two ways. First, the circ\_occurs constraint was added to make sure the action occurrence is present in the constraint store, where it can interact with the constraint handling rules for occlusion, as noted in Section 4.5. Second, the member predicate uses Prolog's built in list syntax rather than the *cons* function, introduced in Section 3.1. Though our implementation of member is not the same as the standard one provided by the Prolog list library. The reason is that if one adds an action, and is subsequently forced to backtrack due to reaching a dead end in the search space, then the library version supplies an infinite number of irrelevant backtracking possibilities. Instead of trying some other action, one would be stuck considering syntactically alternative ways of adding the same action to the list. Action occurrence terms should be collected in a *set*, without consideration of their syntactic order. This is the behaviour of the fluent composition function  $\circ$ , used to represent world states in the Fluent Calculus [37]. Our implementation of member borrows Thielscher's fluent composition predicate holds from the Fluent Calculus FLUX system [36], which provides only relevant backtracking opportunities.

Next, each of the n action reassignment formulas in (13) is of the form:

$$\forall t_1, t_2, \overline{w}, \overline{x}, p \ [t_1 < t_2 \land Occurs(t_1, t_2, a(\overline{w}), p) \land \\ Holds(t_1, f_1(\overline{y_1}), v_1, p) \land \dots \land Holds(t_1, f_m(\overline{y_m}), v_m, p) \rightarrow \\ Holds(t_2, f(\overline{z}), v, p)]$$

with, according to (14), a  $\neg Occlude(t_1, t_2, f_i(\overline{y}), p)$  for those conditions that should persist over the entire action occurrence interval. This is written as a Prolog clause:

$$\begin{array}{ll} \texttt{holds}(\texttt{T2},f(\overline{z}),v,\texttt{P}) &:- \\ & types(\overline{x}), \ \texttt{T1} \ \texttt{\#}<\texttt{T2}, \ \texttt{occurs}(\texttt{T1},\texttt{T2},a(\overline{w}),\texttt{P}), \\ & \texttt{holds}(\texttt{T1},f_1(\overline{y_1}),v_1,\texttt{P}), \ \cdots, \ \texttt{holds}(\texttt{T1},f_m(\overline{y_m}),v_m,\texttt{P}). \end{array}$$

.

with the addition of not\_occlude(T1,T2,  $f_i(\overline{y})$ ,P) for persisting conditions. Since the timepoints  $t_1$  and  $t_2$  are usually not instantiated as integers, the inequality between them can not be expressed by Prolog's built-in arithmetic operator. Instead the inequality is added to the finite domain constraint solver store using #<. Finally,  $\mathcal{L}(FL)$  is an order-sorted logic that may associate type restrictions on fluent arguments. Prolog is not typed, but the types predicate is expanded into unary type clauses that ensure that action argument variables only take on values from the value domain that corresponds to their type. For example, the action of attaching crates involves a variable c ranging over the set of crates. The Prolog translation of its action reassignment formula would include a unary predicate crate(C), and the set of crates that the variable ranges over would be added to the program as assertions:

```
crate(crate1).
crate(crate2).
crate(crate3).
```

A special timepoint type predicate ensure that the timepoint variables T1 and T2 are assigned a finite domain, as required by the finite domain constraint solver.

Before moving on, note that Prolog does not have classical negation. Consequently negative effects, such as  $\neg Holds(t_2, loc(c), l, p)$  in our attach example, can not be encoded explicitly in the logic program. Negation is implicit in the assumption of complete knowledge and negation as failure. Thus, if a fluent has not been explicitly mentioned in the effects of some action, it can be proven false using negation as failure. One might wonder whether persistence would not allow propagating a fluent's value from some earlier timepoint, even in the presence of an intermediate action with the negation of the fluent among its effects, since that intermediary negative effect is not explicitly encoded. However, the fluent is still explicitly *occluded*, as specified by the occlusion part of the action specification. Any action that occludes the fluent while overlapping the persistence interval will, by the definition of interval occlusion, also occlude the persistence interval, and persistence will not be applicable.

Once again returning to the logistics example, the negative effect of *attach* in formula (7) is accordingly left out of the logic program while the positive effect, which is of the form:

$$\begin{aligned} \forall t_1, t_2, u, c, l, p \left[ t_1 < t_2 \land Occurs(t_1, t_2, \operatorname{attach}(u, c), p) \land \\ Holds(t_1, \operatorname{loc}(c), l, p) \land Holds(t_1, \operatorname{loc}(u), l, p) \land \\ \neg Occlude(t_1, t_2, \operatorname{loc}(u), p) \rightarrow \\ Holds(t_2, \operatorname{carrying}(u, c), \operatorname{true}, p) \right] \end{aligned}$$

is compiled into:

holds(T2,carrying(U,C),true,P) : uav(U), crate(C), timepoint(T1), timepoint(T2),
 T1 #< T2, occurs(T1,T2,attach(U,C),P),
 holds(T1,loc(C),L,P), holds(T1,loc(U),L,P),
 not\_occlude(T1,T2,loc(U),P).</pre>

where uav/1 and crate/1 are the type predicates that prevent the action from being applied to anything else than UAVs and crates, and timepoint/1 assign finite domains to the timepoint variables.

According to the declarative CHR semantics (page 7, [10]), each occlusion formula of the form (15), repeated here:

$$\forall t_1, t_2, t_3, t_4, \overline{w}, \overline{x}, p \left[ \text{Overlap}(t_1, t_2, t_3, t_4) \land Occurs(t_3, t_4, a(\overline{w}), p) \rightarrow Occlude(t_1, t_2, f(\overline{z}), p) \right]$$

is equivalent to a constraint handling rule of the following form:

overlap(T1,T2,T3,T4), occurs(T3,T4, $a(\overline{w})$ ,P) ==> occlude(T1,T2, $f(\overline{x})$ ,P).

The purpose of this rule is to occlude fluents that are affected by action occurrences in the action occurrence set p. While planning, however, the set p is intentionally incompletely specified to allow the addition of new actions. Whether or not some specific action is present in the current action occurrence set p may not be known, and supplying the corresponding Prolog goal occurs (T3, T4,  $a(\overline{w})$ , P) would actually *add* the action if not already present, since this is a valid way of proving that goal. The same problem applies to the occlusion constraint on the right hand side of the rule since it also depends on the actions occurrence set. The best we can do is to make sure that fluents affected by actions that we have assumed occur, up to the current point in the planning process, are occluded. Those actions are stored using the circ\_occurs constraint, as noted in the translation of the Occurs definition (17). By replacing the use of predicates that depend on the incompletely specified set P by the corresponding *constraints*, circ\_occurs and occlude, we take all actions that are currently in the plan into consideration. The CHR framework then assumes responsibility for the rest of the work by continually monitoring these constraints for updates. Whenever the action occurrence set P is modified, a new circ\_occurs constraint is added, and the occlusion rule is automatically reevaluated. The new constraint handling rule is then:

$$\operatorname{overlap}(\mathtt{T1},\mathtt{T2},\mathtt{T3},\mathtt{T4}), \operatorname{circ_occurs}(\mathtt{T3},\mathtt{T4},a(\overline{w})) \implies (18)$$
$$\operatorname{occlude}(\mathtt{T1},\mathtt{T2},f(\overline{x})).$$

The persistence formulas are of the form (16), repeated here:

$$\forall t_1, t_2, \overline{x}, v, p \ [t_1 < t_2 \land \neg Occlude(t_1, t_2, f(\overline{x}), p) \land Holds(t_1, f(\overline{x}), v, p) \rightarrow Holds(t_2, f(\overline{x}), v, p)]$$

and are encoded as prolog clauses:

```
holds(T2, f(\overline{x}), v, P) :- (19)
timepoint(T1), timepoint(T2), T1 #< T2,
not_occlude(T1, T2, f(\overline{x})), holds(T1, f(\overline{x}), v, P).
```

The negated Occlude predicate is encoded as a constraint not\_occlude to allow the proper interaction with the occlude constraint in the occlusion CHR rule (18). As before, the best we can do is to consider all decisions made up to the current point in the planning process. CHR will then remember to reevaluate the rules whenever new descisions are made. Specifically, when the persistence formula (19) is used to propagate a fluent's value, the not\_occlude constraint would be added to the constraint store. If an overlapping action interval that affects the fluent's value was already present in the store, the fluent would not really be persistent. The occlusion rule (18) should be reevaluated and detect the conflict, adding a contradictory occlude constraint to the store. The only missing piece is a rule to force backtracking when such conflicts, caused by a persistence assumptions and action occurrences, are detected:

```
not_occlude(T1,T2,f(\overline{x})), occlude(T1,T2,f(\overline{x})) ==> fail.
```

However, there is one problem with this initial encoding attempt and it is related to the overlap constraint. The CHR rule (18) is a syntactic matching rule. It will only trigger if the constraints on its left hand side are explicitly represented in the constraint store. Action occurrences are stored explicitly in the circ\_occurs constraint, but the same is not true for the overlap constraint. Unless we have complete knowledge about the ordering of actions and represent this knowledge as explicit constraints, the rule may not trigger to detect conflicts between persistence assumptions and actions that are added to the plan. But we would like to have both a compact and *incomplete* representation of temporal constraints. This would allow the implementation to follow a strategy of least commitment concerning the orderings of actions and to generate partially ordered plans.

Solving this problem involves restructuring the occlusion formulas so that knowledge of interval overlap is not a *prerequisite* to conflict detection, but a *consequence*. The occlude formulas (15) have their relations rearranged into the logically equivalent form:

$$\forall t_1, t_2, t_3, t_4, \overline{w}, \overline{x}, p \left[\neg Occlude(t_1, t_2, f(\overline{z}), p) \land Occurs(t_3, t_4, a(\overline{w}), p) \rightarrow \\ \neg Overlap(t_1, t_2, t_3, t_4)\right]$$

Likewise, the new constraint handling rule, replacing (18), is:

Both the not\_occlude and circ\_occurs constraints have complete representations in the constraint store, thereby ensuring that the rule will trigger whenever these constraints are modified in the store. Thus, when clause (19) has been used to prove a fluent persistent over an interval  $(t_1, t_2]$ , adding a not\_occlude constraint in the process, and an action that threatens to modify the fluent occurs during interval  $(t_3, t_4]$ , then the rule will act to constrain the two intervals not to overlap. The not\_overlap predicate adds the corresponding interval algebra end-point constraints to the finite domain constraint solver discussed in Section 3.2.1. If the addition of these temporal constraints results in inconsistency, Prolog will backtrack, forcing the choice of another action or the removal of the persistence assumption. In other words, the mechanism resolves threats posed by actions, which could potentially disturb fluents that are required to be persistent, by introducing restrictions on the order of actions and persistence intervals. Backtracking is only necessary when further ordering restrictions are not possible.

In the logistics example the resulting persistence clauses and occlusion rules are as follows:

```
holds(T2,carrying(U,C),V,P) :-
timepoint(T1), timepoint(T2), T1 #< T2,
not_occlude(T1,T2,carrying(U,C)),
holds(T1,carrying(U,C),V,P).
holds(T2,loc(C),V,P) :-
timepoint(T1), timepoint(T2), T1 #< T2,
not_occlude(T1,T2,loc(C)),
holds(T1,loc(C),V,P).
not_occlude(T1,T2,carrying(U,C)),
circ_occurs(T3,T4,attach(U,C)) ==>
not_overlap(T1,T2,T3,T4).
not_occlude(T1,T2,loc(C)),
circ_occurs(T3,T4,attach(U,C)) ==>
not_overlap(T1,T2,T3,T4).
```

Finally, observation statements are easily encoded as:

holds(0, f, v, P).

In the case of the UAV example the encoding is:

holds(0,loc(uav1),pos(300,90),P). holds(0,loc(crate1),pos(300,90),P). The compilation results in executable constraint logic programs for deductive planning and reasoning. An example program is integrated with the UAV system in Chapter 5 and listed in full in the Appendix.

### 4.7 Summary

Although many translation steps are equivalence preserving, some are necessarily not, or else we would not have arrived at a logic program with deductive planning capability. We summarize the points of difference here.

- An obvious restriction on the original TAL is the form of  $\mathcal{L}(ND)$  input theories, which limits the applicability of the compilation to domains that can be adequately described while satisfying these constraints. The translation from  $\mathcal{L}(ND)$  to the first-order  $\mathcal{L}(FL)$  is unmodified. Instead, additional translations are subsequently performed.
- Timepoint occlusion is replaced by interval occlusion in the action specifications. Interval occlusion is defined in terms of TAL's original timepoint occlusion by (8), and the modification is expressed in terms of an equivalence preserving rewrite (Appendix B.3).
- Interval overlap is expressed succinctly using an *Overlap* predicate that is equivalent to the formula expressing interval overlap that resulted from the translation to interval occlusion (Appendix B.4).
- A formula (12) that allows the use of fluent persistency over intervals of any length is added. It is entailed by the definition of interval occlusion and persistence (Appendix B.1, Appendix B.2), but is weaker than the interval persistence formula (2) in Section 3.2 since it does not (by itself) constrain the fluent's value *during* the interval. The stronger formula could presumably be added, if needed, but we have not investigated this further.
- The formulas are expanded into equivalent formulas in Horn form, which is possible due to the restrictions on the original  $\mathcal{L}(FL)$  narrative. An additional requirement is that fluents that are specified to hold a specific value throughout the action occurrence interval are persistent. Otherwise one would need to use universal quantification over timepoints in that interval.
- Occlusion formulas are significantly simplified by removing the action preconditions (which are independently present in action effect formulas).

This weakens the theory since actions may occlude fluents, preventing application of persistence, even though the actions may in fact have had no effects due to their preconditions not being satisfied. Though, when planning, such cases will never actually appear since it would be pointless to add actions without effects to the narrative.

- When the persistence equivalence (12) is turned into a Horn form implication, this weakens the theory to persistence forward in time. We have not investigated whether the other direction could be included to allow *postdictive* conclusions about fluent values at earlier timepoints based on knowledge about later timepoints.
- CHR rules minimize occlusion and occurrences in the same way the circumscription policy would have done, from a theoretical point of view. However, the difference is that the proof *mechanism* is applicable even to a dynamically changing narrative, which is the case in deductive planning.
- The conversion of the Horn formulas into Prolog clauses strengthens the theory in the sense that anything that was previously unknown is now false. But it limits the applicability of our methods, as noted below.
- Finally, the use of a finite domain constraint solver for (in)equalities weakens the theory since we can not draw conclusions e.g. about fluent values outside the bounded time line. In practice this problem can be circumvented by adjusting the size of the bound.

The generated constraint logic programs are thus associated with severe restrictions on the form of TAL theories that can be used as input. Specifically, the requirement of a completely specified initial state and the negation-asfailure semantics of Prolog make the compilation, as it is described above, inapplicable to planning problems that involve incomplete information, and the restrictions imposed by Horn-clause form limit syntactic expressivity and the use of quantifiers.

It is important to note, however, that without some mechanism for incremental minimization of *Occlude* and *Occurs*, deductive planning would be impossible. Previous work with TAL (e.g. the most recent language specification [8] and the most used reasoning tool [18]) has assumed the provision of completely specified narratives, detailing what actions occur when, *before* any reasoning process commences. The constraint handling rules, introduced above, take care of the minimization in an automated fashion and allow the dynamic generation of new narratives. They significantly simplify the proof process and make automated planning efficient. Building on this basis will require augmentation of the basic Prolog inference mechanism through a metainterpreter, or replacing Prolog with a more powerful theorem prover, but such efforts should ultimately benefit from the inherent capability of the underlying TAL language to compactly represent incompletely specified information, and make possible a more liberal use of quantifiers.

# Chapter 5

# **Composite Actions**

The previous chapters have considered plans in the form of sets of primitive action occurrences that are partially ordered on a time line. But such a view is not sufficiently general to capture the entire range of plans that we would intuitively expect an intelligent agent to be able to use. Specifically, there is no concept of conditional action occurrences or of repetition of action occurrences. If such constructs are introduced they form a basic scripting language that can be used to express more complex instructions for the agent to carry out.

This chapter extends the high-level  $\mathcal{L}(ND)$  lanugage syntax to cover sequences of action occurrences, conditional occurrences, nondeterministic choice actions, and loops. A semantics for these *composite actions* is provided by a corresponding extension of the *Trans* function that translates the high-level language into the first-order language  $\mathcal{L}(FL)$ . The result, however, is no longer a first-order language. A move to fixpoint logic is necessary since we make use of the least fixpoint operator to capture the semantics of loop constructs. Finally, a compilation of composite actions into Prolog clauses is described, which permits the execution of complex actions in the logic programming framework that was developed in Chapter 4.

#### 5.1 Syntax and Semantics

The syntax and semantics of composite actions are defined by extending the translation function *Trans* defined in Section 2.3.1. The following additions illustrate both the  $\mathcal{L}(ND)$  syntax of composite actions and the result of translating them to  $\mathcal{L}(FL)$  with the least fixpoint operator. Let *a* be a primitive action, *C* a formula expressing some condition, and *A* a composite action. We

extend *Trans* with sequences, conditional action occurrences, non-deterministic choice, and loops as follows:

$$\begin{aligned} &Trans([t_1, t_2] \ A_1; A_2) \stackrel{\text{def}}{=} \\ & \exists t[t_1 < t < t_2 \land Trans([t_1, t] \ A_1) \land Trans([t, t_2] \ A_2)] \end{aligned}$$

$$\begin{aligned} &Trans([t_1, t_2] \ if \ C \ then \ A) \stackrel{\text{def}}{=} \\ &Trans([t_1] \ C) \rightarrow Trans([t_1, t_2] \ A) \end{aligned}$$

$$\begin{aligned} &Trans([t_1, t_2] \ choose \ x \ A) \stackrel{\text{def}}{=} \\ & \exists x[Trans([t_1, t_2] \ A)] \end{aligned}$$

$$\begin{aligned} &Trans([t_1, t_2] \ do \ A \ until \ C) \stackrel{\text{def}}{=} \\ & \mu X(x, y)[Trans([x] \ \neg C) \rightarrow \exists z[x < z \le y \land Trans([x, z] \ A) \land X(z, y)]](t_1, t_2) \end{aligned}$$

#### 5.2 Fixpoint Expansion

Although the semantics of the first three composite actions is relatively self explanatory, it might not be immediately obvious what the fixpoint formula in the fourth case means. A least fixpoint formula is of the general form:

 $\mu X(\overline{x})[\Phi](\overline{t})$ 

where X is k-ary relational variable,  $\overline{x}$  is k variables, and  $\Phi$  is a formula with only positive occurrences of X. Variables in  $\overline{x}$  that are unbound in  $\Phi$  serve as arguments for the fixpoint function and their values are provided in  $\overline{t}$ . The semantics of the expression can be defined model-theoretically as a least fixpoint relation on sets. However, we will instead illustrate its meaning using the following syntactic expansion theorem:

$$\mu X(\overline{x})[\Phi](\overline{t}) \equiv [\bigvee_{i \in \mathbb{Z}} X^i](\overline{t}) \quad \text{where} \quad X^i \equiv \begin{cases} false & \text{if } i = 0\\ \Phi[X(\overline{x}) \leftarrow X^{i-1}(\overline{x})] & \text{otherwise} \end{cases}$$

The fixpoint formula can thus be rewritten as an (infinite) disjunction where each disjunct is an expansion of the formula  $\Phi$  with occurrences of X replaced by the previous expansion step.

#### 5.2.1 Expansion Example

Consider a UAV logistics task where there is a need for a certain number of crates at a specific location. We would like to instruct the UAV to start delivering crates at timepoint s and stop at some later time t when a sufficient number of crates are available at the target location. For the purpose of illustration we help keep the formulas relatively lucid by abridging the delivery of a crate as an action *deliver* and the condition to be satisfied as a boolean fluent *sufficient*. A composite action to accomplish the given task would then be written as:

[s,t] do deliver until sufficient  $\hat{=}$  true

According to the extended *Trans* function defined above, this macro is expanded into the fixpoint  $\mathcal{L}(FL)$  formula:

$$\mu X(x,y)[\neg Holds(x, \text{sufficient}, true) \rightarrow \\ \exists z \ [x < z \le y \land Occurs(x, z, \text{deliver}) \land X(z, y)]](s, t)$$

The first step of the expansion,  $X^0$ , is simply *false*. The next step,  $X^1$ , is obtained by replacing the occurrence of X in the body of the fixpoint formula with  $X^0$  (which is *false*). The result is rewritten in disjunctive form to help illustrate a uniform structure that will become clear after a couple of expansion steps:

$$\neg Holds(x, \text{sufficient}, true) \rightarrow \\ \exists z \ [x < z \le y \land Occurs(x, z, \text{deliver}) \land X^0(z, y)] \equiv \\ Holds(x, \text{sufficient}, true) \lor \\ \exists z \ [x < z \le y \land Occurs(x, z, \text{deliver})] \end{cases}$$

Again, replacing X by  $X^1$  in the fixpoint formula body and rewriting results in  $X^2$ :

 $\begin{array}{l} \neg Holds(x, \mathrm{sufficient}, true) \rightarrow \\ \exists z \; [x < z \leq y \land Occurs(x, z, \mathrm{deliver}) \land X^1(z, y)] \equiv \\ Holds(x, \mathrm{sufficient}, true) \lor \\ \exists z \; [x < z \leq y \land Occurs(x, z, \mathrm{deliver}) \land \\ (Holds(z, \mathrm{sufficient}, true) \lor \exists z_2 \; [z < z_2 \leq y \land Occurs(z, z_2, \mathrm{deliver})])] \equiv \\ Holds(x, \mathrm{sufficient}, true) \lor \\ \exists z \; [x < z \leq y \land Occurs(x, z, \mathrm{deliver}) \land Holds(z, \mathrm{sufficient}, true)] \lor \\ \exists z, z_2 \; [x < z < z_2 \leq y \land Occurs(x, z, \mathrm{deliver}) \land Occurs(z, z_2, \mathrm{deliver})] \end{cases}$ 

The structure of the formula becomes more clear when completing the third step, replacing X by  $X^2$  and rewriting to get  $X^3$ :

 $\neg Holds(x, \text{sufficient}, true) \rightarrow \\ \exists z \left[ x < z \leq y \land Occurs(x, z, \text{deliver}) \land X^2(z, y) \right] \equiv$ 

```
 \begin{array}{l} Holds(x, \mathrm{sufficient}, true) \lor \\ \exists z \; [x < z \leq y \land Occurs(x, z, \mathrm{deliver}) \land \\ & (Holds(z, \mathrm{sufficient}, true) \lor \\ & \exists z_2 \; [z < z_2 \leq y \land Occurs(z, z_2, \mathrm{deliver}) \land Holds(z_2, \mathrm{sufficient}, true)] \lor \\ & \exists z_2, z_3 \; [z < z_2 < z_3 \leq y \land \\ & Occurs(z, z_2, \mathrm{deliver}) \land Occurs(z_2, z_3, \mathrm{deliver})])] \equiv \\ Holds(x, \mathrm{sufficient}, true) \lor \\ & \exists z \; [x < z \leq y \land Occurs(x, z, \mathrm{deliver}) \land Holds(z, \mathrm{sufficient}, true)] \lor \\ & \exists z, z_2 \; [x < z < z_2 \leq y \land Occurs(x, z, \mathrm{deliver}) \land \\ & Occurs(z, z_2, \mathrm{deliver}) \land Holds(z_2, \mathrm{sufficient}, true)] \lor \\ & \exists z, z_2, z_3 \; [x < z < z_2 < z_3 \leq y \land Occurs(x, z, \mathrm{deliver}) \land \\ & Occurs(z, z_2, \mathrm{deliver}) \land Holds(z_2, \mathrm{sufficient}, true)] \lor \\ & \exists z, z_2, z_3 \; [x < z < z_2 < z_3 \leq y \land Occurs(x, z, \mathrm{deliver}) \land \\ & Occurs(z, z_2, \mathrm{deliver}) \land Occurs(z_2, z_3, \mathrm{deliver})] \end{array}
```

The expansion continues indefinitely but the general form of the resulting disjunction is clear:

```
Holds(x, sufficient, true) \lor
\exists z \, [x < z \le y \land
     Occurs(x, z, deliver) \land
     Holds(z, sufficient, true)] \lor
\exists z, z_2 \ [x < z < z_2 \le y \land
          Occurs(x, z, deliver) \land
          Occurs(z, z_2, \text{deliver}) \land
          Holds(z_2, \text{sufficient}, true)] \lor
\exists z, z_2, z_3 \ [x < z < z_2 < z_3 \leq y \land
              Occurs(x, z, deliver) \land
               Occurs(z, z_2, \text{deliver}) \land
               Occurs(z_2, z_3, \text{deliver}) \land
              Holds(z_3, \text{sufficient}, true)] \lor
\exists z, z_2, z_3, z_4 \ [x < z < z_2 < z_3 < z_4 \leq y \land
                    Occurs(x, z, deliver) \land
                    Occurs(z, z_2, \text{deliver}) \land
                    Occurs(z_2, z_3, \text{deliver}) \land
                    Occurs(z_3, z_4, \text{deliver}) \land
                   Holds(z_4, \text{sufficient}, true)] \lor \cdots
```

The above formula corresponds to the expected behaviour of the composite loop action. Each disjunct represents an additional occurrence of the *deliver* action in the series before the *sufficient* condition is satisfied. The unfolding of the composite loop action (in the context of a narrative that specifies the effects of the action, the initial conditions, and the *sufficient* condition) is given by the first satisfied disjunct in the formula with the unbound variables x and y instantiated by the s and t arguments.

#### 5.3 Compiling Composite Actions

Some automated reasoning method applicable to composite actions is needed before they become useful in practice. Section 3.3 hinted that direct application of first-order theorem proving techniques does not necessarily result in a practical system. Composite actions are translated into fixpoint logic where first-order proof methods are not even applicable. An alternative automated reasoning method is the compilation into constraint logic programs in Chapter 4, which can be extended to the case of composite actions.

The compilation results in a Prolog goal that when evaluated will produce the actual execution trace of primitive action occurrences where conditionals have been resolved, non-deterministic choices instantiated, and iteration fixpoints reached. As before, these primitive action occurrences are stored in a reified action occurrence arument p. Note that only primitive action occurrences are reified while composite action occurrences are Prolog clauses. The reified action occurrence argument does not contain composite actions but is introduced for compatibility with the deductive planning framework. Reified composite actions is a more complex topic, as discussed in Section 5.4.

#### 5.3.1 The Compilation Function

The first compilation step, the translation from  $\mathcal{L}(ND)$  into fixpoint logic, was introduced above. The second, and final step, is the compilation from fixpoint logic into constraint logic programs. The compilation is most easily defined recursively since a composite action may be composed of other composite actions. The compilation will be associated with a function named *Comp* for the purpose of conveniently expressing this recursion. The base case of the compilation function, the primitive action occurrence, is defined as the corresponding Prolog clause:

 $Comp(Occurs(t_1, t_2, a, p)) \stackrel{\text{def}}{=} \\ \text{occurs}(\texttt{T1,T2,}a, \texttt{P})$ 

Sequences are composed of two composite actions in series, with constraints on the shared timepoint:  $\begin{array}{l} Comp(\exists t[t_1 < t < t_2 \land Trans([t_1,t] \ A_1) \land Trans([t,t_2] \ A_2)]) \stackrel{\text{def}}{=} \\ \texttt{T1 \#< T, T \#< T2, } Comp(Trans([t_1,t] \ A_1)), \ Comp(Trans([t,t_2] \ A_2)) \end{array}$ 

Conditionals consist of two complex parts, a condition C and an action A. The translation of the condition C needs special care. If C evaluates to true, the set of action occurrences should include A, however, we do not want the evaluation of C *itself* to affect the set of action occurrences. This is accomplished by minimizing the action occurrences in the variable p through a function circ that removes the unbound tail variable in the corresponding Prolog list variable P. The result is a new action occurrence variable Pp (a legal Prolog variable name, unlike P') where we have assumed complete knowledge about action occurrences. Using this new, circumscribed, variable when evaluating the condition C prevents the addition of new actions. The implication can then be written using the Prolog disjunction operator ; as:

$$Comp(Trans([t_1] \ C) \to Trans([t_1, t_2] \ A)) \stackrel{\text{def}}{=} \\ \texttt{circ}(\mathsf{P}, \mathsf{Pp}), \ Comp(Trans([t_1] \neg C)) \ ; \ Comp(Trans([t_1, t_2] \ A)) \\ \end{cases}$$

Nondeterministic choices have an existentially quantified variable that is simply replaced by a Prolog variable in the resulting expression:

 $\begin{array}{l} Comp(\exists x[Trans([t_1, t_2] \ A)]) \stackrel{\text{def}}{=} \\ Comp(Trans([t_1, t_2] \ A)) \quad \text{with occurrences of } x \text{ replaced by } X \end{array}$ 

Loops, with the additional action occurrence argument, are compiled in two steps. First, the implicit least fixpoint function X is made explicit as a Prolog predicate **x** that defines the loop. As with conditional actions, the test C is evaluated with the action occurrence variable P replaced by a circumscribed version Pp. Secondly, a Prolog goal calls the new predicate to evaluate the actual parameters passed to the function:

$$\begin{array}{l} Comp(\mu X(x,y,p)[Trans([x] \neg C) \rightarrow \\ \quad \exists z[x < z \leq y \land Trans([x,z] \ A) \land X(z,y,p)]](t_1,t_2,p)) \stackrel{\text{def}}{=} \\ \text{a new Prolog predicate:} \\ \texttt{x}(\texttt{X},\texttt{Y},\texttt{P}) := \texttt{circ}(\texttt{P},\texttt{Pp}), \ Comp(Trans([x] \ C)) \ \texttt{;} \\ \quad \texttt{X} \ \texttt{\#} < \texttt{Z}, \ \texttt{Z} \ \texttt{\#} = < \ \texttt{Y}, \ Comp(Trans([x,z] \ A)), \ \texttt{x}(\texttt{Z},\texttt{Y},\texttt{P}). \\ \text{and the goal:} \\ \texttt{x}(t_1,t_2,\texttt{P}) \end{array}$$

Finally, since Prolog is not typed, nondeterministic choice variables and quantified variables in test conditions require additional clauses that represent unary type predicates. These ensure that the variables can only be bound to objects of the correct type. As before, the special timepoint clause assigns a finite domain to the timepoint constraint variables.

#### 5.3.2 An Example Composite Action

Consider the *attach* action TAL narrative from Chapter 4 extended with a converse *drop* action and a *fly* action. These primitive actions can be used to write composite actions for the delivery of crates between locations. E.g., one such logistics task is to fly a UAV to fetch crates at a location *loc1* and deliver them to another location *loc2* until there are at least two crates there. All of the different types of composite actions are present in its specification:

```
 \begin{array}{ll} [0,t] \ do \ (if \neg loc(uav1) \triangleq loc1 \ then \ fly(uav1, loc1); \\ choose \ c \ (attach(uav1, c); \\ fly(uav1, loc2); \\ drop(uav1, c))) \\ until \ \exists c_1, c_2[loc(c_1) \triangleq loc2 \land loc(c_2) \triangleq loc2 \land c_1 \neq c_2] \end{array}
```

The composite action is first translated into  $\mathcal{L}(FL)$ :

$$\begin{split} \mu X(x,y,p) [ \\ \neg \exists c_1, c_2 \left[ Holds(x, \operatorname{loc}(c_1), \operatorname{loc}2, p) \land Holds(x, \operatorname{loc}(c_2), \operatorname{loc}2, p) \land c_1 \neq c_2 \right] \rightarrow \\ \exists z \left[ x < z \leq y \land \\ \exists t_1 \left[ x < t_1 < z \land \\ (\neg Holds(x, \operatorname{loc}(\operatorname{uav}1), \operatorname{loc}1, p) \rightarrow \\ Occurs(x, t_1, \operatorname{fly}(\operatorname{uav}1, \operatorname{loc}1), p)) \land \\ \exists c \left[ \exists t_2 \left[ t_1 < t_2 < z \land \\ Occurs(t_1, t_2, \operatorname{attach}(\operatorname{uav}1, c), p) \land \\ \exists t_3 \left[ t_2 < t_3 < z \land \\ Occurs(t_2, t_3, \operatorname{fly}(\operatorname{uav}1, \operatorname{loc}2), p) \land \\ Occurs(t_3, z, \operatorname{drop}(\operatorname{uav}1, c), p) \right] \right] ] \land \\ X(z, y, p) ] ] (0, t, p) \end{split}$$

The logic program compilation then creates a new Prolog predicate that defines the least fixpoint loop formula, with additional type and timepoint predicates:

```
x(X,Y,P) :-
circ(P,Pp1), crate(C1), crate(C2), holds(X,loc(C1),loc2,Pp1),
holds(X,loc(C2),loc2,Pp1), C1 \== C2 ;
timepoint(Z), X #< Z, Z #=< Y,
timepoint(T1), X #< T1, T1 #< Z,
((circ(P,Pp2), holds(X,loc(uav1),loc1,Pp2)) ;
occurs(X,T1,fly(uav1,loc1),P)),
crate(C), timepoint(T2), timepoint(T3), T1 #< T2, T2 #< Z,
occurs(T1,T2,attach(uav1,C),P),
T2 #< T3, T3 #< Z,</pre>
```

```
occurs(T2,T3,fly(uav1,loc2),P),
occurs(T3,Z,drop(uav1,C),P),
x(Z,Y,P).
```

Finally, the Prolog goal consists of calling the new predicate with the actual parameters:

```
?- x(0,T,P), timepoint_variables(L), labeling([],L).
```

The timepoint\_variables predicate collects all the finite domain constraint variables expressing temporal constraints and the labeling predicate ensures their consistency (or forces backtracking otherwise).

Evaluating the goal in an initial state where the UAV is already at location loc1 binds the action occurrence argument P to the following sequence of action occurrences:

```
\begin{array}{l} occ(t_1, t_2, \text{attach}(\text{uav1}, \text{crate1})) \\ occ(t_2, t_3, \text{fly}(\text{uav1}, \text{loc2})) \\ occ(t_3, t_4, \text{drop}(\text{uav1}, \text{crate1})) \\ occ(t_4, t_5, \text{fly}(\text{uav1}, \text{loc1})) \\ occ(t_5, t_6, \text{attach}(\text{uav1}, \text{crate2})) \\ occ(t_6, t_7, \text{fly}(\text{uav1}, \text{loc2})) \\ occ(t_7, t_8, \text{drop}(\text{uav1}, \text{crate2})) \\ 0 < t_1 < t_2 < t_3 < t_4 < t_5 < t_6 < t_7 < t_8 \leq t \end{array}
```

#### 5.4 Reified Composite Actions

The composite actions that we have presented are given by an operator and executed by the robot. This is similar to how primitive actions were used in TAL before our deductive planning extensions. Just as reified action occurrences made deductive planning with primitive action occurrences possible, one would like to use reification or some other scheme to make deductive planning with composite actions possible. However, since the semantics of composite actions are expressed in fixpoint logic there is no complete proof procedure that can be applied in the general case. Moreover, loops and conditionals make up a Turing complete language, thereby equating the generation of composite actions with the general problem of program synthesis. These considerations bodes for both great challenges and great rewards for future research in this direction.

## Chapter 6

# A UAV Logistics Application

We call the constraint logic programming methodology just described PARA-DOCS for Planning And Reasoning As DeductiOn with ConstraintS. It has been integrated with our autonomous unmanned aerial vehicle system and applied to solving logistics problems to test the viability of the method.

The integrated system can use the set of actions in the logistics scenario from Section 5.3.2 to distribute crates according to a declarative goal specification by generating a plan that involves repeatedly attaching a crate, flying, and then dropping it at the intended location. The UAV operator uses a graphical drag'n'drop user interface to set up the goal specification for a mission. The goals are sent to a Prolog program that is compiled from the logistics TAL narrative. Executing the program corresponds to proving the goal, and a partial order plan is extracted from the constraint stores that result from the proof. The operator uses the GUI to choose a linearization, which is compiled into a sequence of commands that can be executed by the UAV system. The nonocclusion constraints on fluent values are then monitored during the actual execution to make sure that it proceeds according to the plan. A failure of a monitor condition triggers a plan repair where the current state of execution is used by PARADOCS to insert recovery actions that will put the execution back on track and ensure that the goals are satisfied.

This chapter describes the implementation of the above scenario when coupled to the helicopter simulator. We are currently at work building a winch system with an electro-magnetic hook and trying to solve the vision problems associated with detecting crates and hovering with sufficient precision over



Figure 6.1: A logistics mission in the graphical user interface.

them. These enhancements would make it possible to perform the scenario using the helicopter hardware.

### 6.1 Graphical User Interface

The graphical user interface is implemented in Lisp using the Franz Allegro CL Common Graphics multi platform GUI library [15]. It provides a window based interface with an aerial overview of the flying area overlayed by icons representing objects of interest. Setting up a logistics mission simply consists of dragging crate icons to indicate their intended goal locations. Consider e.g. the mission set up in Figure 6.1. The task is to deliver three crates from a common store to three different locations. Clicking the "Generate Plan"

button will call on SICStus Prolog 3.12.5 [30] to evaluate the logic program (found in the appendix) on the following goal generated by the interface:

```
:- holds(T,loc(crate3),pos(113,-96),P),
holds(T,loc(crate1),pos(82,6),P),
holds(T,loc(crate2),pos(277,12),P),
timepoint_variables(L), labeling([],L).
```

Iterative deepening on timepoint variable domains would generate a plan with the smallest consistent temporal network. Such an exhaustive search would be completed on an Intel Pentium M 1.8 GHz within thirteen minutes. But the goal-directed search exhibited by PARADOCS is able to find plans in the logistics scenario without a depth bound. The plans are not guaranteed to be optimal, but they avoid the complete exploration of the search space on each depth bound smaller than the one that produces the first plan. With this approach PARADOCS produces a plan consisting of 12 actions and 15 persistence intervals in a graph containing a total of 22 temporal intervals in less than a tenth of a second.

The need for a graphical interface was evident even during the development and experimentation with deductive planning and temporal constraints. The constraint networks are represented as predicate lists by the constraint solver. These lists quickly grow unreadably large, prompting some form of visualization tool for grasping the temporal relations between action intervals and fluent persistence intervals. The first of two such visualizations in the GUI is an interval algebra graph that displays the partially specified temporal relations between intervals. Figure 6.2 shows the solution graph for the above logistics mission. The interval algebra representation's support for partial knowledge in the form of disjunctions makes it impossible to visualize the order in which the actions of the plan will be executed. In fact, at this point the order has not yet been decided. Still, it is possible to partially order the graph so that intervals higher up in the display are known to occur before intervals below them.

Even though transitive and reflexive relations are hidden the graph is still difficult to decipher for all but the tiniest of plans. Instead, the operator clicks a "Next Linearization" button to apply a backtracking labeling algorithm that returns all possible linearizations of the partial order plan, one at a time. At this point visualization is much easier as the order of actions has been decided. The plan chart in Figure 6.3 shows the first linearization of the above graph.



Figure 6.2: An interval algebra graph visualization of a partially ordered plan.

#### 6.2 Execution and Monitoring

The GUI provides an interface for an operator of the UAV to the PARADOCS planning service, which is part of the software system that controls the robot platform [6]. The software system implements a deliberative/reactive architecture where components use the CORBA framework to call each other's methods. This enables a highly distributed system in which components can use different implementation languages and reside on different computers as long as they are connected in a network.

Planned actions are stored in an action queue. They are popped from the queue during execution and communicated to a Command Executor service that assumes responsibility for calling the appropriate low-level control functionalities to execute each action. The planned actions are guaranteed to be consistent with the persistence assumptions that were made during the plan-



Figure 6.3: A plan chart showing one of many possible linearizations of the plan.

ning process, but unexpected things can happen during execution. The plan's persistence constraints, in the form of non-occlusion formulas that must hold during specific time intervals, are therefore placed in a monitor queue in parallell with the action queue. Monitor constraints that are active during an interval are evaluated by regularly querying the simulator for the values of the fluents and making sure that the values are indeed persistent and do not change.

Assume however, for the sake of example, that the electro-magnet device looses hold of *crate1* while the UAV is flying towards its destination in the middle of executing the plan in Figure 6.3. The planning process had already recorded the assumption that the fluent carrying(uav1,crate1) is persistent during the flight. This is necessary to satisfy the conditions for the action of dropping *crate1* at its destination. But when the hold of the crate is lost, the truth value of the *carrying* fluent changes to false, which violates said monitoring formula. At this point the plan is no longer guaranteed to achieve the goal and execution is halted.

Since PARADOCS can be applied equally well to problems in-between planning and prediction we can use the part of the plan that has not yet been executed as input for a plan repair process. The following goal is automatically generated and passes the actions that are left to execute as input through the action occurrence argument and ensures that they satisfy the previously selected linearization using constraints on the timepoint variables:

The result is, in this case, the insertion of a fly(uav1,pos(153,2)) and an attach(uav1,crate1) action in front of the rest of the plan. These actions constitute a recovery by going back to pick up the crate that was accidentally dropped before continuing with the rest of the plan.

## Chapter 7

# **Related Work**

Much of the inspiration to our work comes from other research groups with similar approaches. We list here only the most important influences and start with Shanahan's abductive Event Calculus planner [33]. It is based on the Event Calculus, a logical formalism for reasoning about action and change that, like TAL, uses an explicit time line and an *Occurs* predicate to link action occurrences to the time line. This makes it suitable for the generation of partially ordered plans and, as with PARADOCS, practical planning is accomplished using logic programs. However, plan synthesis is based on abduction of *Occurs* instead of deduction with reified action occurrences. The use of abduction results in theoretical elegance, but performing abduction in a logic program requires an abductive meta-interpreter, which makes the programs more complex.

The timepoint relations of the abductive Event Calculus planner can only detect inconsistencies caused by conflicts between action preconditions and action effects if the order of timepoints involved in the conflict is known. Consistency is guaranteed by adding new timepoint relations in what could be described as the logical equivalent of the promote/demote strategy from partial order planning algorithms (page 12, [33]). But resolving potential conflicts through promotion or demotion represents an early commitment to a specific ordering, and both orderings must be evaluated in cases where the planning process backtracks. Our constraint logic programs take full advantage of the representational power of disjunctive interval algebra relations. Disjunctive non-overlap constraints allow the planner to postpone decisions on the ordering of actions, even in the case of a possible conflict, thereby reducing the search space. We have in fact experimented with weaker constraint solvers similar to simple temporal networks, but found the added complexity of the implementation that result from the introduction of promotion and demotion to detract from the clarity of presentation. Although the use of disjunctive constraints is associated with an increase in the computational complexity of the constraint propagation algorithms, the complexity is still overshadowed by the complexity of planning in general.

In a feature comparison, the Event Calculus planner extends our basic planning capabilities with hierarchical planning and a form of knowledge producing sensing actions that are also based on abduction [34].

Another *deductive* planning framework is GOLOG [21], which extends the Situation Calculus with composite actions, including the sequences, conditionals, non-deterministic choices, and loops that we introduced in Chapter 5. The paradigm is slightly different in that it views the resulting formalism as a high-level agent *programming language* where the programmer supplies a possibly incomplete program specification, and the robot's task is to execute the specification while resolving non-determinism that may result from its incompleteness.

The Situation Calculus originated the idea of reified action occurrences passed around using an extra predicate argument. But its situation terms contain linearly ordered action sequences without explicit temporal information, which prevents the generation of partially ordered plans. Note however, that such shortcomings can be overcome through various extensions, as is done e.g. in ConGolog [11]. Both the Situation Calculus and the GOLOG framework have been extended in a number of other interesting ways, e.g. for planning with incomplete information about the initial state [9].

While GOLOG uses composite actions to provide a form of search control through domain dependant knowledge, complex actions are not part of the planning process in the way that simple actions are. This is no great surprise since, as we noted in Section 5.4, the resulting system would in fact perform program synthesis. However, Levesque's KPLANNER [20] explores an interesting middleground between the direct synthesis of programs with loops and GOLOG's execution of non-deterministic loops supplied by a programmer. It is applicable to problems where it is possible to identify a single fluent, whose value is unknown or unbound, that is responsible for making the problem unsolvable without the generation of plans containing loops. The method consists of setting a relatively small upper bound on the value of this fluent, generating a plan with help from a clever Prolog function that "winds up" action sequences into loops, and finally testing this plan for a larger bound (since testing will be cheaper than generating). Using this method one can generate plans with loops that are correct up to the testing bound, and, for some classes of problems, provably correct for any value of the unbounded fluent.

Another approach to the use of composite actions in planning is presented by McIlraith and Fadel [27] who were motivated by the needs of planning for the use of semantic web services, but also by the increased plan generation efficiency resulting from the possibility of shorter solution plans. They provide a compilation, expressed in terms of the Situation Calculus, of action sequences, conditionals, and loops, into a new set of actions that are not composite. However, due to the requirements enforced on the composite actions to ensure that the compilation is possible, only bounded loops with a predetermined maximum number of iterations are considered. Our PARADOCS framework does not provide a means for the direct use of composite actions in planning, but our loop semantics, expressed in terms of a fixpoint formula, is not associated with any preset bounds.

The Fluent Calculus serves as the formal basis for FLUX [37], another logic programming methodology that supports deductive planning with linear plans. It was mentioned earlier as the source of the Prolog list membership function implementation used for adding reified action occurrence terms to a list. In FLUX the same predicate is used for a different purpose as the basic mechanism of representing states as sets of fluents. Constraint handling rules are also used in FLUX, but again for different purposes than ours. While we utilize CHR for detecting and resolving potential conflicts between action effects and fluent persistency, in FLUX they enable the planning and reasoning with incomplete information and also reasoning about knowledge and knowledge updates as caused by sensing actions. Although the form of the expressions over which such reasoning is allowed is limited, the limitations ensure that the allowed reasoning is computationally efficient.

Our final example of logic programming based planning is answer set planning, as described e.g. by Lifschitz [23]. Answer set programming is purely declarative, and thus avoids some of the problems inherent in Prolog, such as the possibility of the evaluation becoming stuck in an infinite loop. This property makes it easier to add temporal constraints and state constraints on the answer generation. Lifschitz proposes that its use of both classical negation and negation as failure makes it suitable for specifying the effects of actions and the non-monotonic behaviour of the persistence of fluents that they affect.

In addition, Son, Baral, and McIlraith [35] show how answer set programming can be extended with sequences, conditionals, non-deterministic choice, and loops, to construct an alternative GOLOG interpreter. This can be used, in the same way as any Prolog GOLOG interpreter, to write non-deterministic programs that, in effect, express domain-specific control knowledge restricting the search for solution plans. However, the evaluation mechanism of answer set programs includes instantiating the problem into a finite number of propositions, and is thus only applicable to finite theories with a fixed upper time bound. While such a "propositional" approach, also used in the planning as satisfiability paradigm, may be simple and robust, it can suffer from an exponential increase in the size of the representation of the problem. It also lacks the goal-directedness of Prolog's evaluation mechanism, based on theorem proving, that seems to us to be a desireable property that will be absolutely necessary as the complexity of problems a robot attempts to solve increases, and as the number of different means it has available to apply to the problems grows.

An even stronger programming language focus is displayed by agent programming languages such as 3APL [14]. The concept of a goal is considered to be procedural rather than declarative, and actions are described as state updates that modify an agent's beliefs, rather than being defined by sets of axioms. While 3APL provides a formal semantics, it is an operational semantics distinct from the programming language itself. Although the procedural view of goals might seem restrictive, from the point of view of planning, it is surprisingly close to e.g. GOLOG's programs. In fact, it is possible to embed ConGolog *into* 3APL [13].

Another system with a focus on planning, and with many features in common with PARADOCS, is Allen's temporal planner [2]. It expresses temporal information using the interval temporal logic that axiomatizes the interval algebra and extends it with atomic time periods that behave like timepoints and represents actions by reified events. Allen also stresses the importance of viewing different types of reasoning as inference in a common representation, although, the actual plan synthesis is cast as an algorithm that closely approximates a proof procedure.

A different view is endorsed in planners that utilize special purpose planning algorithms. TALplanner [7] is one such planner that is especially close to ours in that it makes use of the same Temporal Action Logic. However, the purpose of TAL in TALplanner is as a formal semantics of actions, goals, and plans. Employing Temporal Action Logic as a formal semantics for the actual *planning process*, and using it directly for plan synthesis, prompted the extensions and work described in this thesis.

Karlsson proceeds with a more theoretical emphasis when he presents his Narrative Logic, based on TAL but with its own extensions for reified action occurrences [16]. Karlsson formulates the planning task in Narrative Logic but, although he provides a translation from Narrative Logic into classical logic, the result is a second-order theory that is not directly amenable to automated theorem proving. In contrast, our translation of TAL with reified action occurrences into logic programs makes the logic directly applicable in practice.

Both Karlsson's work and ours extend TAL, which itself originates in Sandewall's *Features and Fluents* framework [32]. In fact, Sandewall himself provides a formulation of composite actions, including sequences, conditionals, and loops, whose macro-language syntax is almost identical to ours. However, the translation of loops differs significantly. Sandewall uses a first-order definition involving special *signal features*, which are introduced only to keep track of the looping behaviour. This results in a specification that is rather complex, whereas our formulation in terms of a fixpoint formula is significantly simpler while succinctly capturing the looping property.

Furthermore, we extend Sandewall's set of composite actions with nondeterministic choice, which we found necessary for composite actions to be of practical use, and we make such use possible through the provision of our compilation into logic programs.
# Chapter 8

## Discussion

Although Temporal Action Logic has long served as a formal basis for some of our work in cognitive robotics, we set out to use TAL for deductive planning. This effort necessitated some theoretical additions to the logic but has also resulted in a compilation process that extends the translation from the high-level language  $\mathcal{L}(ND)$  to constraint logic programs that are efficiently executed for practical planning. The persistence constraints that are part of the generated plans can be used for execution monitoring purposes and the integration of plan synthesis and reasoning about plans is put to use in the failure recovery process. The entire set up is integrated with our UAV system and can be used through a graphical user interface with drag'n'drop mission planning to execute logistics missions in our helicopter simulator that adds a winch and crates to our fully operational UAV research platform.

The explicit time representation of TAL exposes qualitative and quantitative temporal primitives that are particularly amenable to reasoning using temporal constraint formalisms. The method can be made to work with a variety of constraint formalisms for both qualitative and quantitative constraints, such as simple timepoint constraints, interval algebra, or general temporal constraint networks. But the combination of constraint handling rules and disjunctive temporal constraints is particularly suited to the TAL formalism. This, together with the occlusion concept, enables a novel solution to potential action/persistence conflict threats where promotion/demotion is postponed, thereby removing search space choice points and resulting in flexible solution plans.

## 8.1 Future Research

While compiling narratives into logic programs that can be used for deductive planning works very well, the technique is ultimately limited. As stated in the introduction Section 1.1, our two main motivations for planning deductively are uniformity and expressiveness. But the expressivity of logic programs are inherently limited, and each attempt to circumvent specific limitations using additional special purpose techniques will result in decreasing uniformity. For these reasons we believe a move towards a more direct use of theorem proving in the first-order base logic of TAL to be important.

A move from logic programming towards more general theorem proving will surely lead to inefficiency and scalability problems. The resulting system might only be able to reason with relatively small problem instances and complexity might prevent it from scaling up to problem instances of a larger size that could have been solved by a less expressive system. Though, the class of problems to which the system is applicable would be larger. And when that class expands to encompass the generation of plans with loops or recursion, as discussed in Section 5.4, the rules of the game suddenly change. Consider e.g. a logistics scenario where the number of crates to be delivered is unspecified. Any solution plan will have to contain some form of looping or recursion behaviour that iterates over the crates until the goal has been satisfied. Plans of this highly expressive sort can probably be small and compact. Moreover, the plan is of constant size, *regardless* of the size of the logistics problem instance that it is later applied to. Finally, the solution simultaneously solves all problem instantiations with any number of crates. In contrast, a less expressive planner would not be applicable to a logistics problem where it is unknown what crates exists. It would exhibit an often exponential, but at least linear, increase in planning time and a linear increase in solution length as the size of the problem instance grows. Finally, it would require more planning and the generation of a new plan every time a new problem instance with a different number of crates needs to be solved.

The problem of generating plans with loops or recursion is a very difficult problem to which deductive planning with theorem proving techniques has been applied with some success [26, 3, 4]. In addition to the "plans as programs" paradigm, deductive planning with a highly expressive representation is potentially applicable to a large number of other interesting problems. Two of the most important are planning in the context of incomplete information and reasoning about knowledge and knowledge-producing sensing actions. Both are possible continuations of the work in this thesis.

## 8.2 Conclusions

Nilsson [29] argues for a distinction between general intelligent systems and specialized systems that exhibit greater than human performance in a relatively narrow area of expertise. A logical framework for reasoning and planning with actions is, by design, aimed towards generality rather than performance when compared with a special purpose planner. This thesis presents a methodology for practical deductive planning in TAL that is a step towards a uniform way of tackling increasing complexity. Moving towards increasingly intelligent autonomous systems means requiring less help from humans who predict and decide what tools need to be applied in solving a given task. At the same time, the increasing diversity and challenge of the tasks themselves require the application of flexible and powerful reasoning methods. By reformulating special-purpose algorithms as proof search in a shared representational formalism, one removes the need to decide in advance what algorithms are needed to solve future tasks. Through the use of highly expressive logical formalisms and automated theorem proving technology one opens the possibility of attacking a great many different and complex reasoning problems in a uniform way. Although this unifying approach might not have been the way of the (recent) past, the growing challenges of artificial intelligence applications is increasing its appeal as the way of the future.

## Appendix A

## The Logistics Scenario Prolog Code

This appendix contains the UAV logistics scenario constraint logic program encoding that has been integrated with the autonomous helicopter system and was used for the planning examples in Chapter 6. Readers interested in running the program can also access this code in electronic form at this thesis' website: http://www.martinmagnusson.com/paradocs/thesis/.

Three additional comments will assist understanding of the code. Firstly, the persistence clauses have a potential problem with infinite looping given the depth-first search strategy of Prolog. A fluent is true at a timepoint if it is true at an earlier timepoint and is not occluded during the interval in between. It is true at the earlier timepoint if it is true at an even earlier timepoint, and so on. The looping is correct but unwanted behaviour. But the observation that if a fluent is persistent over two intervals that meet, then it must be persistent over the union of the intervals, provides the key to a simple solution. Without loss of generality we prohibit two consecutive persistence intervals using an additional constraint that appears in the code below with the comment "Consecutive persistence constraint".

Secondly, although there is only one UAV in our scenario there are few restrictions on the occurrence times of actions, which can lead to modelling problems related to concurrency. The UAV magnetic hook can be seen as a limited resource that can not attach several crates at once. However, modelling this using a fluent precondition *free* that is set to false when executing *attach* does not prevent two concurrent *attach* at the *exact* same time. Actions can only have effects in the future, so one *attach* can not prevent another *attach*  from executing simultaneously by setting *free* to *false* at the next timepoint. Gustafsson [12] suggests a way of solving this problem using TAL narratives where actions do not have direct effects on fluents, but rather activate special *influence* fluents. Dependency constraints are then used to express influence laws that govern the effects of influences on the environment, and thereby the indirect effects of actions. Concurrent planning in TAL, using an explicit time line and powerful temporal constraints solvers, holds great promise but clearly needs more investigation. We avoid the issue for now and instead add two concurrency constraints even though they are not part of the translation specified in Chapter 4. Specifically, the constraints with the comment "Concurrency constraints" prohibit the UAV from carry several crates and from flying to several places concurrently.

Thirdly, and finally, at the same time that non-overlap constraints are added to the finite domain constraint store, they are stored in the CHR store using a constraint named *ia*. The final group of clauses, which appear below the comment "Backup store", are then used to create a copy of the CHR constraint store before the labeling of timepoint variables that determines if the interval algebra network is consistent. This prevents the labeling process from instantiating and unifying timepoint variables in the network copy while searching for a consistent linearization of the original, thereby preserving all of the partialness of the ordering of action occurrences. The final interval algebra graph is then easily read from the *ia* constraint in the constraint store copy, and it is guaranteed to be consistent since the original went through a labeling.

```
:- use_module(library(chr)).
:- use_module(library(clpfd)).
:- use_module(library(charsio)).
% CHR definitions:
handler tal.
constraints not_occlude/3, circ_occurs/3, tp/1, ia/3.
% Occurs definition:
occurs(T1,T2,A,P) :-
member(occ(T1,T2,A),P), circ_occurs(T1,T2,A).
member(A,[A|Pp]).
member(A,P) :-
nonvar(P), P = [A1|P1], A \== A1, member(A,P1).
% Interval relations:
not_overlap(Xs,Xe,Ys,Ye) :-
```

```
Xe \#=\langle Ys \# \rangle / Xs \# \geq Ye,
  ia(Xs-Xe,[<,m,mi,>],Ys-Ye).
% Type clauses:
timepoint(T) := domain([T],0,100), tp(T).
uav(uav1).
crate(crate1).
crate(crate2).
crate(crate3).
% Persistence formulas:
holds(T2,loc(X),V,P) :-
  timepoint(T1), timepoint(T2), T1 #< T2,</pre>
  not_occlude(T1,T2,loc(X)),
  holds(T1,loc(X),V,P).
holds(T2,carrying(U,C),V,P) :-
  timepoint(T1), timepoint(T2), T1 #< T2,</pre>
  not_occlude(T1,T2,carrying(U,C)),
  holds(T1,carrying(U,C),V,P).
% Consecutive persistence constraint:
not_occlude(T1,T,F), not_occlude(T,T2,F) ==> fail.
% Initial state (example):
holds(0,loc(uav1),pos(237,-23),P).
holds(0,loc(crate1),pos(202,0),P).
holds(0,loc(crate2),pos(210,-10),P).
holds(0,loc(crate3),pos(200,-13),P).
% Action specifications:
holds(T2,loc(U),L,P) :-
  uav(U), timepoint(T1), timepoint(T2), T1 #< T2,</pre>
  occurs(T1,T2,fly(U,L),P).
not_occlude(T1,T2,loc(U)),
  circ_occurs(T3,T4,fly(U,L)) ==>
    not_overlap(T1,T2,T3,T4).
holds(T2,carrying(U,C),true,P) :-
  uav(U), crate(C), timepoint(T1), timepoint(T2), T1 #< T2,</pre>
  occurs(T1,T2,attach(U,C),P),
```

```
holds(T1,loc(C),L,P),
  holds(T1,loc(U),L,P),
  not_occlude(T1,T2,loc(U)).
not_occlude(T1,T2,carrying(U,C)),
  circ_occurs(T3,T4,attach(U,C)) ==>
    not_overlap(T1,T2,T3,T4).
not_occlude(T1,T2,loc(C)),
  circ_occurs(T3,T4,attach(U,C)) ==>
    not_overlap(T1,T2,T3,T4).
holds(T2,loc(C),L,P) :-
  uav(U), crate(C), timepoint(T1), timepoint(T2), T1 #< T2,</pre>
  occurs(T1,T2,drop(U,C),P),
  holds(T1,carrying(U,C),true,P),
  holds(T1,loc(U),L,P),
  not_occlude(T1,T2,loc(U)).
not_occlude(T1,T2,carrying(U,C)),
  circ_occurs(T3,T4,drop(U,C)) ==>
    not_overlap(T1,T2,T3,T4).
not_occlude(T1,T2,loc(C)),
  circ_occurs(T3,T4,drop(U,C)) ==>
    not_overlap(T1,T2,T3,T4).
% Concurrency constraints:
not_occlude(T1,T2,carrying(U,C1)),
  not_occlude(T3,T4,carrying(U,C2)) ==>
    not_overlap(T1,T2,T3,T4).
circ_occurs(T1,T2,fly(U,L1)),
  circ_occurs(T3,T4,fly(U,L2)) ==>
    not_overlap(T1,T2,T3,T4).
% Remove redundant constraints:
not_occlude(T1,T2,F) \ not_occlude(T1,T2,F) <=> true.
circ_occurs(T1,T2,A) \ circ_occurs(T1,T2,A) <=> true.
tp(X) \setminus tp(X) \iff true.
tp(X) <=> ground(X) | true.
% Find variables to label:
timepoint_variables(L) :-
  findall_constraints(tp(_),L1),
```

```
extract_variable(L1,L).
extract_variable([],[]).
extract_variable([tp(T)#_|L1],[T|L2]) :-
extract_variable(L1,L2).
% Backup store:
copy_constraints(C) :-
findall_constraints(_,L1),
extract_constraint(L1,L),
format_to_chars('~w',[L],C).
extract_constraint([],[]).
extract_constraint([C#_|L1],[C|L2]) :-
extract_constraint(L1,L2).
```

## Appendix B

## Proofs

This appendix collects proofs of some theorems used in the thesis.

## **B.1** Interval Persistence Formula

The interval persistence formula relates regular, single timepoint, occlusion with the new interval occlusion.

#### Theorem

$$\begin{array}{l} \forall t \left[\neg Occlude(t+1,f) \rightarrow \forall v \left[ Holds(t,f,v) \leftrightarrow Holds(t+1,f,v) \right] \right] \land \\ \forall t_1,t_2,f \left[ Occlude(t_1,t_2,f) \leftrightarrow \exists t \left[ t_1 < t \leq t_2 \land Occlude(t,f) \right] \right] \rightarrow \\ \forall t_1,t_2,f \left[\neg Occlude(t_1,t_2,f) \rightarrow \\ \forall t \left[ t_1 < t \leq t_2 \rightarrow \forall v \left[ Holds(t-1,f,v) \leftrightarrow Holds(t,f,v) \right] \right] \right] \end{array}$$

#### Proof

Given  $\neg Occlude(t_1, t_2, f)$  we know, by the definition of interval occlusion above, that there does not exist a timepoint t during the interval  $(t_1, t_2]$  at which Occlude(t, f) holds, i.e.,  $\neg Occlude(t, f)$  must hold for all timepoints in the interval. The persistence formula then forces the fluent to retain its value from the previous timepoint, which is specified in the above theorem by stating that the *Holds* predicate retains its truth value for all possible fluent values v. We formalize this reasoning in a deductive proof below.

$$1 \quad \forall t \left[\neg Occlude(t+1, f) \rightarrow \forall v \left[Holds(t, f, v) \leftrightarrow Holds(t+1, f, v)\right]\right] \qquad P$$

$$2 \quad \forall t_1, t_2, f \left[ Occlude(t_1, t_2, f) \leftrightarrow \exists t \left[ t_1 < t \le t_2 \land Occlude(t, f) \right] \right] \qquad P$$

$$3 \quad \neg Occlude(t_1, t_2, f) \to \neg \exists t \left[ t_1 < t \le t_2 \land Occlude(t, f) \right]$$

## B.2 Interval End-point Equivalence

The following proof validates the useful property of interval occlusion that a fluent that is known not to be occluded during an interval will have the same value at both end-points of that interval, regardless of the number of timepoints inbetween.

### Theorem

$$\begin{array}{l} \forall t_1, t_2, f \left[ \neg \mathit{Occlude}(t_1, t_2, f) \rightarrow \\ \forall t \left[ t_1 < t \leq t_2 \rightarrow \forall v \left[ \mathit{Holds}(t-1, f, v) \leftrightarrow \mathit{Holds}(t, f, v) \right] \right] \right] \land \\ \forall t_1, t_2, f \left[ \mathit{Occlude}(t_1, t_2, f) \leftrightarrow \exists t \left[ t_1 < t \leq t_2 \land \mathit{Occlude}(t, f) \right] \right] \rightarrow \\ \forall t_1, t_2, f \left[ \neg \mathit{Occlude}(t_1, t_2, f) \rightarrow \forall v \left[ \mathit{Holds}(t_1, f, v) \leftrightarrow \mathit{Holds}(t_2, f, v) \right] \right] \end{array}$$

#### Proof

The proof is by induction over the length of occlusion intervals. The base case consists of intervals of length 1. Such intervals are already covered by the sub-formula  $\forall v [Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)]$  in the interval persistence formula above, which we make use of in the deduction below.

$$\begin{array}{ll} 16 \quad \forall t_1, t_2, f \left[\neg Occlude(t_1, t_2, f) \rightarrow \\ \forall t \left[t_1 < t \leq t_2 \rightarrow \forall v \left[Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)\right]\right] \right] \end{array} P$$

$$17 \quad \Box \neg Occlude(t_1, t_2, f)$$
 H

$$18 \quad \forall t \ [t_1 < t \le t_2 \rightarrow \forall v \ [Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)]]$$

19 
$$t_2 = t_1 + 1$$
 the interval has length 1  
20  $\forall t [t_1 < t < t_1 + 1]$   $\forall x [H_0] d_0(t_1 - 1, f_1 x) + H_0] d_0(t_1 f_1 x) ]$  18 10

$$20 \quad \forall t \ [t_1 < t \le t_1 + 1 \to \forall v \ [Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)]]$$

$$18, 19$$

21 
$$t_1 < t_1 + 1 \le t_1 + 1 \to \\ \forall v [Holds((t_1 + 1) - 1, f, v) \leftrightarrow Holds(t_1 + 1, f, v)]$$
 20

22 
$$\forall v [Holds(t_1, f, v) \leftrightarrow Holds(t_1 + 1, f, v)]$$
 F, 21

23 
$$\forall v [Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v)]$$
 19,22

$$\begin{array}{ll} 24 \quad \forall t_1, t_2, f \left[\neg Occlude(t_1, t_2, f) \rightarrow \\ \forall v \left[Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v)\right]\right] \end{array}$$

In the inductive case we assume that the theorem holds for intervals of length n and show that it holds for intervals of length n + 1. First note that, by the definition of interval occlusion, if the fluent is interval occluded over an interval (t, t + n + 1], of length n + 1, then it must also be interval occluded over the sub-intervals (t, t + n] and (t + n, t + n + 1], of length n and 1 respectively. Since we assumed that the theorem holds for intervals of length n, the fluent will have the same value at t and t + n. As with the interval of length 1 in the base case, the interval persistence formula forces the fluent to retain its value between t + n and t + n + 1. Taken together these two conditions establish the conclusion that the fluent will necessarily have the same value at both endpoints of intervals of length n + 1. This is formalized, somewhat cumbersomely, in the deduction below.

28 
$$t_2 = t_1 + n + 1$$
 the interval has length  $n + 1$ 

$$29 \quad \neg Occlude(t_1, t_2, f) \to \neg \exists t \, [t_1 < t \le t_2 \land Occlude(t, f)]$$

$$30 \quad | \quad \neg \exists t \left[ t_1 < t \le t_2 \land Occlude(t, f) \right]$$

$$31 \quad \forall t \left[ t_1 < t \le t_2 \to \neg Occlude(t, f) \right]$$

$$\begin{array}{c} 1 & t & = 1 \\ 34 \\ 1 & t & = 1 \\ 1 & t & \leq t_2 \\ 0 & t & t & (t - t) \\ 0 & t & t & t \\ 0 & t & t &$$

$$36 \mid t_1 < t \le t_1 + n \to \neg Occlude(t, f)$$

$$32 - 35$$

$\forall t \left[ t_1 < t \leq t_1 + n \rightarrow \neg Occlude(t, f) \right]$	36
$\neg \exists t  [t_1 < t \le t_1 + n \land Occlude(t, f)]$	37
$\neg Occlude(t_1, t_1 + n, f)$	26, 38
$\forall v \left[ Holds(t_1, f, v) \leftrightarrow Holds(t_1 + n, f, v) \right]$	39, inductive assumption
	H
$t_1 < t \le t_1 + n + 1$	41, F
$t_1 < t \le t_2$	28, 42
$\neg Occlude(t, f)$	31, 43
$t_1 + n < t \le t_1 + n + 1 \to \neg Occlude(t, f)$	41 - 44
$\forall t [t_1 + n < t \le t_1 + n + 1 \rightarrow \neg Occlude(t, f)]$	45
$\neg \exists t \left[ t_1 + n < t \le t_1 + n + 1 \land Occlude(t, f) \right]$	46
$\neg Occlude(t_1+n,t_1+n+1,f)$	26,47
$\forall t [t_1 + n < t \leq t_1 + n + 1 \rightarrow$	
$\forall v [Holds(t-1, f, v) \leftrightarrow Holds(t, f, v)]]$	25, 48
$t_1 + n < t_1 + n + 1 \le t_1 + n + 1$	F
$\forall v [Holds(t_1 + n, f, v) \leftrightarrow Holds(t_1 + n + 1, f, v)]$	49,50
$\forall v \left[ Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v) \right]$	28, 40, 51
$\forall t_1, t_2, f [\neg Occlude(t_1, t_2, f)] \rightarrow$	
$\forall v \left[ Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v) \right] \right]$	27 - 52
	$ \begin{array}{l} \forall t \left[t_1 < t \leq t_1 + n \rightarrow \neg Occlude(t, f)\right] \\ \neg \exists t \left[t_1 < t \leq t_1 + n \wedge Occlude(t, f)\right] \\ \neg Occlude(t_1, t_1 + n, f) \\ \forall v \left[Holds(t_1, f, v) \leftrightarrow Holds(t_1 + n, f, v)\right] \\ \hline t_1 + n < t \leq t_1 + n + 1 \\ t_1 < t \leq t_2 \\ \neg Occlude(t, f) \\ \hline t_1 + n < t \leq t_1 + n + 1 \rightarrow \neg Occlude(t, f) \\ \forall t \left[t_1 + n < t \leq t_1 + n + 1 \rightarrow \neg Occlude(t, f)\right] \\ \neg \exists t \left[t_1 + n < t \leq t_1 + n + 1 \wedge Occlude(t, f)\right] \\ \neg \exists t \left[t_1 + n < t \leq t_1 + n + 1, f\right) \\ \forall t \left[t_1 + n < t \leq t_1 + n + 1, f\right) \\ \forall t \left[t_1 + n < t \leq t_1 + n + 1, f\right) \\ \forall t \left[t_1 + n < t \leq t_1 + n + 1, f\right) \\ \forall t \left[t_1 + n < t \leq t_1 + n + 1, f\right) \\ \forall t \left[Holds(t - 1, f, v) \leftrightarrow Holds(t, f, v)\right] \\ t_1 + n < t_1 + n + 1 \leq t_1 + n + 1 \\ \forall v \left[Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v)\right] \\ \forall t_1, t_2, f \left[\neg Occlude(t_1, t_2, f) \rightarrow \\ \forall v \left[Holds(t_1, f, v) \leftrightarrow Holds(t_2, f, v)\right] \\ \end{array} \right]$

By the principle of mathematical induction, the theorem holds for intervals of any length.

## **B.3** Point-interval Rewrite

This particular equivalence-preserving rewrite is used in the compilation of a TAL theory to replace timepoint occlusion formulas with equivalent formulas expressed using interval occlusion.

### Theorem

$$\begin{aligned} &\forall t \left[ P(t) \to Q(t) \right] \leftrightarrow \\ &\forall t_1, t_2 \left[ \exists t \left[ t_1 < t \leq t_2 \land P(t) \right] \to \exists t \left[ t_1 < t \leq t_2 \land Q(t) \right] \right] \end{aligned}$$

### Proof

We start with the  $\Rightarrow$  direction and assume that the left hand side holds. Assume further that there is some t in the interval  $(t_1, t_2]$  for which P holds, i.e.,

5

71

 $\exists t \ [t_1 < t \leq t_2 \land P(t)]$ . Apply the assumed implication  $P(t) \to Q(t)$  to this t to show the existence of a timepoint in the interval  $(t_1, t_2]$  for which Q holds, i.e.,  $\exists t \ [t_1 < t \leq t_2 \land Q(t)]$ . For the  $\Leftarrow$  direction, assume that the right hand side holds. Assume further that P(t') holds, for an arbitrary timepoint t'. Instantiate the universally quantified  $t_1$  and  $t_2$ , in the right hand side of the theorem, to t' - 1 and t' respectively. Since t' is in the interval (t' - 1, t'], and we assumed P(t'), we have  $\exists t \ [t' - 1 < t \leq t' \land P(t)]$  and conclude  $\exists t \ [t' - 1 < t \leq t' \land Q(t)]$ . The narrow interval permits only one value for the existentially quantified timepoint t, namely t', and we thus have Q(t'). Since t' was arbitrary, we conclude that  $\forall t \ [P(t) \to Q(t)]$ . The argument is restated formally below.

54	$  \forall t \left[ P(t) \to Q(t) \right] $	Н
55	$\Box \exists t [t_1 < t \le t_2 \land P(t)]$	Н
56	$    -t_1 < t < t_2 \land P(t) $	Н

$$\begin{array}{c|c} 6\\7\\ \end{array} & \begin{bmatrix} t_1 < t \le t_2 \land P(t) \\ P(t) \to Q(t) \\ \end{array} & \begin{array}{c} H\\54 \\ \end{array}$$

$$60 \qquad \qquad \exists t [t_1 < t \le t_2 \land Q(t)] \tag{59}$$

61 
$$\exists t [t_1 < t \le t_2 \land Q(t)]$$
 55,56 - 60

$$62 \qquad \forall t_1, t_2 \left[ \exists t \left[ t_1 < t \le t_2 \land P(t) \right] \to \exists t \left[ t_1 < t \le t_2 \land Q(t) \right] \right] \qquad 55 - 61$$

72 
$$Q(t')$$
 68,69 - 71

$$73 \quad \boxed{\forall t \left[ P(t) \to Q(t) \right]}$$

$$74 \quad \forall t \left[ P(t) \to Q(t) \right] \leftrightarrow$$

$$64 - 72$$

$$\forall t_1, t_2 \left[ \exists t \ [t_1 < t \le t_2 \land P(t)] \to \exists t \ [t_1 < t \le t_2 \land Q(t)] \right] \qquad 54 - 62, 63 - 73$$

## **B.4** Shared Timepoint Overlap Equivalence

This proof relates a formula asserting the existence of a timepoint that is shared between two intervals, and a simpler timepoint relation that indicates interval overlap.

### Theorem

 $\exists t \left[ t_1 < t \leq t_2 \land t_3 < t \leq t_4 \right] \leftrightarrow t_2 > t_3 \land t_1 < t_4$ 

### Proof

Negating both hand sides provides an alternative formulation of the theorem in terms of *non-overlap*:

$$\neg \exists t \ [t_1 < t \le t_2 \land t_3 < t \le t_4] \leftrightarrow t_2 < t_3 \lor t_2 = t_3 \lor t_1 = t_4 \lor t_1 > t_4$$

Of Allen's 13 primitive interval relations [1], only the four cases displayed in Figure B.1 below share no point in common between the two intervals  $(t_1, t_2]$  and  $(t_3, t_4]$ , as required by the left hand side of the theorem. As can be seen by inspection, these cases correspond exactly to the disjuncts on the right hand side of the theorem.



Figure B.1: The four primitive interval relations where the two intervals share no common timepoint.

An alternative, formal deductive, proof is provided below.

75	$\Box \exists t  [t_1 < t \le t_2 \land t_3 < t \le t_4]$	H
76	$-t_1 < t \le t_2 \land t_3 < t \le t_4$	Н
77	$t_3 < t \land t \le t_2$	76
78	$t_2 > t_3$	77, F
79	$t_1 < t \land t \le t_4$	76
80	$t_1 < t_4$	79, F
81	$t_2 > t_3 \wedge t_1 < t_4$	78,80
82	$t_2 > t_3 \wedge t_1 < t_4$	75,76-81

83	$ t_2 > t_3 \wedge t_1 < t_4 $	H
84	$t_4 \le t_2 \lor t_4 > t_2$	F
85	$-t_4 \leq t_2$	H
86	$t_1 < t_4 \le t_2$	83, 85
87	$t_3 < t_4 \le t_4$	$t_3$ and $t_4$ are endpoints of an interval
88	$\exists t \ [t_1 < t \le t_2 \land t_3 < t \le t_4]$	86,87
89	$-t_4 > t_2$	Н
90	$t_1 < t_2 \le t_2$	$t_1$ and $t_2$ are endpoints of an interval
91	$t_3 < t_2 < t_4$	83,89
92	$t_3 < t_2 \le t_4$	91, F
93	$\exists t \ [t_1 < t \le t_2 \land t_3 < t \le t_4]$	90, 92
94	$\exists t \ [t_1 < t \le t_2 \land t_3 < t \le t_4]$	84,85 - 88,89 - 93
95	$\exists t \left[ t_1 < t \le t_2 \land t_3 < t \le t_4 \right] \leftrightarrow t_2 >$	$t_3 \wedge t_1 < t_4 \qquad \qquad 75 - 82,83 - 94$

## Bibliography

- James F. Allen. Maintaining knowledge about temporal intervals. Communications of the ACM, 26(11):832–843, 1983.
- [2] James F. Allen. Planning as temporal reasoning. In James F. Allen, Richard Fikes, and Erik Sandewall, editors, *Principles of Knowledge Rep*resentation and Reasoning (KR'91), pages 3–14. Morgan Kaufmann, San Mateo, California, 1991.
- [3] Stephen Cresswell, Alan Smaill, and Julian Richardson. Deductive synthesis of recursive plans in linear logic. In *Proceedings of the 5th European Conference on Planning (ECP'99)*, pages 252–264, London, UK, 2000. Springer-Verlag.
- [4] Lucas Dixon, Alan Smaill, and Alan Bundy. Planning as deductive synthesis in intuitionistic linear logic. Technical report, The University of Edinburgh Shool of informatics, 2006. http://homepages.inf.ed.ac. uk/ldixon/papers/infrep-06-planill.pdf.
- [5] Patrick Doherty. PMON<sup>+</sup>: a fluent logic for action and change, formal specification, version 1.0. Technical Report LiTH-IDA-R-96-33, Department of Computer and Information Science, Linköping University, 1996. http://www.ida.liu.se/publications/techrep/96/tr96.html.
- [6] Patrick Doherty, Patrik Haslum, Fredrik Heintz, Torsten Merz, Per Nyblom, Tommy Persson, and Björn Wingman. A distributed architecture for autonomous unmanned aerial vehicle experimentation. In *Proceedings* of the 7th International Symposium on Distributed Autonomous Robotic Systems, 2004.
- [7] Patrick Doherty and Jonas Kvarnström. TALplanner: A temporal logic based planner. AI Magazine, 22(3):95–102, 2001.

- [8] Patrick Doherty and Jonas Kvarnström. Handbook of Knowledge Representation, chapter 18. Elsevier, 2007. To appear.
- [9] Alberto Finzi, Fiora Pirri, and Ray Reiter. Open world planning in the situation calculus. In Proceedings of the 7th Conference on Artificial Intelligence (AAAI'00) and of the 12th Conference on Innovative Applications of Artificial Intelligence (IAAI'00), pages 754–760, Menlo Park, CA, July 30–3 2000. AAAI Press.
- [10] Thom Frühwirth. Theory and practice of constraint handling rules. Journal of Logic Programming, Special Issue on Constraint Logic Programming, 37(1-3):95–138, October 1998.
- [11] Giuseppe De Giacomo, Yves Lesperance, and Hector J. Levesque. Congolog, a concurrent programming language based on the situation calculus. *Artificial Intelligence*, 121(1-2):109–169, 2000.
- [12] Joakim Gustafsson. Extending Temporal Action Logic. PhD thesis, Linköping University, 2001. Dissertation No. 689.
- [13] Koen Hindriks, Yves Lespérance, and Hector Levesque. An embedding of ConGolog in 3APL. Technical Report UU-CS-2000-13, Department of Computer Science, University Utrecht, 2000.
- [14] Koen V. Hindriks, Frank S. De Boer, Wiebe Van der Hoek, and John-Jules Ch. Meyer. Agent programming in 3APL. Autonomous Agents and Multi-Agent Systems, 2(4):357–401, 1999.
- [15] Franz Inc. Graphical user interface tools. http://www.franz.com/ products/gui\_tools/. Visited November 2006.
- [16] Lars Karlsson. Anything can happen: On narratives and hypothetical reasoning. In Proceedings of the Sixth International Conference on the Principles of Knowledge Representation and Reasoning (KR'98), pages 36-47, 1998.
- [17] Lars Karlsson. Actions, Interactions and Narratives. PhD thesis, Linköping University, 1999. Dissertation No. 593.
- [18] Jonas Kvarnström. VITAL: Visualization and implementation of temporal action logics. http://www.ida.liu.se/~jonkv/vital/. Visited January 2007.
- [19] Jonas Kvarnström. TALplanner and Other Extensions to Temporal Action Logic. PhD thesis, Linköping University, 2005. Dissertation No. 937.

- [20] Hector J. Levesque. Planning with loops. In Proceedings of the Nineteenth International Joint Conference on Artificial Intelligence (IJCAI'05), pages 509–515, 2005.
- [21] Hector J. Levesque, Raymond Reiter, Yves Lesperance, Fangzhen Lin, and Richard B. Scherl. GOLOG: A logic programming language for dynamic domains. *Journal of Logic Programming*, 31(1-3):59–83, 1997.
- [22] Vladimir Lifschitz. Circumscription. In Handbook of Artificial Intelligence and Logic Programming, volume 3, pages 297 – 352. Oxford University Press, 1991.
- [23] Vladimir Lifschitz. Answer set planning. In Proceedings of the 1999 International Conference on Logic Programming, pages 23–37, 1999.
- [24] Martin Magnusson and Patrick Doherty. Deductive planning with temporal constraints using TAL. In Proceedings of the International Symposium on Practical Cognitive Agents and Robots (PCAR'06), pages 141– 152, 2006.
- [25] Martin Magnusson and Patrick Doherty. Deductive planning with temporal constraints. In Eyal Amir, Vladimir Lifschitz, and Rob Miller, editors, Logical Formalizations of Commonsense Reasoning: Papers from 2007 AAAI Spring Symposium. AAAI Press, 2007. Technical Report SS-07-05 http://www.ucl.ac.uk/commonsense07/papers/ magnusson-and-doherty.pdf.
- [26] Zohar Manna and Richard J. Waldinger. How to clear a block: A theory of plans. *Journal of Automated Reasoning*, 3(4):343–377, 1987.
- [27] Sheila McIlraith and Ronald Fadel. Planning with complex actions. In Proceedings of the Ninth International Workshop on Non-Monotonic Reasoning (NMR'02), pages 356–364, Toulouse, France, April 19-21 2002.
- [28] Itay Meiri. Combining qualitative and quantitative constraints in temporal reasoning. In Proceedings of the Ninth National Conference on Artificial Intelligence, pages 260–267, 1991.
- [29] Nils J. Nilsson. Eye on the prize. AI Magazine, 16(2):9–17, 1995.
- [30] Swedish Institute of Computer Science SICS AB. SICStus prolog homepage. http://www.sics.se/sicstus/. Visited November 2006.
- [31] Francis Jeffry Pelletier. A brief history of natural deduction. History and Philosophy of Logic, 20:1–31, 1999.

- [32] Erik Sandewall. Features and Fluents: The Representation of Knowledge about Dynamical Systems, volume 1. Oxford University Press, 1994.
- [33] Murray Shanahan. An abductive event calculus planner. Journal of Logic Programming, 44(1-3):207–240, 2000.
- [34] Murray Shanahan and Mark Witkowski. High-level robot control through logic. Lecture Notes in Computer Science, 1986, 2001.
- [35] Tran Cao Son, Chitta Baral, and Sheila McIlraith. Extending answer set planning with sequence, conditional, loop, non-deterministic choice, and procedure constructs. In Proceedings of the AAAI Spring Symposium on Answer Set Programming: Towards Efficient and Scalable Knowledge Representation and Reasoning, pages 202–209, Stanford, ca, USA, March 26-28 2001.
- [36] Michael Thielscher. FLUX webpage. http://www.fluxagent.org/. Visited December 2006.
- [37] Michael Thielscher. FLUX: A logic programming method for reasoning agents. Theory and Practice of Logic Programming, 5(4-5):533-565, 2005.
- [38] John Thornton, Matthew Beaumont, Abdul Sattar, and Michael J. Maher. A local search approach to modelling and solving interval algebra problems. *Journal of Logic and Computation*, 14(1):93–112, 2004.
- [39] Wikipedia. Presburger arithmetic wikipedia, the free encyclopedia, 2007. Visited January 2007.

	<b>Avd</b> Divis	elning, Institution sion, Department		Datum Date
Linköpings universitet	AII Dep 581	CS, t. of Computer and Inf 83 Linköping	ormation Science	2007-09-28
Språk Language		Rapporttyp Report category	<b>ISBN</b> 978-91-85895-9	3-9
□ Svenska/Swedish		⊠ Licentiatavhandling	ISRN	
$\boxtimes$ Engelska/English		$\Box$ Examensarbete	LiU-Tek-Lic-20	07:38
		C-uppsats D-uppsats	Serietitel och seri	enummer ISSN
□		□ Övrig rapport		<u>0280–7971</u>
URL för elektronisk version http://www.martinmagnusson.com/ nublications/magnusson-2007-lic.pdf		Linköping Studies Thesis No. 1329	in Science and Technology	

### $\mathbf{Titel}$

#### Title

Deductive Planning and Composite Actions in Temporal Action Logic

#### Författare

Author

Martin Magnusson

#### Sammanfattning

Abstract

Temporal Action Logic is a well established logical formalism for reasoning about action and change that has long been used as a formal specification language. Its first-order characterization and explicit time representation makes it a suitable target for automated theorem proving and the application of temporal constraint solvers. We introduce a translation from a subset of Temporal Action Logic to constraint logic programs that takes advantage of these characteristics to make the logic applicable, not just as a formal specification language, but in solving practical reasoning problems. Extensions are introduced that enable the generation of action sequences, thus paving the road for interesting applications in deductive planning. The use of qualitative temporal constraints makes it possible to follow a least commitment strategy and construct partially ordered plans. Furthermore, the logical language and logic program translation is extended with the notion of composite actions that can be used to formulate and execute scripted plans with conditional actions, non-deterministic choices, and loops. The resulting planner and reasoner is integrated with a graphical user interface in our autonomous helicopter research system and applied to logistics problems. Solution plans are synthesized together with monitoring constraints that trigger the generation of recovery actions in cases of execution failures.

Nyckelord Keywords Temporal Action Logic, deductive planning, composite actions, interval algebra, constraint logic programming, execution monitoring

## Department of Computer and Information Science Linköpings universitet

## Linköping Studies in Science and Technology Faculty of Arts and Sciences - Licentiate Theses

No 17	<b>Vojin Plavsic:</b> Interleaved Processing of Non-Numerical Data Stored on a Cyclic Memory. (Available at: FOA, Box 1165, S-581 11 Linköping, Sweden, FOA Report B30062E)
No 28	Arne Jönsson, Mikael Patel: An Interactive Flowcharting Technique for Communicating and Realizing Al- gorithms. 1984.
No 29	Johnny Eckerland: Retargeting of an Incremental Code Generator 1984
No 48	Henrik Nordin: On the Use of Typical Cases for Knowledge-Based Consultation and Teaching 1985
No 52	<b>Tabo Pana:</b> Stone Towards the Formalization of Designing VI SI Systems 1985
No 52	<b>Lehon Fenge</b> , steps flowards the Formanzation of Designing v List Systems, 1765.
No 71	John Fagerstrom, Simulation and Evaluation of Architecture based on Asynchronous Flocesses, 1985.
NO /1	Jaia Malek: ICONStraint, A Dependency Directed Constraint Maintenance System, 1987.
NO 72	Tony Larsson: On the Spectrication and Vernication of VLSI Systems, 1980.
No /3	Ola Stromfors: A Structure Editor for Documents and Programs, 1986.
No 74	<b>Christos Levcopoulos:</b> New Results about the Approximation Behavior of the Greedy Triangulation, 1986.
No 104	Shamsul I. Chowdhury: Statistical Expert Systems - a Special Application Area for Knowledge-Based Com- puter Methodology, 1987.
No 108	Rober Bilos: Incremental Scanning and Token-Based Editing, 1987.
No 111	Hans Block: SPORT-SORT Sorting Algorithms and Sport Tournaments, 1987.
No 113	<b>Ralph Rönnquist:</b> Network and Lattice Based Approaches to the Representation of Knowledge, 1987.
No 118	Mariam Kamkar, Nahid Shahmehri: Affect-Chaining in Program Flow Analysis Applied to Oueries of Pro-
	grams, 1987.
No 126	<b>Dan Strömberg:</b> Transfer and Distribution of Application Programs, 1987.
No 127	<b>Kristian Sandahl:</b> Case Studies in Knowledge Acquisition. Migration and User Acceptance of Expert Sys-
110 127	tems. 1987.
No 139	Christer Bäckström: Reasoning about Interdependent Actions 1988
No 140	Mats Wirén: On Control Strategies and Incrementality in Unification-Based Chart Parsing 1988
No 146	<b>John Hultman</b> : A Software System for Defining and Controlling Actions in a Mechanical System 1088
No 140	Johan Hutman. A Software System for Defining and Controlling Actions in a Mechanical System, 1988.
No 150	Tim nansen: Diagnosing Faults using Knowledge about Manunctioning Benavior, 1988.
NO 165	Jonas Lowgren: Supporting Design and Management of Expert System User Interfaces, 1989.
No 166	Ola Petersson: On Adaptive Sorting in Sequential and Parallel Models, 1989.
No 174	<b>Yngve Larsson:</b> Dynamic Configuration in a Distributed Environment, 1989.
No 177	<b>Peter Aberg:</b> Design of a Multiple View Presentation and Interaction Manager, 1989.
No 181	Henrik Eriksson: A Study in Domain-Oriented Tool Support for Knowledge Acquisition, 1989.
No 184	Ivan Rankin: The Deep Generation of Text in Expert Critiquing Systems, 1989.
No 187	Simin Nadjm-Tehrani: Contributions to the Declarative Approach to Debugging Prolog Programs, 1989.
No 189	Magnus Merkel: Temporal Information in Natural Language, 1989.
No 196	<b>Ulf Nilsson:</b> A Systematic Approach to Abstract Interpretation of Logic Programs, 1989.
No 197	Staffan Bonnier: Horn Clause Logic with External Procedures: Towards a Theoretical Framework, 1989.
No 203	<b>Christer Hansson:</b> A Prototype System for Logical Reasoning about Time and Action, 1990.
No 212	<b>Biörn Fiellborg:</b> An Approach to Extraction of Pipeline Structures for VLSI High-Level Synthesis, 1990.
No 230	Patrick Doherty: A Three-Valued Approach to Non-Monotonic Reasoning 1990
No 237	Tomas Sokolnicki: Coaching Partial Plans: An Approach to Knowledge-Based Tutoring, 1990
No 250	Lore Strömberg: Destrontem Debugging of Distributed Systems (100)
No 252	<b>Lars Stromberg.</b> I ostinotem Debugging of Distributed Systems, 1790.
No 255	Deter D. Holmony Using Corner structure Comparing Answers for Negative Quettes, 1990.
No 200	Feler D. Homes: Osnig Connectivity Graphics to Support Map-Kelated Keasoning, 1991.
NO 285	Dior Jonansson: Improving Implementation of Graphical User Interfaces for Object-Oriented Knowledge-
No 209	Dates, 1771.
No 290	Kon G Laisson: Akuviteispaseiau kaikyteinigi etti tyti ekonomisysteini, 1991.
NO 518	Lena Sromback: Studies in Extended Unification-Based Formatism for Linguistic Description: An Algorithm for Eastern Structures quick Distinguistics and a Damaged for Elevided Southers 1002
N. 210	for realize Structures with Disjunction and a Proposal for Flexible Systems, 1992.
NO 319	Mikael referencesion: DML-A Language and System for the Generation of Efficient Compilers from Denotatio-
NL 226	hai Specification, 1992.
NO 326	Andreas Kagedai: Logic Programming with External Procedures: an Implementation, 1992.
No 328	Patrick Lambrix: Aspects of Version Management of Composite Objects, 1992.
No 333	Xinli Gu: Testability Analysis and Improvement in High-Level Synthesis Systems, 1992.
No 335	<b>Torbjörn Näslund:</b> On the Role of Evaluations in Iterative Development of Managerial Support Sytems,
	1992.
No 348	Ulf Cederling: Industrial Software Development - a Case Study, 1992.
No 352	Magnus Morin: Predictable Cyclic Computations in Autonomous Systems: A Computational Model and Im-
	plementation, 1992.
No 371	Mehran Noghabai: Evaluation of Strategic Investments in Information Technology, 1993.
No 378	Mats Larsson: A Transformational Approach to Formal Digital System Design, 1993.
No 380	Johan Ringström: Compiler Generation for Parallel Languages from Denotational Specifications, 1993.
No 381	Michael Jansson: Propagation of Change in an Intelligent Information System, 1993.
No 383	Jonni Harrius: An Architecture and a Knowledge Representation Model for Expert Critiquing Systems, 1993.
No 386	<b>Per Österling:</b> Symbolic Modelling of the Dynamic Environments of Autonomous Agents. 1993.
No 398	Johan Boye: Dependency-based Groudness Analysis of Functional Logic Programs, 1993.

No 402	Lars Degerstedt: Tabulated Desolution for Wall Founded Semantics 1003
NO 402	Lars begenstedt: 1 abunated Resolution for wen Founded Semantics, 1995.
No 406	Anna Moberg: Satellitkontor - en studie av kommunikationsmonster vid arbete på distans, 1993.
No 414	<b>Peter Carlsson:</b> Separation av företagsledning och finansiering - fallstudier av företagsledarutköp ur ett agent-
	teoretiskt perspektiv, 1994.
No 417	<b>Camilla Sjöström:</b> Revision och lagreglering - ett historiskt perspektiv, 1994.
No 436	<b>Cecilia Sjöberg:</b> Voices in Design: Argumentation in Participatory Development, 1994.
No 437	Lars Viklund: Contributions to a High-level Programming Environment for a Scientific Computing, 1994.
No 440	Peter Lohore: Error Recovery Support in Manufacturing Control Systems 1994
EUS 2/04	<b>Owen Emiltage</b> Informationswitzen med warksambatelwalitet utwärdering begenzt på ett warksambatelwalitet
гпз 5/94	Wen Ernsson; mormationsystem med verksammetskvantet - utvardering baserat på ett verksammetsmirk-
FUG 4/04	ta och samskapande perspektiv, 1774.
FHS 4/94	<b>Karin Pettersson:</b> Informationssystemstrukturering, ansvarsforderining och anvandarinflytande - En kompa-
	rativ studie med utgangspunkt i tva informationssystemstrategier, 1994.
No 441	Lars Poignant: Informationsteknologi och företagsetablering - Effekter på produktivitet och region, 1994.
No 446	<b>Gustav Fahl:</b> Object Views of Relational Data in Multidatabase Systems, 1994.
No 450	<b>Henrik Nilsson:</b> A Declarative Approach to Debugging for Lazy Functional Languages, 1994.
No 451	Jonas Lind: Creditor - Firm Relations: an Interdisciplinary Analysis, 1994.
No 452	Martin Sköld: Active Rules based on Object Relational Queries - Efficient Change Monitoring Techniques,
	1994.
No 455	<b>Pär Carlshamre:</b> A Collaborative Approach to Usability Engineering: Technical Communicators and System
	Developers in Usability-Oriented Systems Development, 1994.
FHS 5/94	Stefan Cronholm: Varför CASE-verktyg i systemutyeckling? - En motiv- och konsekvensstudie avseende ar-
1110 0/91	betssätt och arbetsformer 1994
No 462	Mikaal Lindvall: A Study of Traceability in Object-Oriented Systems Development 1994
No 462	<b>Fradel Enlysin</b> , A Study of Traceability in Object-Oriented Systems Development, 1994.
NO 405	Fredrik Ninsson: Strategi och ekonomisk styrning - En studie av Sandviks forvarv av Banco Verktyg, 1994.
No 464	Hans Olsen: Collage Induction: Proving Properties of Logic Programs by Program Synthesis, 1994.
No 469	Lars Karlsson: Specification and Synthesis of Plans Using the Features and Fluents Framework, 1995.
No 473	<b>Ulf Söderman:</b> On Conceptual Modelling of Mode Switching Systems, 1995.
No 475	<b>Choong-ho Yi:</b> Reasoning about Concurrent Actions in the Trajectory Semantics, 1995.
No 476	<b>Bo Lagerström:</b> Successiv resultatavräkning av pågående arbeten Fallstudier i tre byggföretag, 1995.
No 478	<b>Peter Jonsson:</b> Complexity of State-Variable Planning under Structural Restrictions, 1995.
FHS 7/95	Anders Avdic: Arbetsintegrerad systemutyeckling med kalkylkprogram, 1995.
No 482	<b>Eva I. Bagnemalm:</b> Towards Student Modelling through Collaborative Dialogue with a Learning Compani-
110 402	on 1005
No 488	Fig. 1755.
110 400	Evaluation of a land in with paradigin Languages. Combining Object-Oriented and Kule-based
No 490	Flogramming, 1995.
No 489	Erik Stoy: A Petri Net Based Unified Representation for Hardware/Software Co-Design, 1995.
No 497	Johan Herber: Environment Support for Building Structured Mathematical Models, 1995.
No 498	<b>Stefan Svenberg:</b> Structure-Driven Derivation of Inter-Lingual Functor-Argument Trees for Multi-Lingual
	Generation, 1995.
No 503	Hee-Cheol Kim: Prediction and Postdiction under Uncertainty, 1995.
FHS 8/95	<b>Dan Fristedt:</b> Metoder i användning - mot förbättring av systemutveckling genom situationell metodkunskap
	och metodanalys, 1995.
FHS 9/95	Malin Bergvall: Systemförvaltning i praktiken - en kvalitativ studie avseende centrala begrepp, aktiviteter och
	ansvarsroller, 1995.
No 513	Joachim Karlsson: Towards a Strategy for Software Requirements Selection, 1995.
No 517	Jakob Axelsson: Schedulability-Driven Partitioning of Heterogeneous Real-Time Systems 1995
No 518	<b>Garan Earshund:</b> Toward Cooperative Advice-Giving Systems: The Expert Systems Experience 1995
No 522	Järgan Andenson, Bilder et ensförstater der en mit sträming 1005
No 522	Jorgen Andersson: Binder av sinarofetagares ekonomistyrining, 1993.
NO 558	Statian Florin: Enclient Management of Object-Oriented Queres with Late Binding, 1996.
NO 545	value Engelson: An Approach to Automatic Construction of Graphical User Interfaces for Applications in
N. 546	Scientific Computing, 1996.
No 546	Magnus werner : Multidatabase Integration using Polymorphic Queries and Views, 1996.
FiF-a 1/96	<b>Mikael Lind:</b> Affärsprocessinriktad förändringsanalys - utveckling och tillämpning av synsätt och metod,
	1996.
No 549	Jonas Hallberg: High-Level Synthesis under Local Timing Constraints, 1996.
No 550	Kristina Larsen: Förutsättningar och begränsningar för arbete på distans - erfarenheter från fyra svenska fö-
	retag. 1996.
No 557	Mikael Johansson: Quality Functions for Requirements Engineering Methods, 1996.
No 558	<b>Patrik Nordling:</b> The Simulation of Rolling Bearing Dynamics on Parallel Computers, 1996.
No 561	Anders Ekman: Exploration of Polygonal Environments, 1996.
No 563	Niclas Andersson: Compilation of Mathematical Models to Parallel Code, 1996
No 567	<b>John Janvald</b> : Simulation and Data Collection in Battle Training 1006
No 575	<b>Solar schradt</b> , Similation and Data Concerning in Battle Hamilies, 1770.
No 575	Mihoal Existence Communities Contense on Design Superior - Million of Fault-Flore Modules, 1990.
INU 370	Winkaci Ericessoni Commenting Systems as Design Support—A Wilzard-of-OZ Study, 1996.
INO 38/	Jorgen Lindstrom: Cneters anvandning av kommunikationsteknik, 1996.
No 589	<b>Esa Faikenrotn:</b> Data Management in Control Applications - A Proposal Based on Active Database Systems,
N. 861	1996.
No 591	<b>Niclas Wahllot:</b> A Default Extension to Description Logics and its Applications, 1996.
No 595	Annika Larsson: Ekonomisk Styrning och Organisatorisk Passion - ett interaktivt perspektiv, 1997.
NI 507	$\mathbf{L}_{1} = \mathbf{L}_{2} = \mathbf{A} \mathbf{V}_{1} \mathbf{L}_{2} + \mathbf{L}_{2} \mathbf{L}_{2}$

**Ling Lin:** A Value-based Indexing Technique for Time Sequences, 1997. No 597

N. 500	Description of C <sup>3</sup> E <sup>+</sup> A Mission 11 Second Second Second Technology 1007
NO 598	Rego Granuna: C Fire - A Microworld Supporting Emergency Management Training, 1997.
No 599	Peter Ingels: A Robust Text Processing Technique Applied to Lexical Error Recovery, 1997.
No 607	<b>Per-Arne Persson:</b> Toward a Grounded Theory for Support of Command and Control in Military Coalitions, 1997.
No 609	Jonas S Karlsson: A Scalable Data Structure for a Parallel Data Server, 1997.
FiF-a 4	<b>Carita Åbom:</b> Videomötesteknik i olika affärssituationer - möjligheter och hinder, 1997.
FiF-a 6	<b>Tommy Wedlund</b> : Att skapa en företagsanpassad systemutvecklingsmodell - genom rekonstruktion, värde- ring och vidarentveckling i T50-bolag inom ABB 1997
No 615	Silvia Coradeschi: A Decision-Mechanism for Reactive and Coordinated Agents 1997
No 623	<b>Jan Olinen</b> : Det flevible kontorets utveckling nå Digital – Ett stöd för multiflev? 1007
No 626	<b>Jan Olimen.</b> Det lexible contores avecenting for Digital - Ent stor for infinitely (1977).
No 627	<b>David Byets:</b> Towards Estimating Software Testability Using Static Analysis, 1997.
NO 627	Fredrik Ektind: Declarative Error Diagnosis of GAPLog Programs, 1997.
No 629	Gunilla lyetors: Krigsspel coh Informationsteknik inför en oforutsägbar framtid, 1997.
No 631	Jens-Olof Lindh: Analysing Traffic Safety from a Case-Based Reasoning Perspective, 1997
No 639	Jukka Mäki-Turja:. Smalltalk - a suitable Real-Time Language, 1997.
No 640	Juha Takkinen: CAFE: Towards a Conceptual Model for Information Management in Electronic Mail, 1997.
No 643	Man Lin: Formal Analysis of Reactive Rule-based Programs, 1997.
No 653	Mats Gustafsson: Bringing Role-Based Access Control to Distributed Systems, 1997.
FiF-a 13	Boris Karlsson: Metodanalys för förståelse och utveckling av systemutvecklingsverksamhet. Analys och vär-
	dering av systemutvecklingsmodeller och dess användning. 1997.
No 674	Marcus Biäreland: Two Aspects of Automating Logics of Action and Change - Regression and Tractability
110 07 1	1998.
No 676	Jan Håkegård: Hiera rchical Test Architecture and Board-Level Test Controller Synthesis, 1998.
No 668	<b>Per-Ove Zetterlund</b> : Normering av svensk redovisning - En studie av tillkomsten av Redovisningsrådets re-
	kommendation om koncernredovisning (RR01:91), 1998.
No 675	Jimmy Tjäder: Projektledaren & planen - en studie av projektledning i tre installations- och systemutveck-
	lingsprojekt, 1998.
FiF-a 14	Ulf Melin: Informationssystem vid ökad affärs- och processorientering - egenskaper, strategier och utveck-
	ling, 1998.
No 695	<b>Tim Hever</b> : COMPASS: Introduction of Formal Methods in Code Development and Inspection, 1998.
No 700	Patrik Hägglund: Programming Languages for Computer Algebra 1998
FiF-a 16	Marie-Therese Christiansson: Inter-organistorisk verksamhetsutveckling - metoder som stöd vid utveckling
111 11 10	av nartnerskap och informationssystem 1998
No 712	Christing Wennestam: Information on immateriella resurser. Investeringer i forskning och utveckling samt
110 /12	i nersonal inom skogsindustrin 1998
No 719	Jostim Custafsson: Extending Temporal Action Logic for Ramification and Concurrency 1998
No 723	Hanrik André, Jönsson- Indeving timpotal rection begin for trainfeation and Concentrative, 1990.
No 725	<b>Exit</b> A mater sons son. Indexing time-series data dang text indexing includes, 1997.
NO 725	Erik Larsson: Figh-Level Testaoliniy Anarysis and Ennancement Techniques, 1998.
NO 750	<b>Cari-Jonan Westin</b> : informationsforsorjining: en fraga om ansvar - aktiviteter och uppdrag i tem stora svenska
N. 721	organisationers operativa miorinationsforsorphing, 1998.
NO /31	Ase Jansson: Miljonansyn - en dei i foretags styrning, 1998.
No 733	Thomas Padron-McCarthy: Performance-Polymorphic Declarative Queries, 1998.
No 734	Anders Bäckström: Värdeskapande kreditgivning - Kreditriskhantering ur ett agentteoretiskt perspektiv,
<b>E</b> ' <b>E A</b> 1	1998.
F1F-a 21	Ulf Seigerroth: Integration av förändringsmetoder - en modell för valgrundad metodintegration, 1999.
FiF-a 22	<b>Fredrik Oberg:</b> Object-Oriented Frameworks - A New Strategy for Case Tool Development, 1998.
No 737	Jonas Mellin: Predictable Event Monitoring, 1998.
No 738	Joakim Eriksson: Specifying and Managing Rules in an Active Real-Time Database System, 1998.
FiF-a 25	Bengt E W Andersson: Samverkande informationssystem mellan aktörer i offentliga åtaganden - En teori om
	aktörsarenor i samverkan om utbyte av information, 1998.
No 742	Pawel Pietrzak: Static Incorrectness Diagnosis of CLP (FD), 1999.
No 748	Tobias Ritzau: Real-Time Reference Counting in RT-Java, 1999.
No 751	Anders Ferntoft. Elektronisk affärskommunikation - kontaktkostnader och kontaktprocesser mellan kunder
	<b>Thiutis I thiutis Liektionisk unuiskommunikuton</b> Kontuktootinudei oon kontuktoitoeosoo menui kunuei
No 752	och leverantörer på producentmarknader,1999.
No 753	Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.
	Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999. Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.
No 754	Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999. Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999. Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie
No 754	Jo Skåmedal: Arbete på producentmarknader, 1999. Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999. Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999. Magnus Lindahl: Bankens vilkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.
No 754 No 766	<ul> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data. 1999.</li> </ul>
No 754 No 766 No 769	<ul> <li>Johan Alvehar, På producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures. 1999.</li> </ul>
No 754 No 766 No 769 No 775	<ul> <li>Johan Alvehar, På producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Anders Henriksson: Unique kernel diagnosis. 1999.</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30	John Alvehus: Möter på producentmarknader, 1999. Jo Skåmedal: Arbete på groducentmarknader, 1999. Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999. Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000. Martin V. Howard: Designing dynamic visualizations of temporal data, 1999. Jesper Andersson: Towards Reactive Software Architectures, 1999. Anders Henriksson: Unique kernel diagnosis, 1999. Pär L fägerfalk: Pragmatizion of Information Systems - A Theoretical and Methodological Outling, 1000
No 754 No 766 No 769 No 775 FiF-a 30 No 787	Jo Skåmedal: Arbete på producentmarknader, 1999. Jo Skåmedal: Arbete på producentmarknader, 1999. Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999. Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000. Martin V. Howard: Designing dynamic visualizations of temporal data, 1999. Jesper Andersson: Towards Reactive Software Architectures, 1999. Anders Henriksson: Unique kernel diagnosis, 1999. Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999. Charlotte Biörkenzer: Learning for the next project - Bearers and barriers in knowledge transfer within an
No 754 No 766 No 769 No 775 FiF-a 30 No 787	<ul> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Anders Henriksson: Unique kernel diagnosis, 1999.</li> <li>Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organization 1999.</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787	<ul> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organisation, 1999.</li> <li>Håkan Nilsson: Informationsteknik som drivkraft i granekningenrocessen - En studie av fore revisionsburåer</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787 No 788	<ul> <li>Joshi Verantörer på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organisation, 1999.</li> <li>Håkan Nilsson: Informationsteknik som drivkraft i granskningsprocessen - En studie av fyra revisionsbyråer, 2000</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787 No 788 No 790	<ul> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens vilkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Anders Henriksson: Unique kernel diagnosis, 1999.</li> <li>Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organisation, 1999.</li> <li>Håkan Nilsson: Informationsteknik som drivkraft i granskningsprocessen - En studie av fyra revisionsbyråer, 2000.</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787 No 788 No 790 No 791	<ul> <li>Jo Skåmedal: Arbete på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Anders Henriksson: Unique kernel diagnosis, 1999.</li> <li>Pär J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organisation, 1999.</li> <li>Håkan Nilsson: Informationsteknik som drivkraft i granskningsprocessen - En studie av fyra revisionsbyråer, 2000.</li> <li>Erik Berglund: Use-Oriented Documentation in Software Development, 1999.</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787 No 788 No 790 No 791 No 90	<ul> <li>John Verantörer på producentmarknader, 1999.</li> <li>Jo Skåmedal: Arbete på distans och arbetsformens påverkan på resor och resmönster, 1999.</li> <li>Johan Alvehus: Mötets metaforer. En studie av berättelser om möten, 1999.</li> <li>Magnus Lindahl: Bankens villkor i låneavtal vid kreditgivning till högt belånade företagsförvärv: En studie ur ett agentteoretiskt perspektiv, 2000.</li> <li>Martin V. Howard: Designing dynamic visualizations of temporal data, 1999.</li> <li>Jesper Andersson: Towards Reactive Software Architectures, 1999.</li> <li>Pår J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Pår J. Ågerfalk: Pragmatization of Information Systems - A Theoretical and Methodological Outline, 1999.</li> <li>Charlotte Björkegren: Learning for the next project - Bearers and barriers in knowledge transfer within an organisation, 1999.</li> <li>Håkan Nilsson: Informationsteknik som drivkraft i granskningsprocessen - En studie av fyra revisionsbyråer, 2000.</li> <li>Erik Berglund: Use-Oriented Documentation in Software Development, 1999.</li> <li>Klas Gäre: Verksamhetsförändringar i samband med IS-införande, 1999.</li> </ul>
No 754 No 766 No 769 No 775 FiF-a 30 No 787 No 788 No 790 No 791 No 800	<ul> <li>Johnstein Kunstein Ku</li></ul>

No 809	Flavius Gruian: Energy-Aware Design of Digital Systems, 2000.
FiF-a 32	Karin Hedström: Kunskapsanvändning och kunskapsutveckling hos verksamhetskonsulter - Erfarenheter
N. 000	från ett FOU-samarbete, 2000.
No 808	Linda Askenas: Affarssystemet - En studie om teknikens aktiva och passiva roll i en organisation, 2000.
No 820	Jean Paul Meynard: Control of industrial robots through high-level task programming, 2000.
No 823	Lars Hult: Publika Gransytor - ett designexempel, 2000.
No 832	<b>Paul Pop:</b> Scheduling and Communication Synthesis for Distributed Real-Time Systems, 2000.
FiF-a 34	Göran Hultgren: Nätverksinriktad Förändringsanalys - perspektiv och metoder som stöd för förståelse och
	utveckling av affärsrelationer och informationssystem, 2000.
No 842	<b>Magnus Kald:</b> The role of management control systems in strategic business units, 2000.
No 844	<b>Mikael Cäker:</b> Vad kostar kunden? Modeller för intern redovisning, 2000.
FiF-a 37	<b>Ewa Braf</b> : Organisationers kunskapsverksamheter - en kritisk studie av "knowledge management", 2000.
FiF-a 40	Henrik Lindberg: Webbaserade affärsprocesser - Möjligheter och begränsningar, 2000.
FiF-a 41	Benneth Christiansson: Att komponentbasera informationssystem - Vad säger teori och praktik?, 2000.
No. 854	<b>Ola Pettersson:</b> Deliberation in a Mobile Robot, 2000.
No 863	Dan Lawesson: Towards Behavioral Model Fault Isolation for Object Oriented Control Systems, 2000.
No 881	Johan Moe: Execution Tracing of Large Distributed Systems, 2001.
No 882	Yuxiao Zhao: XML-based Frameworks for Internet Commerce and an Implementation of B2B
	e-procurement, 2001.
No 890	Annika Flycht-Eriksson: Domain Knowledge Management inInformation-providing Dialogue systems,
FiF-a 47	2001. Par. Arne Segerkvist: Webbaserade imaginära organisationers samverkansformer: Informationssystemarki.
in u ii	tektur och aktörssamverkan som förutsättningar för affärsprocesser, 2001.
No 894	Stefan Svarén: Styrning av investeringar i divisionaliserade företag - Ett koncernperspektiv, 2001.
No 906	Lin Han: Secure and Scalable E-Service Software Delivery, 2001.
No 917	Emma Hansson: Optionsprogram för anställda - en studie av svenska börsföretag, 2001.
No 916	Susanne Odar: IT som stöd för strategiska beslut, en studie av datorimplementerade modeller av verksamhet
E'E - 40	som stod for beslut om anskaffning av JAS 1982, 2002.
F1F-a-49	Steran Holgersson: 11-system och nittrering av verksammetskunskap - kväntetsproblem vid analyser och be- slutfortande som hurgers på uppgrifter kämtede från palisans IT austarn 2001
FiF-a-51	Substatiande som bygger på uppginter nämade nän ponsens 11-system, 2001. <b>Per Oscarsson-</b> Informationssäkerhet i verksamheter - begrenn och modeller som stöd för förståelse av infor-
111 4.51	mationssäkerhet och dess hantering. 2001.
No 919	Luis Alejandro Cortes: A Petri Net Based Modeling and Verification Technique for Real-Time Embedded
	Systems, 2001.
No 915	Niklas Sandell: Redovisning i skuggan av en bankkris - Värdering av fastigheter. 2001.
No 931	Fredrik Elg: Ett dynamiskt perspektiv på individuella skillnader av heuristisk kompetens, intelligens, mentala
N. 022	modeller, mål och konfidens i kontroll av mikrovärlden Moro, 2002.
NO 935	Peter Aronsson: Automatic Parallelization of Simulation Code from Equation Based Simulation Languages, 2002
No 938	Rourhane Kadmiry: Euzzy Control of Unmanned Heliconter, 2002
No 942	<b>Patrik Haslum:</b> Prediction as a Knowledge Representation Problem: A Case Study in Model Design, 2002.
No 956	Robert Sevenius: On the instruments of governance - A law & economics study of capital instruments in li-
	mited liability companies, 2002.
FiF-a 58	Johan Petersson: Lokala elektroniska marknadsplatser - informationssystem för platsbundna affärer, 2002.
No 964	Peter Bunus: Debugging and Structural Analysis of Declarative Equation-Based Languages, 2002.
No 973	Gert Jervan: High-Level Test Generation and Built-In Self-Test Techniques for Digital Systems, 2002.
No 958	<b>Fredrika Berglund:</b> Management Control and Strategy - a Case Study of Pharmaceutical Drug Development,
$\mathbf{E} = \mathbf{E} + \mathbf{C} \mathbf{I}$	2002.
FIF-a 01	Freurik Karisson: Meta-Method for Method Configuration - A Rational United Process Case, 2002.
NO 965	Sorth Manorache: Schedulability Analysis of Real-Time Systems with Stochastic Task Execution Times, 2002
No 982	Diana Szentiványi: Performance and Availability Trade-offs in Fault-Tolerant Middleware 2002
No 989	<b>Jakov Nakhimovski:</b> Modeling and Simulation of Contacting Flexible Bodies in Multibody Systems, 2002.
No 990	Levon Saldamli: PDEModelica - Towards a High-Level Language for Modeling with Partial Differential
	Equations, 2002.
No 991	Almut Herzog: Secure Execution Environment for Java Electronic Services, 2002.
No 999	Jon Edvardsson: Contributions to Program- and Specification-based Test Data Generation, 2002
No 1000	Anders Arpteg: Adaptive Semi-structured Information Extraction, 2002.
No 1001	Andrzej Bednarski: A Dynamic Programming Approach to Optimal Retargetable Code Generation for
No. 099	Irregular Architectures, 2002.
NO 988 FiE-2 62	<b>Mattus Arvoia:</b> Good to use : Use quarty of munt-user applications in the nome, 2005.
111-0.02	projektivitetsmoden om organisationers formaga at unanpa
No 1003	<b>Pernilla Ovarfordt:</b> User experience of spoken feedback in multimodal interaction, 2003.
No 1005	Alexander Siemers: Visualization of Dynamic Multibody Simulation With Special Reference to Contacts,
	2003.
No 1008	Jens Gustavsson: Towards Unanticipated Runtime Software Evolution, 2003.
No 1010	Calin Curescu: Adaptive QoS-aware Resource Allocation for Wireless Networks, 2003.
No 1015	Anna Andersson: Management Information Systems in Process-oriented Healthcare Organisations, 2003.
No 1018	<b>D</b> JOFH JOHANSON: FEEDFORWATG CONTROL IN DYNAMIC SITUATIONS, 2003.
110 1022	Systems 2003
FiF-a 65	Britt-Marie Johansson: Kundkommunikation nå distans - en studie om kommunikationsmediete betydelse i

FiF-a 65 **Britt-Marie Johansson:** Kundkommunikation på distans - en studie om kommunikationsmediets betydelse i affärstransaktioner, 2003.

- No 1024 Aleksandra Tešanovic: Towards Aspectual Component-Based Real-Time System Development, 2003. No 1034 Arja Vainio-Larsson: Designing for Use in a Future Context - Five Case Studies in Retrospect, 2003. No 1033 Peter Nilsson: Svenska bankers redovisningsval vid reservering för befarade kreditförluster - En studie vid införandet av nva redovisningsregler. 2003. FiF-a 69 Fredrik Ericsson: Information Technology for Learning and Acquiring of Work Knowledge, 2003. No 1049 Marcus Comstedt: Towards Fine-Grained Binary Composition through Link Time Weaving, 2003. Åsa Hedenskog: Increasing the Automation of Radio Network Control, 2003. No 1052 No 1054 Claudiu Duma: Security and Efficiency Tradeoffs in Multicast Group Key Management, 2003. FiF-a 71 Emma Eliason: Effektanalys av IT-systems handlingsutrymme, 2003. No 1055 Carl Cederberg: Experiments in Indirect Fault Injection with Open Source and Industrial Software, 2003. No 1058 Daniel Karlsson: Towards Formal Verification in a Component-based Reuse Methodology, 2003. FiF-a 73 Anders Hjalmarsson: Att etablera och vidmakthålla förbättringsverksamhet - behovet av koordination och interaktion vid förändring av systemutvecklingsverksamheter, 2004. No 1079 Pontus Johansson: Design and Development of Recommender Dialogue Systems, 2004. No 1084 Charlotte Stoltz: Calling for Call Centres - A Study of Call Centre Locations in a Swedish Rural Region, 2004. FiF-a 74 Björn Johansson: Deciding on Using Application Service Provision in SMEs, 2004. No 1094 Genevieve Gorrell: Language Modelling and Error Handling in Spoken Dialogue Systems, 2004. No 1095 Ulf Johansson: Rule Extraction - the Key to Accurate and Comprehensible Data Mining Models, 2004. No 1099 Sonia Sangari: Computational Models of Some Communicative Head Movements, 2004. No 1110 Hans Nässla: Intra-Family Information Flow and Prospects for Communication Systems, 2004. No 1116 Henrik Sällberg: On the value of customer loyalty programs - A study of point programs and switching costs, 2004 FiF-a 77 Ulf Larsson: Designarbete i dialog - karaktärisering av interaktionen mellan användare och utvecklare i en systemutvecklingsprocess, 2004. No 1126 No 1127 Andreas Borg: Contribution to Management and Validation of Non-Functional Requirements, 2004. Per-Ola Kristensson: Large Vocabulary Shorthand Writing on Stylus Keyboard, 2004. No 1132 Pär-Anders Albinsson: Interacting with Command and Control Systems: Tools for Operators and Designers, 2004No 1130 Ioan Chisalita: Safety-Oriented Communication in Mobile Networks for Vehicles, 2004. No 1138 Thomas Gustafsson: Maintaining Data Consistency im Embedded Databases for Vehicular Systems, 2004. No 1149 Vaida Jakoniené: A Study in Integrating Multiple Biological Data Sources, 2005. Abdil Rashid Mohamed: High-Level Techniques for Built-In Self-Test Resources Optimization, 2005. Adrian Pop: Contributions to Meta-Modeling Tools and Methods, 2005. No 1156 No 1162 No 1165 Fidel Vascós Palacios: On the information exchange between physicians and social insurance officers in the Sick leave process: an Activity Theoretical perspective, 2005. Jenny Lagsten: Verksamhetsutvecklande utvärdering i informationssystemprojekt, 2005. FiF-a 84 No 1166 Emma Larsdotter Nilsson: Modeling, Simulation, and Visualization of Metabolic Pathways Using Modelica, 2005 Christina Keller: Virtual Learning Environments in higher education. A study of students' acceptance of edu-No 1167 cational technology, 2005. No 1168 Cécile Åberg: Integration of organizational workflows and the Semantic Web, 2005. FiF-a 85 Anders Forsman: Standardisering som grund för informationssamverkan och IT-tjänster - En fallstudie baserad på trafikinformationstjänsten RDS-TMC, 2005. Yu-Hsing Huang: A systemic traffic accident model, 2005. No 1171 FiF-a 86 Jan Olausson: Att modellera uppdrag - grunder för förståelse av processinriktade informationssystem i transaktionsintensiva verksamheter, 2005. Petter Ahlström: Affärsstrategier för seniorbostadsmarknaden, 2005. No 1172 No 1183 Mathias Cöster: Beyond IT and Productivity - How Digitization Transformed the Graphic Industry, 2005. Asa Horzella: Beyond IT and Productivity - Effects of Digitized Information Flows in Grocery Distribution, No 1184 2005 Maria Kollberg: Beyond IT and Productivity - Effects of Digitized Information Flows in the Logging No 1185 Industry, 2005 No 1190 David Dinka: Role and Identity - Experience of technology in professional settings, 2005. No 1191 Andreas Hansson: Increasing the Storage Capacity of Recursive Auto-associative Memory by Segmenting Data, 2005. No 1192 Nicklas Bergfeldt: Towards Detached Communication for Robot Cooperation, 2005. No 1194 Dennis Maciuszek: Towards Dependable Virtual Companions for Later Life, 2005. No 1204 Beatrice Alenljung: Decision-making in the Requirements Engineering Process: A Human-centered Approach, 2005 Anders Larsson: System-on-Chip Test Scheduling and Test Infrastructure Design, 2005. No 1206 No 1207 John Wilander: Policy and Implementation Assurance for Software Security, 2005. No 1209 Andreas Käll: Översättningar av en managementmodell - En studie av införandet av Balanced Scorecard i ett landsting, 2005. No 1225 He Tan: Aligning and Merging Biomedical Ontologies, 2006. No 1228 Artur Wilk: Descriptive Types for XML Query Language Xcerpt, 2006. No 1229 Per Olof Pettersson: Sampling-based Path Planning for an Autonomous Helicopter, 2006. No 1231 Kalle Burbeck: Adaptive Real-time Anomaly Detection for Safeguarding Critical Networks, 2006. No 1233 Daniela Mihailescu: Implementation Methodology in Action: A Study of an Enterprise Systems Implementation Methodology, 2006. No 1244 Jörgen Skågeby: Public and Non-public gifting on the Internet, 2006. No 1248 Karolina Eliasson: The Use of Case-Based Reasoning in a Human-Robot Dialog System, 2006. No 1263 Misook Park-Westman: Managing Competence Development Programs in a Cross-Cultural Organisation-What are the Barriers and Enablers, 2006. FiF-a 90 Amra Halilovic: Ett praktikperspektiv på hantering av mjukvarukomponenter, 2006.
- No 1272 Raquel Flodström: A Framework for the Strategic Management of Information Technology, 2006.

- No 1277 Viacheslav Izosimov: Scheduling and Optimization of Fault-Tolerant Embedded Systems, 2006.
- No 1283 Håkan Hasewinkel: A Blueprint for Using Commercial Games off the Shelf in Defence Training, Education and Research Simulations, 2006.
- Hanna Broberg: Verksamhetsanpassade IT-stöd Designteori och metod, 2006. Robert Kaminski: Towards an XML Document Restructuring Framework, 2006 FiF-a 91
- No 1286
- No 1293 Jiri Trnka: Prerequisites for data sharing in emergency management, 2007.
- No 1302 Björn Hägglund: A Framework for Designing Constraint Stores, 2007.
- No 1303 Daniel Andreasson: Slack-Time Aware Dynamic Routing Schemes for On-Chip Networks, 2007.
- No 1305 Magnus Ingmarsson: Modelling User Tasks and Intentions for Service Discovery in Ubiquitous Computing, 2007.
- No 1306 Gustaf Svedjemo: Ontology as Conceptual Schema when Modelling Historical Maps for Database Storage, 2007
- No 1307 Gianpaolo Conte: Navigation Functionalities for an Autonomous UAV Helicopter, 2007.
- No 1309 Ola Leifler: User-Centric Critiquing in Command and Control: The DKExpert and ComPlan Approaches, 2007
- No 1312 Henrik Svensson: Embodied simulation as off-line representation, 2007.
- No 1313 Zhiyuan He: System-on-Chip Test Scheduling with Defect-Probability and Temperature Considerations, 2007
- Jonas Elmqvist: Components, Safety Interfaces and Compositional Analysis, 2007. No 1317
- No 1320
- Wilson Dimposite Composition of Systems 2007.
  Magnus Lundqvist: Information Demand and Use: Improving Information Flow within Small-scale Business No 1323 Contexts, 2007.
- No 1329 Martin Magnusson: Deductive Planning and Composite Actions in Temporal Action Logic, 2007.